UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

Amendment No. 1 to FORM S-1 REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933

Dave & Buster's Entertainment, Inc.

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation or organization) 5812 (Primary Standard Industrial Classification Code Number) 35-2382255 (I.R.S. Employer Identification Number)

2481 Mañana Drive Dallas, Texas 75220 (214) 357-9588

(Address, including zip code, and telephone number, including area code, of registrant's principal executive offices)

Stephen M. King Chief Executive Officer Dave & Buster's Entertainment, Inc. 2481 Mañana Drive Dallas, Texas 75220 (214) 357-9588

(Name, address, including zip code, and telephone number, including area code, of agent for service)

Copies to:

Corey R. Chivers, Esq. Alexander D. Lynch, Esq. Weil, Gotshal & Manges LLP 767 Fifth Avenue New York, New York 10153 (212) 310-8000 (Phone) (212) 310-8007 (Fax) Marc D. Jaffe, Esq. lan D. Schuman, Esq. Latham & Watkins LLP 885 Third Avenue, Suite 1000 New York, New York 10022 (212) 906-1200 (Phone) (212) 751-4864 (Fax)

Approximate date of commencement of proposed sale to the public: As soon as practicable after the effective date of this registration statement.

If any of the securities being registered on this Form are to be offered on a delayed or continuous basis pursuant to Rule 415 under the Securities Act of 1933, check the following box.

If this Form is filed to register additional securities for an offering pursuant to Rule 462(b) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering.

If this Form is a post effective amendment filed pursuant to Rule 462(c) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering.

If this Form is a post effective amendment filed pursuant to Rule 462(d) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering.

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer," and "smaller reporting company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer $\ \square$

Accelerated filer

ier 🗆

Non-accelerated filer 🗵

Smaller reporting company

The registrant hereby amends this registration statement on such date or dates as may be necessary to delay its effective date until the registrant shall file a further amendment which specifically states that this registration statement shall thereafter become effective in accordance with Section 8(a) of the Securities Act of 1933 or until the registration statement shall become effective on such date as the Commission, acting pursuant to Section 8(a), may determine.

EXPLANATORY NOTE

The sole purpose of this amendment is to file certain exhibits to the registration statement as indicated in Item 16(a) of Part II of this amendment. No change is made to the preliminary prospectus constituting Part I of the registration statement or Items 13, 14, 15, or 17 of Part II of the registration statement. Accordingly, this amendment consists only of the facing page, this explanatory note and Item 16(a) of Part II and the signatures of the registration statement.

Item 16. Exhibits and financial statement schedules.

EXHIBIT NUMBER	DESCRIPTION OF EXHIBITS	
1.1	Form of Underwriting Agreement	
3.1	Form of Second Amended and Restated Certificate of Incorporation of the Registrant	
3.2	Form of Second Amended and Restated Bylaws of the Registrant	
4.1	Form of Stock Certificate	
4.2	Form of Stockholders' Agreement, among Dave & Buster's Entertainment, Inc., Oak Hill Capital Partners III, L.P., and Oak Hill Capital Management Partners III, L.P.	
4.3	Form of Registration Rights Agreement, among Dave & Buster's Entertainment, Inc., Oak Hill Capital Partners III, L.P., Oak Hill Capital Management Partners III, L.P. and the additional stockholders named therein	
5.1†	Opinion of Weil, Gotshal & Manges LLP	
10.1	Second Amendment, dated as of May 14, 2013, to the Credit Agreement, dated as of June 1, 2010, among Dave & Buster's Holdings, Inc., Dave & Buster's, Inc., 6131646 Canada, Inc. and the several banks and other financial institutions or entities from time to time parties thereto (incorporated by reference to Exhibit 10.8 to the quarterly report on Form 10-Q filed by Dave & Buster's, Inc. on June 17, 2013)	
10.2	Form of Amended and Restated Employment Agreement, dated as of May 2, 2010, by and among Dave & Buster's Management Corporation, Dave & Buster's, Inc., and the various executive officers of Dave & Buster's, Inc. (incorporated by reference to Exhibit 10.2 to the Form S-4 Registration Statement filed by Dave & Buster's, Inc. on August 11, 2010 (No. 333-168759))	
10.3	Dave & Buster's Parent, Inc. 2010 Management Incentive Plan (incorporated by reference to Exhibit 10.3 to the Form S-4 Registration Statement filed by Dave & Buster's, Inc. on August 11, 2010 (No. 333-168759))	
10.4	Amendment No. 1 to the Dave & Buster's Parent, Inc. 2010 Management Incentive Plan (incorporated by reference to Exhibit 10.2 to the quarterly report on Form 10-Q filed by Dave & Buster's, Inc. on June 15, 2011)	
10.5	Amendment No. 2 to the Dave & Buster's Parent, Inc. 2010 Management Incentive Plan (incorporated by reference to Exhibit 10.6 to the annual report on Form 10-K filed by Dave & Buster's, Inc. on April 16, 2013)	
10.6	Expense Reimbursement Agreement, dated as of June 1, 2010, by and between Dave & Buster's, Inc. and Oak Hill Capital Management LLC (incorporated by reference to Exhibit 10.6 to the annual report on Form 10-K filed by Dave & Buster's, Inc. on April 12, 2012)	
10.7	Form of Dave & Buster's Entertainment, Inc. 2014 Omnibus Incentive Plan	
10.8*	Employment Agreement, dated as of February 14, 2011, by and among Dave & Buster's Management Corporation, Dave & Buster's, Inc. and Dolf Berle	
10.9	Form of Amended and Restated Nonqualified Stock Option Agreement, by and between Dave & Buster's Entertainment, Inc. and each of Stephen M. King, Kevin Bachus, Dolf Berle, Joe DeProspero, Sean Gleason, Brian A. Jenkins, Margo L. Manning, Michael J. Metzinger, John B. Mulleady, J. Michael Plunkett, Jay L. Tobin, David A. Jones, Alan J. Lacy, Kevin M. Sheehan, Michael J. Griffith and Jonathan S. Halkyard	

II-1

exhibit Number	DESCRIPTION OF EXHIBITS
10.10	Form of Nonqualified Stock Option Award Agreement under the Dave & Buster's Entertainment, Inc. 2014 Omnibus Incentive Plan
10.11*	Dave & Buster's Select Executive Retirement Plan as amended and restated by Dave & Buster's I, L.P., effective January 1, 2005
10.12	Form of Indemnification Agreement for directors, executive officers and key employees
10.13*	Credit Agreement, dated as of July 25, 2014, among Dave & Buster's Holdings, Inc., Dave & Buster's, Inc., the other guarantors from time to time parties thereto and Jefferies Finance LLC, as administrative agent.
10.14*	Offer Letter, dated October 1, 2011, by and between Dave & Buster's, Inc. and Kevin Bachus
11.1	Statement regarding computation of per share earnings (incorporated by reference to Notes to the Financial Statements included in Part I of this Registration Statement)
21.1*	List of subsidiaries of the Registrant
23.1*	Consent of KPMG LLP, Independent Registered Public Accounting Firm
23.2†	Consent of Weil, Gotshal & Manges LLP (included in the opinion filed as Exhibit 5.1 hereto)
24.1*	Power of Attorney of Stephen M. King (included on signature page)
24.2*	Power of Attorney of Brian A. Jenkins (included on signature page)
24.3*	Power of Attorney of Tyler J. Wolfram (included on signature page)
24.4*	Power of Attorney of J. Taylor Crandall (included on signature page)
24.5*	Power of Attorney of Michael J. Griffith (included on signature page)
24.6*	Power of Attorney of Jonathan S. Halkyard (included on signature page)
24.7*	Power of Attorney of David A. Jones (included on signature page)
24.8*	Power of Attorney of Alan J. Lacy (included on signature page)
24.9*	Power of Attorney of Kevin M. Mailender (included on signature page)
24.10*	Power of Attorney of Kevin M. Sheehan (included on signature page)

* Previously filed.† To be filed by amendment.

No financial statement schedules are provided because the information called for is not applicable or is shown in the financial statements or notes thereto.

II-2

SIGNATURES

Pursuant to the requirements of the Securities Act of 1933, as amended, the registrant has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Dallas, State of Texas, on the 23rd day of September, 2014.

DAVE & BUSTER'S ENTERTAINMENT, INC.

By: /s/ Stephen M. King Name: Stephen M. King Title: Chief Executive Officer

Pursuant to the requirements of the Securities Act of 1933, as amended, this registration statement has been signed below by the following persons in the capacities indicated on the 23rd of September, 2014.

SIGNATURE	TITLE
/s/ Stephen M. King	Chief Executive Officer and Director
Stephen M. King	(Principal Executive Officer)
/s/ Brian A. Jenkins	Senior Vice President and
Brian A. Jenkins	Chief Financial Officer
	(Principal Financial and Accounting Officer)
*	Director
J. Taylor Crandall	Director
*	
Michael J. Griffith	
*	
Jonathan S. Halkyard	Director
*	
David A. Jones	
*	
Alan J. Lacy	
*	Director
Kevin M. Mailender	
*	Director
Kevin M. Sheehan	
*	Director
Tyler J. Wolfram	
*By: /s/ Jay L.Tobin	
*By: /s/ Jay L.Tobin Jay L.Tobin	

Dave & Buster's Entertainment, Inc.

[] Shares of Common Stock, par value \$0.01 per share

Underwriting Agreement

[], 2014

Jefferies LLC Piper Jaffray & Co. As representatives of the several Underwriters named in Schedule I hereto,

c/o Jefferies LLC 520 Madison Avenue New York, New York 10022

c/o Piper Jaffray & Co. 800 Nicollet Mall Minneapolis, Minnesota 55402

Ladies and Gentlemen:

Dave & Buster's Entertainment, Inc., a Delaware corporation (the "Company"), proposes, subject to the terms and conditions stated herein, to issue and sell to the Underwriters named in Schedule I hereto (the "Underwriters") an aggregate of [___] shares (the "Firm Shares") and, at the election of the Underwriters, up to [___] additional shares (the "Optional Shares") of Common Stock, par value \$0.01 per share ("Stock"), of the Company. The Firm Shares and the Optional Shares that the Underwriters elect to purchase pursuant to Section 2 hereof are herein collectively called the "Shares".

1. (a) The Company represents and warrants to, and agrees with, each of the Underwriters that:

(i) A registration statement on Form S-1 (File No. 333-198641) (the "Initial Registration Statement") in respect of the Shares has been filed with the Securities and Exchange Commission (the "Commission"); the Initial Registration Statement, any pre-effective amendment thereto or any post-effective amendment thereto, each in the form heretofore delivered to you, and, excluding exhibits thereto, to you for each of the other Underwriters, have been declared effective by the

Commission in such form; other than a registration statement, if any, increasing the size of the offering (a "Rule 462(b) Registration Statement"), filed pursuant to Rule 462(b) under the Securities Act of 1933, as amended (the "Act"), which became effective upon filing, no other document with respect to the Initial Registration Statement (other than any pre-effective amendment thereto or any preliminary prospectus included in the Initial Registration Statement or filed with the Commission pursuant to Rule 424(a) of the rules and regulations of the Commission under the Act, including the Pricing Prospectus and, if so filed, any Issuer Free Writing Prospectus) has heretofore been filed by the Company with the Commission; and no stop order suspending the effectiveness of the Initial Registration Statement, any posteffective amendment thereto or the Rule 462(b) Registration Statement, if any, has been issued and no proceeding for that purpose has been initiated or, to the knowledge of the Company, threatened by the Commission (any preliminary prospectus included in the Initial Registration Statement or filed with the Commission pursuant to Rule 424(a) of the rules and regulations of the Commission under the Act is hereinafter called a "Preliminary Prospectus"; the various parts of the Initial Registration Statement and the Rule 462(b) Registration Statement, if any, including all exhibits thereto and including the information contained in the form of final prospectus filed with the Commission pursuant to Rule 424(b) under the Act in accordance with Section 5(a) hereof and deemed by virtue of Rule 430A under the Act to be part of the Initial Registration Statement at the time it was declared effective, each as amended at the time such part of the Initial Registration Statement became effective or such part of the Rule 462(b) Registration Statement, if any, became or hereafter becomes effective, are hereinafter collectively called the "Registration Statement"; the Preliminary Prospectus relating to the Shares that was included in the Registration Statement immediately prior to the Applicable Time (as defined in Section 1(a)(iii) hereof) is hereinafter called the "Pricing Prospectus"; such final prospectus, in the form first filed pursuant to Rule 424(b) under the Act, is hereinafter called the "Prospectus"; any "issuer free writing prospectus" as defined in Rule 433 under the Act relating to the Shares is hereinafter called an "Issuer Free Writing Prospectus"); and any oral or written communications undertaken in reliance on Section 5(d) of the Act are hereinafter called "Section 5(d) Communications";

(ii) No order preventing or suspending the use of any Preliminary Prospectus or any Issuer Free Writing Prospectus has been issued by the Commission, and each Preliminary Prospectus, at the time of filing thereof, conformed in all material respects to the requirements of the Act and the rules and regulations of the Commission thereunder, and did not contain an untrue statement of a material fact or omit to state a material fact necessary in order to make the statements therein, in the light of the

circumstances under which they were made, not misleading; provided, however, that this representation and warranty shall not apply to any statements or omissions made in reliance upon and in conformity with information furnished in writing to the Company by an Underwriter through Jefferies LLC and Piper Jaffray & Co. expressly for use therein;

(iii) For the purposes of this Agreement, the "Applicable Time" is : m (Eastern time) on the date of this Agreement. The Pricing Prospectus, as supplemented by those Issuer Free Writing Prospectuses, if any, and other information listed on Schedule II hereto, taken together (collectively, the "Pricing Disclosure Package"), as of the Applicable Time, did not include any untrue statement of a material fact or omit to state any material fact necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading; and each Issuer Free Writing Prospectus listed on Schedule II(a) hereto does not conflict with the information contained in the Registration Statement, the Pricing Prospectus or the Prospectus and each such Issuer Free Writing Prospectus, as supplemented by and taken together with the Pricing Prospectus as of the Applicable Time, did not include any untrue statement of a material fact or omit to state any material fact necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading; provided, however, that this representation and warranty shall not apply to statements or omissions made in any of the above documents in reliance upon and in conformity with information furnished in writing to the Company by an Underwriter through Jefferies LLC and Piper Jaffray & Co. expressly for use therein;

(iv) The Registration Statement conforms, and the Prospectus and any further amendments or supplements to the Registration Statement and the Prospectus will conform, in all material respects to the requirements of the Act and the rules and regulations of the Commission thereunder and do not and will not, as of the applicable effective date as to each part of the Registration Statement and as of the applicable filing date as to the Prospectus and any amendment or supplement thereto, contain an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading; provided, however, that this representation and warranty shall not apply to any statements or omissions made in reliance upon and in conformity with information furnished in writing to the Company by an Underwriter through Jefferies LLC and Piper Jaffray & Co. expressly for use therein;

(v) Since the date of the most recent financial statements of the Company included in the Pricing Prospectus, and except as described therein, (1) there has not been any change in the long term-debt or any material change in the capital stock of the Company or Dave & Buster's, Inc., a Missouri corporation and indirect wholly owned subsidiary of the Company ("Dave & Buster's"), or any dividend or distribution of any kind declared, set aside for payment, paid or made by the Company or Dave & Buster's on any class of capital stock, or any material adverse change, or any development involving a prospective material adverse change, in or affecting the business, properties, management, financial position or results of operations of the Company and the Company's subsidiaries taken as a whole; (2) none of the Company or any of the Company's subsidiaries taken as a whole, other than in the ordinary course of business, or incurred any liability or obligation, direct or contingent, that is material to the Company or any of the Company's subsidiaries taken as a whole, other than in the ordinary course of business; and (3) none of the Company or any of the Company's subsidiaries has sustained any material loss or interference with its business from fire, explosion, flood or other calamity, whether or not covered by insurance, or from any labor disturbance or dispute or any action, order or decree of any court or arbitrator or governmental or regulatory authority, except in each case as otherwise disclosed in the Pricing Prospectus;

(vi) Except as described in the Pricing Prospectus, the Company and the Company's subsidiaries have good and marketable title in fee simple to, or have valid rights to lease or otherwise use, all items of real and personal property that are material to the respective businesses of the Company and the Company's subsidiaries, in each case free and clear of all liens, encumbrances, claims and defects and imperfections of title except those that (1) are permitted under the Senior Secured Credit Facility (as referred to in the Preliminary Prospectus) together with any other documents, agreements or instruments delivered in connection therewith (the "Senior Credit Facility Documentation"), (2) do not materially and adversely affect the value of such property, (3) do not materially interfere with the use made and proposed to be made of such property by the Company and the Company's subsidiaries or (4) could not reasonably be expected, individually or in the aggregate, to have a material adverse effect on the business, properties, management, financial position or results of operations of the Company and the Company's subsidiaries taken as a whole (a "Material Adverse Effect");

(vii) The Company and each of the Company's subsidiaries have been duly organized and are validly existing and in good standing under the laws of their respective jurisdictions of organization, are duly qualified to do business and are in good standing in each jurisdiction in which their respective ownership or lease of property or the conduct of their respective businesses requires such qualification, and have all power and authority necessary to own or hold their respective properties and to conduct the businesses in which they are engaged, except where the failure to be so qualified, in good standing or have such power or authority would not, individually or in the aggregate, have a Material Adverse Effect. The Company does not own or control, directly or indirectly, any corporation, association or other entity other than the subsidiaries listed in Schedule IV to this Agreement;

(viii) The Company has an authorized capitalization as set forth in each of the Pricing Prospectus and the Prospectus under the heading "Description of Capital Stock"; and all the outstanding shares of capital stock or other equity interests of each subsidiary of the Company have been duly authorized and validly issued, are fully paid and non-assessable (except as otherwise described in each of the Pricing Prospectus and the Prospectus) and are owned directly or indirectly by the Company, free and clear of any lien, charge, encumbrance, security interest, restriction on voting or transfer or any other claim of any third party, except for those (i) created pursuant the Senior Credit Facility Documentation or (ii) disclosed in the Pricing Prospectus and the Prospectus;

(ix) The Shares to be issued and sold by the Company to the Underwriters hereunder have been duly authorized and, when issued and delivered against payment therefor as provided herein, will be validly issued, fully paid and non-assessable and will conform to the description of the Stock contained in the Prospectus;

(x) The Company has all power and authority to execute and deliver this Agreement, and to perform its obligations hereunder; and all action required to be taken for the due and proper authorization, execution and delivery of this Agreement and the consummation of the transactions contemplated hereby have been duly or validly taken by the Company;

(xi) The execution, delivery and performance by the Company of this Agreement, the issuance and sale of the Shares and compliance by the Company with the terms thereof and the consummation of the transactions contemplated by this Agreement will not (1) conflict with

or result in a breach or violation of any of the terms or provisions of, or constitute a default under, or result in the creation or imposition of any lien, charge or encumbrance upon any property or assets of the Company or any of the Company's subsidiaries pursuant to, any indenture, mortgage, deed of trust, loan agreement or other agreement or instrument to which the Company or any of the Company's subsidiaries is a party or by which the Company or any of the Company's subsidiaries is bound or to which any of the property or assets of the Company or any of the Company's subsidiaries is subject, (2) result in any violation of the provisions of the charter or by-laws or similar organizational documents of the Company or any of the Company's subsidiaries or (3) assuming the accuracy of, and the Underwriters' compliance with, the representations, warranties and agreements of the Underwriters herein, result in the violation of any applicable law or statute or any judgment, order, rule or regulation of any court or arbitrator or governmental or regulatory authority, except, in the case of clauses (1) and (3) above, for any such conflict, breach, violation or default that could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect;

(xii) Assuming the accuracy of, and the Underwriters' compliance with, the representations, warranties and agreements of the Underwriters herein, no consent, approval, authorization, order, registration or qualification of or with any court or arbitrator or governmental or regulatory authority is required for the execution, delivery and performance by the Company of this Agreement, the issuance and sale of the Shares and compliance by the Company with the terms thereof and the consummation of the transactions contemplated by this Agreement, except for such consents, approvals, authorizations, orders and registrations or qualifications (1) as have been obtained, (2) as may be required under the Act and applicable state securities laws, or (3) under the Conduct Rules of the Financial Industry Regulatory Authority ("FINRA") in connection with the purchase and resale of the Shares by the Underwriters;

(xiii) Neither the Company nor any of the Company's subsidiaries is (1) in violation of its charter or by-laws or similar organizational documents; (2) in default, and no event has occurred that, with notice or lapse of time or both, would constitute such a default, in the due performance or observance of any term, covenant or condition contained in any indenture, mortgage, deed of trust, loan agreement or other agreement or instrument to which the Company or any of the Company's subsidiaries is a party or by which the Company or any of the Company's subsidiaries is bound or to which any of the property or assets of the Company or any of the Company's subsidiaries is subject; or (3) in violation of any applicable law or statute or any judgment, order, rule or

regulation of any court or arbitrator or governmental or regulatory authority, except, in the case of clauses (2) and (3) above, for any such default or violation that could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect;

(xiv) The statements set forth in the Pricing Prospectus and Prospectus under the caption "Description of Capital Stock", insofar as they purport to constitute a summary of the terms of the Stock, under the caption "Certain Material United States Federal Income and Estate Tax Considerations", and under the caption "Underwriting", insofar as they purport to describe the provisions of the laws and documents referred to therein, are accurate, complete and fair;

(xv) Except as described in the Pricing Prospectus, there are no legal, governmental or regulatory investigations, actions, suits or proceedings pending, or to the best knowledge of the Company threatened, to which the Company or any of the Company's subsidiaries is or may be a party or to which any property of the Company or any of the Company's subsidiaries is or may be the subject of that, individually or in the aggregate, if determined adversely to the Company or any of the Company's subsidiaries, could reasonably be expected to have a Material Adverse Effect; and no such investigations, actions, suits or proceedings are threatened or, to the best knowledge (without having undertaken any independent inquiry) of the Company, contemplated by any governmental or regulatory authority or by others, except which, individually or in the aggregate, could not reasonably be expected to have a Material Adverse Effect;

(xvi) The Company is not and after giving effect to the offering and sale of the Shares and the application of the proceeds thereof as described in the Pricing Prospectus, the Company will not be, an "investment company" within the meaning of the Investment Company Act of 1940, as amended, and the rules and regulations of the Commission thereunder (collectively, the "Investment Company Act");

(xvii) At the time of filing the Initial Registration Statement the Company was not and is not an "ineligible issuer," as defined under Rule 405 under the Act;

(xviii) KPMG LLP, who have certified certain financial statements of the Company and its subsidiaries, are independent public accountants with respect to the Company and its subsidiaries within the applicable rules and regulations adopted by the Commission and the Public Company Accounting Oversight Board (United States) and as required by the Act;

(xix) Dave & Buster's and Dave & Buster's subsidiaries maintain systems of "internal control over financial reporting" (as defined in Rule 13a-15(f) of the Securities Exchange Act of 1934, as amended (the "Exchange Act")) that comply with the requirements of the Exchange Act and have been designed by, or under the supervision of, their respective principal executive and principal financial officers, or persons performing similar functions, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles. Dave & Buster's and Dave & Buster's subsidiaries maintain internal accounting controls sufficient to provide reasonable assurance that (1) transactions are executed in accordance with management's general or specific authorizations; (2) transactions are recorded as necessary to permit preparation of financial statements in conformity with generally accepted accounting principles and to maintain asset accountability; (3) access to assets is permitted only in accordance with management's general or specific authorization; and (4) the recorded accountability for assets is compared with the existing assets at reasonable intervals and appropriate action is taken with respect to any differences. Except as disclosed in the Pricing Prospectus, there are no material weaknesses or significant deficiencies in Dave & Buster's internal controls;

(xx) Except as disclosed in the Pricing Prospectus, since the date of the latest audited financial statements included in the Pricing Prospectus, there has been no change in the Company's internal control over financial reporting that has materially affected, or is reasonably likely to materially affect, the Company's internal control over financial reporting;

(xxi) Dave & Buster's and Dave & Buster's subsidiaries, on a consolidated basis, maintain an effective system of "disclosure controls and procedures" (as defined in Rule 13a-15(e) of the Exchange Act) that is designed to ensure that information required to be disclosed by Dave & Buster's in reports that it files or submits under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the Commission's rules and forms, including controls and procedures designed to ensure that such information is accumulated and communicated to Dave & Buster's management as appropriate to allow timely decisions regarding required disclosure. Dave & Buster's and Dave & Buster's subsidiaries, on a consolidated basis, have carried out evaluations of the effectiveness of their disclosure controls and procedures as required by Rule 13a-15 of the Exchange Act;

(xxii) The Company and the Company's subsidiaries own, possess or license adequate rights to use any patents, patent applications, trademarks, service marks, trade names, trademark registrations, service mark registrations, copyrights, licenses and knowhow (including trade secrets and other unpatented and/or unpatentable proprietary or confidential information, systems or procedures) reasonably necessary for the conduct of their respective businesses as described in the Pricing Prospectus, except where the failure to own or possess such rights could not reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect; and the conduct of their respective businesses will not conflict with any such rights of others, except which, individually or in the aggregate, could not reasonably be expected to have a Material Adverse Effect; and the Company and the Company's subsidiaries have not received any notice of any claim of infringement of or conflict with any such rights of others, which infringement or conflict, if the subject of an unfavorable decision, ruling or finding, would have a Material Adverse Effect.

(xxiii) Except as would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect, the Company and the Company's subsidiaries have paid all federal, state, local and foreign taxes and filed all tax returns required to be paid or filed through the date hereof; and except as otherwise disclosed in the Pricing Prospectus, there is no tax deficiency that has been, or could reasonably be expected to be, asserted against the Company or any of the Company's subsidiaries or any of their respective properties or assets except which, individually or in the aggregate, could not reasonably be expected to have a Material Adverse Effect.

(xxiv) The Company and the Company's subsidiaries possess any licenses, certificates, permits and other authorizations issued by, and have made any declarations and filings with, the appropriate federal, state, local or foreign governmental or regulatory authorities that are reasonably necessary for the ownership or lease of their respective properties or the conduct of their respective businesses as described in the Pricing Prospectus, except where the failure to possess or make the same could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect; and except as described in the Pricing Prospectus, neither the Company nor any of the Company's subsidiaries has received notice of any revocation or modification of any such license, certificate, permit or authorization will not be renewed in the ordinary course which, if the subject of an unfavorable decision, ruling, or finding would have a Material Adverse Effect;

(xxv) No labor disturbance by or dispute with employees of the Company or any of the Company's subsidiaries exists or, to the best knowledge (without having undertaken any independent inquiry) of the Company, is contemplated or to the best knowledge of the Company, is threatened and the Company is not aware of any existing or imminent labor disturbance by, or dispute with, the employees of any of the Company's or any of the Company's subsidiaries' principal suppliers, contractors or customers, except as would not have a Material Adverse Effect;

(xxvi)(1) The Company and the Company's subsidiaries (x) are, and at all prior times were, in compliance with any and all applicable federal, state, local and foreign laws, rules, regulations, requirements, decisions and orders relating to the protection of human health or safety, the environment, natural resources, hazardous or toxic substances or wastes, pollutants or contaminants (collectively, "Environmental Laws"), (y) have received and are in compliance with all permits, licenses, certificates or other authorizations or approvals required of them under applicable Environmental Laws to conduct their respective businesses, and (z) have not received notice of any actual or potential liability under or relating to any Environmental Laws, including for the investigation or remediation of any disposal or release of hazardous or toxic substances or wastes, pollutants or contaminants, and have no knowledge of any event or condition that would reasonably be expected to result in any such notice; and (2) there are no costs or liabilities associated with Environmental Laws of or relating to the Company or the Company's subsidiaries, except in the case of each of (1) and (2) above, for any such failure to comply, or failure to receive required permits, licenses or approvals, or cost or liability, as could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect; and (3) except as described in the Pricing Prospectus, (x) there are no proceedings that are pending, or that are known to be contemplated, against the Company or any of the Company's subsidiaries under any Environmental Laws in which a governmental entity is also a party, other than such proceedings regarding which it is reasonably believed no monetary sanctions of \$100,000 or more will be imposed, (y) the Company and the Company's subsidiaries are not aware of any issues regarding compliance with Environmental Laws, or liabilities or other obligations under Environmental Laws or concerning hazardous or toxic substances or wastes, pollutants or contaminants, that would have a Material Adverse Effect, and (z) none of the Company or the Company's subsidiaries anticipates material capital expenditures relating to any Environmental Laws;

(xxvii)(1) Each employee benefit plan, within the meaning of Section 3(3) of the Employee Retirement Income Security Act of 1974,

as amended ("ERISA"), for which the Company or any member of its "Controlled Group" (defined as any organization which is a member of a controlled group of corporations within the meaning of Section 414 of the Internal Revenue Code of 1986, as amended (the "Code")) would have any liability (each, a "Plan") has been maintained in compliance in all material respects with its terms and the requirements of any applicable statutes, orders, rules and regulations, including but not limited to ERISA and the Code; (2) no prohibited transaction, within the meaning of Section 406 of ERISA or Section 4975 of the Code, has occurred with respect to any Plan excluding transactions effected pursuant to a statutory or administrative exemption; (3) for each Plan that is subject to the funding rules of Section 412 of the Code or Section 302 of ERISA, no "accumulated funding deficiency" as defined in Section 412 of the Code, whether or not waived, has occurred or is reasonably expected to occur; (4) for each Plan that is subject to the funding rules of ERISA or the Code, the fair market value of the assets of each such Plan is not less than the present value of all benefits accrued under such Plan (determined based on those assumptions used to fund such Plan); (5) no "reportable event" (within the meaning of Section 4043(c) of ERISA) has occurred or is reasonably expected to occur; and (vi) neither the Company nor any member of the Controlled Group has incurred, nor reasonably expects to incur, any liability under Title IV of ERISA (other than contributions to the Plan or premiums to the Pension Benefit Guaranty Corporation (the "PBGC"), in the ordinary course and without default) in respect of a Plan (including a "multiemployer plan", within the meaning of Section 4001(a)(3) of ERISA);

(xxviii) The Company and the Company's subsidiaries have insurance in amounts and against such losses and risks as such party believes to be customary for companies engaged in similar business in similar industries and markets; and neither the Company nor any of the Company's subsidiaries has (1) received written notice from any insurer or agent of such insurer that capital improvements or other expenditures are required or necessary to be made in order to continue such insurance or (2) any reason to believe that it will not be able to renew its existing insurance coverage as and when such coverage expires or to obtain similar coverage from similar insurers as may be reasonably necessary to continue its business as described in the Pricing Prospectus and at a cost that could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect;

(xxix) Neither the Company nor any of the Company's subsidiaries nor, to the best knowledge of the Company, any director, officer, agent, employee or other person associated with or acting on behalf of the Company or any of the Company's subsidiaries has, in the

course of its actions for, or on behalf of, the Company (1) used any corporate funds for any unlawful contribution, gift, entertainment or other unlawful expense relating to political activity; (2) made any direct or indirect unlawful payment to any domestic government official, "foreign official" (as defined in the U.S. Foreign Corrupt Practices Act of 1977, as amended, and the rules and regulations thereunder (collectively, the "FCPA")) or employee from corporate funds; (3) violated or is in violation of any provision of the FCPA or any applicable non-U.S. anti-bribery statute or regulation; or (4) made any bribe, rebate, payoff, influence payment, kickback or other unlawful payment to any domestic government official, such foreign official or employee; and the Company and its subsidiaries and, to the knowledge of the Company, the Company's controlled affiliates have conducted their respective businesses in compliance with the FCPA and have instituted and maintain policies and procedures designed to ensure, and which are reasonably expected to continue to ensure, continued compliance therewith.

(xxx) The operations of the Company and the Company's subsidiaries are and have been conducted at all times in compliance with applicable financial recordkeeping and reporting requirements of the Currency and Foreign Transactions Reporting Act of 1970, as amended, the money laundering statutes of all jurisdictions, the rules and regulations thereunder and any related or similar rules, regulations or guidelines, issued, administered or enforced by any governmental agency (collectively, the "Money Laundering Laws") and no action, suit or proceeding by or before any court or governmental agency, authority or body or any arbitrator involving the Company or any of the Company's subsidiaries with respect to the Money Laundering Laws is pending or, to the best knowledge of the Company, threatened;

(xxxi) Neither the Company nor any of its subsidiaries nor, to the knowledge of the Company, after due inquiry, any director, officer, agent, employee, controlled affiliate or person acting on behalf of the Company or any of its subsidiaries is currently subject to any U.S. sanctions administered by the Office of Foreign Assets Control of the U.S. Treasury Department ("OFAC"); and the Company will not directly or indirectly use the proceeds of this offering, or lend, contribute or otherwise may available such proceeds to any subsidiary, or any joint venture partner or other person or entity, for the purpose of financing the activities of or business with any person, or in any country or territory, that currently is the subject to any U.S. sanctions administered by OFAC or in any other manner that will result in a violation by any person (including any person participating in the transaction whether as underwriter, advisor, investor or otherwise) of U.S. sanctions administered by OFAC.

(xxxii) Except as described in the Pricing Prospectus, no subsidiary of the Company is currently prohibited, directly or indirectly, under any agreement or other instrument to which it is a party or is subject, from paying any dividends to the Company, from making any other distribution on such subsidiary's capital stock, from repaying to the Company any loans or advances to such subsidiary from the Company or from transferring any of such subsidiary's properties or assets to the Company or any other subsidiary of the Company.

(xxxiii) Neither the Company nor any of the Company's subsidiaries is a party to any contract, agreement or understanding with any person (other than this Agreement) that would give rise to a valid claim against any of them or the Underwriters for a brokerage commission, finder's fee or like payment in connection with the offering and sale of the Shares;

(xxxiv) Neither the issuance, sale and delivery of the Shares nor the application of the proceeds thereof by the Company as described in the Pricing Prospectus will violate Regulation T, U or X of the Board of Governors of the Federal Reserve System or any other regulation of such Board of Governors;

(xxxv) No forward-looking statement (within the meaning of Section 27A of the Act and Section 21E of the Exchange Act) included in any of the Pricing Prospectus or the Prospectus has been made or reaffirmed without a reasonable basis or has been disclosed other than in good faith;

(xxxvi) Nothing has come to the attention of the Company that has caused the Company to believe that the statistical and market-related data included in each of the Pricing Prospectus and the Prospectus is not based on or derived from sources that are reliable and accurate in all material respects;

(xxxvii) To the Company's knowledge, there is and has been no failure on the part of Dave & Buster's or any of Dave & Buster's directors or officers, in their capacities as such, to comply with any provision of the Sarbanes-Oxley Act of 2002 and the rules and regulations promulgated in connection therewith, including Section 402 related to loans and Sections 302 and 906 related to certifications; and

(xxxviii) From the time of filing of the Initial Registration Statement (or, if earlier, the first date on which a Section 5(d) Communication was made) through the date hereof, the Company would be considered and is an "emerging growth company" as such term is defined in Section 2(a)(19) of the Act (an "Emerging Growth Company").

2. Subject to the terms and conditions herein set forth, (a) the Company agrees to issue and sell to each of the Underwriters, and each of the Underwriters agrees, severally and not jointly, to purchase from the Company, at a purchase price per share of \$[], the number of Firm Shares set forth opposite the name of such Underwriter in Schedule I hereto and (b) in the event and to the extent that the Underwriters shall exercise the election to purchase Optional Shares as provided below, the Company agrees to sell to each of the Underwriters, and each of the Underwriters agrees, severally and not jointly, to purchase from the Company agrees to sell to each of the Underwriters, and each of the Underwriters agrees, severally and not jointly, to purchase from the Company, at the purchase price per share set forth in clause (a) of this Section 2, that portion of the number of Optional Shares as to which such election shall have been exercised (to be adjusted by you so as to eliminate fractional shares) determined by multiplying such number of Optional Shares by a fraction, the numerator of which is the maximum number of Optional Shares which such Underwriter is entitled to purchase as set forth opposite the name of such Underwriter in Schedule I hereto and the denominator of which is the maximum number of Optional Shares that all of the Underwriters are entitled to purchase hereunder.

The Company hereby grants to the Underwriters the right to purchase at their election up to [] Optional Shares, at the purchase price per share set forth in the paragraph above, for the sole purpose of covering sales of shares in excess of the number of Firm Shares, provided that the purchase price per Optional Share shall be reduced by an amount per share equal to any dividends or distributions declared by the Company and payable on the Firm Shares but not payable on the Optional Shares. Any such election to purchase Optional Shares may be exercised only by written notice from you to the Company, given within a period of 30 calendar days after the date of this Agreement, setting forth the aggregate number of Optional Shares to be purchased and the date on which such Optional Shares are to be delivered, as determined by you but in no event earlier than the First Time of Delivery (as defined in Section 4 hereof) or, unless you and the Company otherwise agree in writing, earlier than two or later than ten business days after the date of such notice.

3. Upon the authorization by the Company of the release of the Firm Shares, the several Underwriters propose to offer the Firm Shares for sale upon the terms and conditions set forth in the Prospectus.

4.(a) The Shares to be purchased by each Underwriter hereunder, in definitive form, and in such authorized denominations and registered in such names as Jefferies LLC and Piper Jaffray & Co. may request upon at least forty-eight hours' prior notice to the Company, shall be delivered by or on behalf of the

Company to Jefferies LLC and Piper Jaffray & Co., through the facilities of the Depository Trust Company ("DTC"), for the account of such Underwriter, against payment by or on behalf of such Underwriter of the purchase price therefor by wire transfer of Federal (same-day) funds to the account specified by the Company to Jefferies LLC and Piper Jaffray & Co. at least forty-eight hours in advance. The Company will cause the certificates representing the Shares to be made available for checking and packaging at least twenty-four hours prior to the Time of Delivery (as defined below) with respect thereto at the office of DTC or its designated custodian (the "Designated Office"). The time and date of such delivery and payment shall be, with respect to the Firm Shares, [9:30 a.m.], New York City time, on [_____], 2014 or such other time and date as Jefferies LLC, Piper Jaffray & Co. and the Company may agree upon in writing, and, with respect to the Optional Shares, 9:30 a.m., New York time, on the date specified by Jefferies LLC and Piper Jaffray & Co. in the written notice given by Jefferies LLC and Piper Jaffray & Co. of the Underwriters' election to purchase such Optional Shares, or such other time and date as Jefferies LLC, Piper Jaffray & Co. and the Company may agree upon in writing. Such time and date for delivery of the Firm Shares is herein called the "First Time of Delivery", such time and date for delivery of the Optional Shares, if not the First Time of Delivery, is herein called the "Second Time of Delivery", and each such time and date for delivery is herein called a "Time of Delivery".

(b) The documents to be delivered at each Time of Delivery by or on behalf of the parties hereto pursuant to Section 8 hereof, including the cross receipt for the Shares and any additional documents requested by the Underwriters pursuant to Section 8(i) hereof, will be delivered at the offices of Latham & Watkins LLP, 885 Third Avenue, New York, New York 10022 (the "Closing Location"), and the Shares will be delivered at the Designated Office, all at such Time of Delivery. A meeting will be held at the Closing Location at [] p.m., New York City time, on the New York Business Day next preceding such Time of Delivery, at which meeting the final drafts of the documents to be delivered pursuant to the preceding sentence will be available for review by the parties hereto. For the purposes of this Section 4, "New York Business Day" shall mean each Monday, Tuesday, Wednesday, Thursday and Friday which is not a day on which banking institutions in New York City are generally authorized or obligated by law or executive order to close.

5. The Company agrees with each of the Underwriters:

(a) To prepare the Prospectus in a form approved by you and to file such Prospectus pursuant to Rule 424(b) under the Act not later than the Commission's close of business on the second business day following the execution and delivery of this Agreement, or, if applicable, such earlier time as may be required by Rule 430A(a)(3) under the Act; to make no further

amendment or any supplement to the Registration Statement or the Prospectus prior to the last Time of Delivery which shall be reasonably disapproved by you promptly after reasonable notice thereof; to advise you, promptly after it receives notice thereof, of the time when any amendment to the Registration Statement has been filed or becomes effective or any amendment or supplement to the Prospectus has been filed and to furnish you with copies thereof; to file promptly all material required to be filed by the Company with the Commission pursuant to Rule 433(d) under the Act; to advise you, promptly after it receives notice thereof, of the issuance by the Commission of any stop order or of any order preventing or suspending the use of any Preliminary Prospectus or other prospectus in respect of the Shares, of the suspension of the qualification of the Shares for offering or sale in any jurisdiction, of the initiation or threatening of any proceeding for any such purpose, or of any request by the Commission for the amending or supplementing of the Registration Statement or the Prospectus or for additional information; and, in the event of the issuance of any stop order or of any order preventing or suspending the use of any Preliminary Prospectus or other prospectus or suspending any such qualification, to promptly use its best efforts to obtain the withdrawal of such order;

(b) Promptly from time to time to take such action as you may reasonably request to qualify the Shares for offering and sale under the securities laws of such jurisdictions as you may reasonably request and to comply with such laws so as to permit the continuance of sales and dealings therein in such jurisdictions for as long as may be necessary to complete the distribution of the Shares, provided that in no event shall the Company be required to qualify to do business in any jurisdiction where it is not now so qualified, to take any action which would subject it to service of process in any jurisdiction where it is not now so subject;

(c) Prior to 10:00 a.m., New York City time, on the New York Business Day next succeeding the date of this Agreement and from time to time, to furnish the Underwriters with written and electronic copies of the Prospectus in New York City in such quantities as you may reasonably request, and, if the delivery of a prospectus (or in lieu thereof, the notice referred to in Rule 173(a) under the Act) is required at any time prior to the expiration of nine months after the time of issue of the Prospectus in connection with the offering or sale of the Shares and if at such time any event shall have occurred as a result of which the Prospectus as then amended or supplemented would include an untrue statement of a material fact or omit to state any material fact necessary in order to make the statements therein, in the light of the circumstances under which they were made when such Prospectus (or in lieu thereof, the notice referred to in Rule 173(a) under the Act) is delivered, not misleading, or, if for any other reason it shall be necessary during such same period to amend or supplement the Prospectus, to

notify you and upon your request to prepare and furnish without charge to each Underwriter and to any dealer in securities as many written and electronic copies as you may from time to time reasonably request of an amended Prospectus or a supplement to the Prospectus which will correct such statement or omission or effect such compliance; and in case any Underwriter is required to deliver a prospectus (or in lieu thereof, the notice referred to in Rule 173(a) under the Act) in connection with sales of any of the Shares at any time nine months or more after the time of issue of the Prospectus, upon your request but at the expense of such Underwriter, to prepare and deliver to such Underwriter as many written and electronic copies as you may request of an amended or supplemented Prospectus complying with Section 10(a)(3) of the Act;

(d) To make generally available to its securityholders as soon as practicable, but in any event not later than sixteen months after the effective date of the Registration Statement (as defined in Rule 158(c) under the Act), an earnings statement of the Company and its subsidiaries (which need not be audited) complying with Section 11(a) of the Act and the rules and regulations of the Commission thereunder (including, at the option of the Company, Rule 158);

(e)(i) During the period commencing on and including the date hereof and continuing through and including the 180th day following the date of the Prospectus (the "Lock-Up Period"), the Company will not (1) offer, sell, contract to sell, pledge, grant any option to purchase, make any short sale or otherwise transfer or dispose of, except as provided hereunder, any securities of the Company that are substantially similar to the Shares, including but not limited to any options or warrants to purchase shares of Stock or any securities that are convertible into or exchangeable for, or that represent the right to receive, Stock or any such substantially similar securities or publicly disclose the intention to make any offer, sale, pledge, disposition or filing or (2) enter into any swap or other agreement that transfers, in whole or in part, any of the economic consequences of ownership of the Stock or any such other securities, whether any such transaction described in clause (1) or (2) above is to be settled by delivery of Stock or such other securities, in cash or otherwise (other than (i) the issuance and sale of the Stock to be sold pursuant to this Agreement, (ii) pursuant to employee stock option plans existing on the date of this Agreement or otherwise disclosed in the Pricing Prospectus, or (iii) upon the exercise of an option or upon the exercise, conversion or exchange of exercisable convertible or exchangeable securities outstanding as of, the date of this Agreement), without the prior written consent of Jefferies LLC and Piper Jaffray & Co.

(ii) If Jefferies LLC and Piper Jaffray & Co, in their sole discretion, agrees to release or waive the restrictions set forth in a lock-up letter described in Section 8(i) hereof for an officer or director of the Company and provides the Company with notice of the impending release or waiver at least three business

days before the effective date of the release or waiver, the Company agrees to announce the impending release or waiver by a press release substantially in the form of Schedule III hereto through a major news service at least two business days before the effective date of the release or waiver.

(f) During a period of five years from the effective date of the Registration Statement, to furnish to you copies of all reports or other communications (financial or other) furnished to public stockholders generally and not filed on EDGAR, and to make generally available to Jefferies LLC and Piper Jaffray & Co. by filing on EDGAR, copies of any reports and financial statements furnished to or filed with the Commission or any national securities exchange on which any class of securities of the Company is listed;

(g) To use the net proceeds received by it from the sale of the Shares pursuant to this Agreement in the manner specified in the Pricing Prospectus under the caption "Use of Proceeds";

(h) To use its best efforts to list for quotation the Shares on The Nasdaq Stock Market LLC's NASDAQ Global Select Market ("NASDAQ");

(i) To file with the Commission such information on Form 10-Q or Form 10-K as may be required by Rule 463 under the Act;

(j) If the Company elects to rely upon Rule 462(b), the Company shall file a Rule 462(b) Registration Statement with the Commission in compliance with Rule 462(b) by 10:00 P.M., Washington, D.C. time, on the date of this Agreement, and the Company shall at the time of filing either pay to the Commission the filing fee for the Rule 462(b) Registration Statement or give irrevocable instructions for the payment of such fee pursuant to Rule 111 under the Act;

(k) Upon request of any Underwriter, to furnish, or cause to be furnished, to such Underwriter an electronic version of the Company's trademarks, servicemarks and corporate logo for use on the website, if any, operated by such Underwriter for the purpose of facilitating the lawful on-line offering of the Shares by such Underwriter (the "License"); provided, however, that the License shall be used solely for the purpose described above, is granted without any fee and may not be assigned or transferred; and

(1) To promptly notify you if the Company ceases to be an Emerging Growth Company at any time prior to the later of (i) completion of the distribution of the Shares within the meaning of the Act and (ii) completion of the 180-day restricted period referred to in Section 5(e) hereof.

6. (a) The Company represents and agrees that, without the prior consent of Jefferies LLC and Piper Jaffray & Co., it has not made and will not make any offer relating to the Shares that would constitute a "free writing prospectus" as defined in Rule 405 under the Act; each Underwriter represents and agrees that, without the prior consent of the Company, Jefferies LLC and Piper Jaffray & Co., it has not made and will not make any offer relating to the Shares that would constitute a free writing prospectus; any such free writing prospectus the use of which has been consented to by the Company, Jefferies LLC and Piper Jaffray & Co. is listed on Schedule II(a) hereto;

(b) The Company represents and agrees that it has not engaged in, or authorized any other person to engage in, any Section 5(d) Communications;

(c) The Company has complied and will comply with the requirements of Rule 433 under the Act applicable to any Issuer Free Writing Prospectus, including timely filing with the Commission or retention where required and legending; and the Company represents that it has satisfied and agrees that it will satisfy the conditions under Rule 433 under the Act to avoid a requirement to file with the Commission any electronic road show; and

(d) The Company agrees that if at any time following issuance of an Issuer Free Writing Prospectus any event occurred or occurs as a result of which such Issuer Free Writing Prospectus would conflict with the information in the Registration Statement, the Pricing Prospectus or the Prospectus or would include an untrue statement of a material fact or omit to state any material fact necessary in order to make the statements therein, in the light of the circumstances then prevailing, not misleading, the Company will give prompt notice thereof to Jefferies LLC and Piper Jaffray & Co. and, if requested by Jefferies LLC and Piper Jaffray & Co., will prepare and furnish without charge to each Underwriter an Issuer Free Writing Prospectus or other document which will correct such conflict, statement or omission; provided, however, that this representation and warranty shall not apply to any statements or omissions in an Issuer Free Writing Prospectus made in reliance upon and in conformity with information furnished in writing to the Company by an Underwriter through Jefferies LLC and Piper Jaffray & Co. expressly for use therein.

7. The Company covenants and agrees with the several Underwriters that the Company will pay or cause to be paid the following: (a) the fees, disbursements and expenses of the Company's counsel and accountants in connection with the registration of the Shares under the Act and all other expenses in connection with the preparation, printing, reproduction and filing of the Registration Statement, any Preliminary Prospectus, any Issuer Free Writing Prospectus and the Prospectus and amendments and supplements thereto and the mailing and delivering of copies thereof to the Underwriters and dealers;

(b) the cost of printing or producing any Agreement among Underwriters, this Agreement, the Blue Sky Memorandum, closing documents (including any compilations thereof) and any other documents in connection with the offering, purchase, sale and delivery of the Shares; (c) all expenses in connection with the qualification of the Shares for offering and sale under state securities laws as provided in Section 5(b) hereof, including the fees and disbursements of counsel for the Underwriters in connection with such qualification and in connection with the Blue Sky Memorandum; (d) all fees and expenses in connection with listing the Shares on NASDAQ; (e) the filing fees incident to, and the fees and disbursements of counsel for the Underwriters in connection with any required review by the FINRA of the terms of the sale of the Shares (such fees and disbursements of counsel for the Underwriters not to exceed \$\); (f) the cost of preparing stock certificates, if applicable; (g) the cost and charges of any transfer agent or registrar; (h) the cost and expenses of the Company relating to investor presentations on any "road show" undertaken in connection with the marketing of the Shares, including without limitation, expenses associated with the production of road show slides and graphics, and travel and lodging expenses of the representatives and officers of the Company (provided, however, that the Underwriters and the Company shall each pay 50% of the cost of any aircraft chartered in connection with any road show meetings) and (i) all other costs and expenses incident to the performance of its obligations hereunder which are not otherwise specifically provided for in this Section. It is understood, however, that, except as provided in this Section, and Sections 9 and 12 hereof, the Underwriters will pay all of their own costs and expenses, including the fees of their counsel, stock transfer taxes on resale of any of the Shares by them, and any advertising expenses connected with any offers they may make.

8. The obligations of the Underwriters hereunder, as to the Shares to be delivered at each Time of Delivery, shall be subject, in their discretion, to the condition that all representations and warranties and other statements of the Company herein are, at and as of such Time of Delivery, true and correct, the condition that the Company shall have performed all of its and their obligations hereunder theretofore to be performed, and the following additional conditions:

(a) The Prospectus shall have been filed with the Commission pursuant to Rule 424(b) under the Act within the applicable time period prescribed for such filing by the rules and regulations under the Act and in accordance with Section 5(a) hereof; all material required to be filed by the Company pursuant to Rule 433(d) under the Act shall have been filed with the Commission within the applicable time period prescribed for such filing by Rule 433; if the Company has elected to rely upon Rule 462(b) under the Act, the Rule 462(b) Registration Statement shall have become effective by 10:00 P.M., Washington, D.C. time, on the date of this Agreement; no stop order suspending the effectiveness of the Registration Statement or any part thereof shall have been instituted or threatened by the

Commission; no stop order suspending or preventing the use of the Prospectus or any Issuer Free Writing Prospectus shall have been initiated or threatened by the Commission; and all requests for additional information on the part of the Commission shall have been complied with to your reasonable satisfaction;

(b) Latham & Watkins LLP, counsel for the Underwriters, shall have furnished to you such written opinion letter and negative assurance letter, dated such Time of Delivery, in form and substance satisfactory to you, and such counsel shall have received such papers and information as they may reasonably request to enable them to pass upon such matters;

(c) Weil, Gotshal & Manges LLP, counsel for the Company, shall have furnished to you their written opinion letter and negative assurance letter, dated such Time of Delivery, in form and substance reasonably satisfactory to you (substantially in the form attached as Annex III hereto);

(d) On the date of the Prospectus at a time prior to the execution of this Agreement, at [9:30 a.m.], New York City time, on the effective date of any post-effective amendment to the Registration Statement filed subsequent to the date of this Agreement and also at each Time of Delivery, KPMG LLP shall have furnished to you a letter or letters, dated the respective dates of delivery thereof, in form and substance satisfactory to you, to the effect set forth in Annex I hereto (the executed copy of the letter delivered prior to the execution of this Agreement is attached as Annex I(a) hereto and a draft of the form of letter to be delivered on the effective date of any post-effective amendment to the Registration Statement and as of each Time of Delivery is attached as Annex I(b) hereto);

(e)(i) Neither the Company nor any of its subsidiaries shall have sustained since the date of the latest audited financial statements included in the Pricing Prospectus any loss or interference with its business from fire, explosion, flood or other calamity, whether or not covered by insurance, or from any labor dispute or court or governmental action, order or decree, otherwise than as set forth or contemplated in the Pricing Prospectus, and (ii) since the respective dates as of which information is given in the Pricing Prospectus and except as described therein there shall not have been any change in the capital stock or long-term debt of the Company or any of its subsidiaries or any change, or any development involving a prospective change, in or affecting the general affairs, management, financial position, stockholders' equity or results of operations of the Company and its subsidiaries, otherwise than as set forth or contemplated in the Pricing Prospectus, the effect of which, in any such case described in clause (i) or (ii), is in your judgment so material and adverse as to make it impracticable or inadvisable to proceed with the public offering or the delivery of the Shares being delivered at such Time of Delivery on the terms and in the manner contemplated in the Prospectus;

(f) On or after the Applicable Time (i) no downgrading shall have occurred in the rating accorded the Company's debt securities by any "nationally recognized statistical rating organization", as that term is defined by the Commission for purposes of Rule 436(g)(2) under the Act, and (ii) no such organization shall have publicly announced that it has under surveillance or review, with possible negative implications, its rating of any of the Company's debt securities;

(g) On or after the Applicable Time there shall not have occurred any of the following: (i) a suspension or material limitation in trading in securities generally on NASDAQ; (ii) a suspension or material limitation in trading in the Company's securities on NASDAQ; (iii) a general moratorium on commercial banking activities declared by either Federal or New York State authorities or a material disruption in commercial banking or securities settlement or clearance services in the United States; (iv) the outbreak or escalation of hostilities involving

the United States or the declaration by the United States of a national emergency or war or (v) the occurrence of any other calamity or crisis or any change in financial, political or economic conditions in the United States or elsewhere, if the effect of any such event specified in clause (iv) or (v) in your judgment makes it impracticable or inadvisable to proceed with the public offering or the delivery of the Shares being delivered at such Time of Delivery on the terms and in the manner contemplated in the Prospectus;

(h) The Shares to be sold at such Time of Delivery shall have been duly listed for quotation on NASDAQ; and

(i) The Company shall have obtained and delivered to the Underwriters executed copies of an agreement from each of its directors, executive officers and substantially all of its stockholders as set forth in Annex II in form and substance satisfactory to you;

(j) The Company shall have complied with the provisions of Section 5(c) hereof with respect to the furnishing of prospectuses on the New York Business Day next succeeding the date of this Agreement; and

(k) The Company shall have furnished or caused to be furnished to you at such Time of Delivery certificates of officers of the Company, satisfactory to you as to the accuracy of the representations and warranties of the Company, herein at and as of such Time of Delivery, as to the performance by the Company of all of its obligations hereunder to be performed at or prior to such Time of Delivery, as to the matters set forth in subsections (a) and (e) of this Section and as to such other matters as you may reasonably request.

9.(a) The Company will indemnify and hold harmless each Underwriter, its affiliates, directors, officers, employees, agents and each person, if any, who controls the Underwriter within the meaning of the Securities Act or the Exchange Act against any losses, claims, damages or liabilities, joint or several, to which such Underwriter may become subject, under the Act or otherwise, insofar as such losses, claims, damages or liabilities (or actions in respect thereof) arise out of or are based upon an untrue statement or alleged untrue statement of a material fact contained in the Registration Statement, any Preliminary Prospectus, the Pricing Prospectus or the Prospectus, or any amendment or supplement thereto, any Issuer Free Writing Prospectus or any "issuer information" filed or required to be filed pursuant to Rule 433(d) under the Act, or arise out of or are based upon the omission or alleged omission to state therein a material fact required to be stated therein or necessary to make the statements therein not misleading (in the case of the Preliminary Prospectus, or any amendment or supplement thereto, or any Issuer Free Writing Prospectus, in the light of the circumstances under which they were made) and will reimburse each Underwriter for any legal or other

expenses reasonably incurred by such Underwriter in connection with investigating or defending any such action or claim as such expenses are incurred; provided, however, that the Company shall not be liable to any Underwriter in any such case to the extent that any such loss, claim, damage or liability arises out of or is based upon an untrue statement or alleged untrue statement or omission or alleged omission relating to such Underwriter and made in the Registration Statement, any Preliminary Prospectus, the Pricing Prospectus or the Prospectus, or any amendment or supplement thereto, or any Issuer Free Writing Prospectus or any "issuer information" filed or required to be filed pursuant to Rule 433(d) under the Act, in reliance upon and in conformity with written information furnished to the Company by any Underwriter through Jefferies LLC and Piper Jaffray & Co. expressly for use therein, it being understood and agreed that the only such information consists of the information described in Section 9(b) below.

(b) Each Underwriter, severally and not jointly, will indemnify and hold harmless the Company, each of its directors, each of its officers who signed the Registration Statement, and each person, if any, who controls the Company within the meaning of the Securities Act or the Exchange Act against any losses, claims, damages or liabilities to which the Company may become subject, under the Act or otherwise, insofar as such losses, claims, damages or liabilities (or actions in respect thereof) arise out of or are based upon an untrue statement or alleged untrue statement of a material fact contained in the Registration Statement, any Preliminary Prospectus, the Pricing Prospectus or the Prospectus, or any amendment or supplement thereto, any Issuer Free Writing Prospectus or any "issuer information" filed or required to be filed pursuant to Rule 433(d) under the Act, or arise out of or are based upon the omission or alleged omission to state therein a material fact required to be stated therein or necessary to make the statements therein not misleading (in the case of the Preliminary Prospectus, or any amendment or supplement thereto, or any Issuer Free Writing Prospectus, in the light of the circumstances under which they were made), in each case to the extent, but only to the extent, that such untrue statement or alleged untrue statement or supplement thereto, or any Issuer information" filed or required to be filed pursuant to Rule 433(d) under the Act, any Issuer Free Writing Prospectus, in the light of the circumstances under which they were made), in each case to the extent, but only to the extent, that such untrue statement or alleged untrue statement or supplement thereto, or any Issuer Free Writing Prospectus, or any amendment or supplement thereto, or any "issuer information" filed or required to be filed pursuant to Rule 433(d) under the Act, in reliance upon and in conformity with written information furnished to the Company by such Underwriter through Jefferies LLC and Piper Jaffray & Co. expressly fo

Registration Statement, any Preliminary Prospectus, the Pricing Prospectus, any free writing prospectus that the Company has filed, or is required to file, pursuant to Rule 433(d) of the Securities Act, any Section 5(d) Communication or the Prospectus (or any amendment or supplement to the foregoing), are the statements set forth in (i) the first sentence of the third paragraph under the caption "Underwriting," (ii) the first paragraph under the caption "Underwriting," (ii) the first paragraph under the caption "Underwriting— Commission and Expenses," (iii) the statements concerning stabilizing transactions, syndicate covering transactions and penalty bids in accordance with Regulation M of the Exchange Act under the caption "Underwriting—Stabilization" and (iv) the statements concerning electronic prospectus distribution under the caption "Underwriting Prospectus and the Prospectus. The indemnity agreement set forth in this Section 9(b) shall be in addition to any liabilities that each Underwriter may otherwise have.

(c) Promptly after receipt by an indemnified party under subsection (a) or (b) above of notice of the commencement of any action, such indemnified party shall, if a claim in respect thereof is to be made against the indemnifying party under such subsection, notify the indemnifying party in writing of the commencement thereof; but the omission so to notify the indemnifying party shall not relieve it from any liability which it may have to any indemnified party otherwise than under such subsection. In case any such action shall be brought against any indemnified party and it shall notify the indemnifying party of the commencement thereof, the indemnifying party shall be entitled to participate therein and, to the extent that it shall wish, jointly with any other indemnifying party similarly notified, to assume the defense thereof, with counsel reasonably satisfactory to such indemnified party (who shall not, except with the consent of the indemnified party, be counsel to the indemnifying party), and, after notice from the indemnifying party to such indemnified party of its election so to assume the defense thereof, the indemnifying party shall not be liable to such indemnified party under such subsection for any legal expenses of other counsel or any other expenses, in each case subsequently incurred by such indemnified party, in connection with the defense thereof other than reasonable costs of investigation. It is understood that the indemnifying party or parties shall not, in connection with any one action or proceeding or separate but substantially similar actions or proceedings arising out of the same general allegations, be liable for the fees and expenses of more than one separate firm of attorneys at any time for all indemnified parties except to the extent that local counsel or counsel with specialized expertise (in addition to any regular counsel) is required to effectively defend against any such action or proceeding. No indemnifying party shall, without the written consent of the indemnified party, effect the settlement or compromise of, or consent to the entry of any judgment with respect to, any pending or threatened action or claim in respect of which indemnification or contribution may be sought hereunder (whether or not the indemnified party is an actual or potential party to such action or claim) unless such settlement,

compromise or judgment (i) includes an unconditional release of the indemnified party from all liability arising out of such action or claim and (ii) does not include a statement as to or an admission of fault, culpability or a failure to act, by or on behalf of any indemnified party.

(d) If the indemnification provided for in this Section 9 is unavailable to or insufficient to hold harmless an indemnified party under subsection (a) or (b) above in respect of any losses, claims, damages or liabilities (or actions in respect thereof) referred to therein, then each indemnifying party shall contribute to the amount paid or payable by such indemnified party as a result of such losses, claims, damages or liabilities (or actions in respect thereof) in such proportion as is appropriate to reflect the relative benefits received by the Company on the one hand and the Underwriters on the other from the offering of the Shares. If, however, the allocation provided by the immediately preceding sentence is not permitted by applicable law or if the indemnified party failed to give the notice required under subsection (c) above, then each indemnifying party shall contribute to such amount paid or payable by such indemnified party in such proportion as is appropriate to reflect not only such relative benefits but also the relative fault of the Company on the one hand and the Underwriters on the other in connection with the statements or omissions which resulted in such losses, claims, damages or liabilities (or actions in respect thereof), as well as any other relevant equitable considerations. The relative benefits received by the Company on the one hand and the Underwriters on the other shall be deemed to be in the same proportion as the total net proceeds from the offering (before deducting expenses) received by the Company bear to the total underwriting discounts and commissions received by the Underwriters, in each case as set forth in the table on the cover page of the Prospectus. The relative fault shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission or alleged omission to state a material fact relates to information supplied by the Company on the one hand or the Underwriters on the other and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such statement or omission. The Company and the Underwriters agree that it would not be just and equitable if contributions pursuant to this subsection (d) were determined by pro rata allocation (even if the Underwriters were treated as one entity for such purpose) or by any other method of allocation which does not take account of the equitable considerations referred to above in this subsection (d). The amount paid or payable by an indemnified party as a result of the losses, claims, damages or liabilities (or actions in respect thereof) referred to above in this subsection (d) shall be deemed to include any legal or other expenses reasonably incurred by such indemnified party in connection with investigating or defending any such action or claim. Notwithstanding the provisions of this subsection (d), no Underwriter shall be required to contribute any amount in excess of the amount by which the total price at which the Shares underwritten by it and distributed to the public were offered to

the public exceeds the amount of any damages which such Underwriter has otherwise been required to pay by reason of such untrue or alleged untrue statement or omission or alleged omission. No person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Act) shall be entitled to contribution from any person who was not guilty of such fraudulent misrepresentation. The Underwriters' obligations in this subsection (d) to contribute are several in proportion to their respective underwriting obligations and not joint.

(e) The obligations of the Company under this Section 9 shall be in addition to any liability which the Company may otherwise have and shall extend, upon the same terms and conditions, to each person, if any, who controls any Underwriter within the meaning of the Act and each broker-dealer affiliate of any Underwriter; and the obligations of the Underwriters under this Section 9 shall be in addition to any liability which the respective Underwriters may otherwise have and shall extend, upon the same terms and conditions, to each officer and director of the Company and to each person, if any, who controls the Company within the meaning of the Act.

10.(a) If any Underwriter shall default in its obligation to purchase the Shares which it has agreed to purchase hereunder at a Time of Delivery, you may in your discretion arrange for you or another party or other parties to purchase such Shares on the terms contained herein. If within thirty-six hours after such default by any Underwriter you do not arrange for the purchase of such Shares, then the Company shall be entitled to a further period of thirty-six hours within which to procure another party or other parties reasonably satisfactory to you to purchase such Shares on such terms. In the event that, within the respective prescribed periods, you notify the Company that you have so arranged for the purchase of such Shares, or the Company notifies you that it has so arranged for the purchase of such Shares, you or the Company shall have the right to postpone such Time of Delivery for a period of not more than seven days, in order to effect whatever changes may thereby be made necessary in the Registration Statement or the Prospectus, or in any other documents or arrangements, and the Company agrees to file promptly any amendments or supplements to the Registration Statement or the Prospectus which in your opinion may thereby be made necessary. The term "Underwriter" as used in this Agreement shall include any person substituted under this Section with like effect as if such person had originally been a party to this Agreement with respect to such Shares.

(b) If, after giving effect to any arrangements for the purchase of the Shares of a defaulting Underwriter or Underwriters by you and the Company as provided in subsection (a) above, the aggregate number of such Shares which remains unpurchased does not exceed one-eleventh of the aggregate number of all the Shares to be purchased at such Time of Delivery, then the Company shall have the right to require each non-defaulting Underwriter to purchase the number of Shares which such Underwriter agreed to purchase hereunder at such Time of

Delivery and, in addition, to require each non-defaulting Underwriter to purchase its pro rata share (based on the number of Shares which such Underwriter agreed to purchase hereunder) of the Shares of such defaulting Underwriter or Underwriters for which such arrangements have not been made; but nothing herein shall relieve a defaulting Underwriter from liability for its default.

(c) If, after giving effect to any arrangements for the purchase of the Shares of a defaulting Underwriter or Underwriters by you and the Company as provided in subsection (a) above, the aggregate number of such Shares which remains unpurchased exceeds one-eleventh of the aggregate number of all the Shares to be purchased at such Time of Delivery, or if the Company shall not exercise the right described in subsection (b) above to require non-defaulting Underwriters to purchase Shares of a defaulting Underwriter or Underwriters, then this Agreement (or, with respect to the Second Time of Delivery, the obligations of the Underwriters to purchase and of the Company to sell the Optional Shares) shall thereupon terminate, without liability on the part of any non-defaulting Underwriter or the Company except for the expenses to be borne by the Company and the Underwriters as provided in Section 7 hereof and the indemnity and contribution agreements in Section 9 hereof; but nothing herein shall relieve a defaulting Underwriter from liability for its default.

11. The respective indemnities, agreements, representations, warranties and other statements of the Company and the several Underwriters, as set forth in this Agreement or made by or on behalf of them, respectively, pursuant to this Agreement, shall remain in full force and effect, regardless of any investigation (or any statement as to the results thereof) made by or on behalf of any Underwriter or any controlling person of any Underwriter, or the Company, or any officer or director or controlling person of the Company, and shall survive delivery of and payment for the Shares.

12. If this Agreement shall be terminated pursuant to Section 10 hereof, the Company shall not then be under any liability to any Underwriter except as provided in Sections 7 and 9 hereof; but, if for any other reason, any Shares are not delivered by or on behalf of the Company as provided herein, the Company will reimburse the Underwriters through you for all out-of-pocket expenses approved in writing by you, including fees and disbursements of counsel, reasonably incurred by the Underwriters in making preparations for the purchase, sale and delivery of the Shares not so delivered, but the Company shall then be under no further liability to any Underwriter except as provided in Sections 7 and 9 hereof.

13. In all dealings hereunder, you shall act on behalf of each of the Underwriters, and the parties hereto shall be entitled to act and rely upon any statement, request, notice or agreement on behalf of any Underwriter made or given by you jointly or by Jefferies LLC and Piper Jaffray & Co. on behalf of you as the representatives.

All statements, requests, notices and agreements hereunder shall be in writing, and if to the Underwriters shall be delivered or sent by mail, telex or facsimile transmission to you as the representatives in care of Jefferies LLC, 520 Madison Avenue, New York, New York, 10022, Attention: General Counsel and Piper Jaffray & Co., 800 Nicollet Mall, Minneapolis, Minnesota 55402, Attention: General Counsel; and if to the Company shall be delivered or sent by mail, telex or facsimile transmission to the address of the Company set forth in the Registration Statement, Attention: Secretary, with a copy to Corey Chivers, Esq., Weil, Gotshal & Manges LLP, 767 Fifth Avenue, New York, New York 10153; provided, however, that any notice to an Underwriter pursuant to Section 9(c) hereof shall be delivered or sent by mail, telex or facsimile transmission to such Underwriter at its address set forth in its Underwriters' Questionnaire, or telex constituting such Questionnaire, which address will be supplied to the Company by you upon request; provided, however, that notices under subsection 5(e) shall be in writing, and if to the Underwriters shall be delivered or sent by mail, telex or facsimile transmission to you as representatives at Jefferies LLC, 520 Madison Avenue, New York, New York, 10022, Attention: General Counsel and Piper Jaffray & Co., 800 Nicollet Mall, Minneapolis, Minnesota 55402, Attention: General Counsel. Any such statements, requests, notices or agreements shall take effect upon receipt thereof.

In accordance with the requirements of the USA Patriot Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001)), the underwriters are required to obtain, verify and record information that identifies their respective clients, including the Company, which information may include the name and address of their respective clients, as well as other information that will allow the underwriters to properly identify their respective clients.

14. This Agreement shall be binding upon, and inure solely to the benefit of, the Underwriters, the Company, and, to the extent provided in Sections 9 and 11 hereof, the officers and directors of the Company and each person who controls the Company or any Underwriter, and their respective heirs, executors, administrators, successors and assigns, and no other person shall acquire or have any right under or by virtue of this Agreement. No purchaser of any of the Shares from any Underwriter shall be deemed a successor or assign by reason merely of such purchase.

15. Time shall be of the essence of this Agreement. As used herein, the term "business day" shall mean any day when the Commission's office in Washington, D.C. is open for business.

16. The Company acknowledges and agrees that (a) the purchase and sale of the Shares pursuant to this Agreement is an arm's-length commercial transaction between the Company, on the one hand, and the several Underwriters, on the other, (b) in connection therewith and with the process leading to such transaction each Underwriter is acting solely as a principal and not

the agent or fiduciary of the Company, (c) no Underwriter has assumed an advisory or fiduciary responsibility in favor of the Company with respect to the offering contemplated hereby or the process leading thereto (irrespective of whether such Underwriter has advised or is currently advising the Company on other matters) or any other obligation to the Company except the obligations expressly set forth in this Agreement and (d) the Company has consulted its own legal and financial advisors to the extent it deemed appropriate. The Company agrees that it will not claim that the Underwriters, or any of them, has rendered advisory services of any nature or respect, or owes a fiduciary or similar duty to the Company, in connection with such transaction or the process leading thereto.

17. This Agreement supersedes all prior agreements and understandings (whether written or oral) between the Company and the Underwriters, or any of them, with respect to the subject matter hereof.

18. THIS AGREEMENT AND ANY MATTERS RELATED TO THIS TRANSACTION SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK. The Company agrees that any suit or proceeding arising in respect of this agreement or our engagement will be tried exclusively in the U.S. District Court for the Southern District of New York or, if that court does not have subject matter jurisdiction, in any state court located in The City and County of New York and the Company agrees to submit to the jurisdiction of, and to venue in, such courts.

19. The Company and each of the Underwriters hereby irrevocably waives, to the fullest extent permitted by applicable law, any and all right to trial by jury in any legal proceeding arising out of or relating to this Agreement or the transactions contemplated hereby.

20. This Agreement may be executed by any one or more of the parties hereto in any number of counterparts, each of which shall be deemed to be an original, but all such counterparts shall together constitute one and the same instrument.

If the foregoing is in accordance with your understanding, please sign and return to us six counterparts hereof, and upon the acceptance hereof by you, on behalf of each of the Underwriters, this letter and such acceptance hereof shall constitute a binding agreement among each of the Underwriters and the Company. It is understood that your acceptance of this letter on behalf of each of the Underwriters is pursuant to the authority set forth in a form of Agreement among Underwriters, the form of which shall be submitted to the Company for examination upon request, but without warranty on your part as to the authority of the signers thereof.

Very truly yours,

DAVE & BUSTER'S ENTERTAINMENT, INC.

By: Name: Title:

Accepted as of the date hereof:

Jefferies LLC

By: Name: Title:

Piper Jaffray & Co.

By:

Name: Title:

On behalf of each of the Underwriters

SCHEDULE I

	Total Number of Firm Shares to be	Number of Optional Shares to be Purchased if Maximum Option
Underwriter Jefferies LLC	Purchased	Exercised
Piper Jaffray & Co.		
William Blair & Company, L.L.C.		
Raymond James & Associates Inc.		
Stifel, Nicolaus & Company, Incorporated		
LOYAL3 Securities, Inc.		
Total		
Total		

- (a) Issuer Free Writing Prospectuses:
- (b) Additional Documents Incorporated by Reference:

SCHEDULE III

[Form of Press Release]

Dave & Buster's Entertainment, Inc. [Date]

Dave & Buster's Entertainment, Inc. announced today that Jefferies LLC and Piper Jaffray & Co., the lead book-running managers in the Company's recent public sale of shares of common stock, are [waiving] [releasing] a lock-up restriction with respect to shares of the Company's common stock held by [certain officers or directors] [an officer or director] of the Company. The [waiver] [release] will take effect on , 201 , and the shares may be sold on or after such date.

This press release is not an offer for sale of the securities in the United States or in any other jurisdiction where such offer is prohibited, and such securities may not be offered or sold in the United States absent registration or an exemption from registration under the United States Securities Act of 1933, as amended.

SCHEDULE IV

D&B Leasing, Inc. D&B Marketing Company, LLC DANB Texas, Inc. Dave & Buster's I, L.P. Dave & Buster's, Inc. Dave & Buster's Holdings, Inc. Dave & Buster's Management Corporation, Inc. Dave & Buster's of California, Inc. Dave & Buster's of Colorado, Inc. Dave & Buster's of Connecticut, Inc. Dave & Buster's of Florida, Inc. Dave & Buster's of Georgia, Inc. Dave & Buster's of Hawaii, Inc. Dave & Buster's of Idaho, Inc. Dave & Buster's of Illinois, Inc. Dave & Buster's of Indiana, Inc. Dave & Buster's of Kansas, Inc. Dave & Buster's of Louisiana, Inc. Dave & Buster's of Maryland, Inc. Dave & Buster's of Massachusetts, Inc. Dave & Buster's of Nebraska, Inc. Dave & Buster's of New Mexico, Inc. Dave & Buster's of New York, Inc. Dave & Buster's of Oklahoma, Inc. Dave & Buster's of Oregon, Inc. Dave & Buster's of Pennsylvania, Inc. Dave & Buster's of Pittsburgh, Inc. Dave & Buster's of South Carolina, Inc. Dave & Buster's of Virginia, Inc. Dave & Buster's of Washington, Inc. Dave & Buster's of Wisconsin, Inc. Tango Acquisition, Inc. Tango License Corporation Tango of Arizona, Inc. Tango of Arundel, Inc. Tango of Farmingdale, Inc. Tango of Franklin, Inc. Tango of Houston, Inc. Tango of North Carolina, Inc. Tango of Tennessee, Inc. Tango of Westbury, Inc. 6131646 Canada, Inc.

SECOND AMENDED AND RESTATED CERTIFICATE

OF INCORPORATION OF

DAVE & BUSTER'S ENTERTAINMENT, INC.

Dave & Buster's Entertainment, Inc., a corporation organized and existing under the General Corporation Law of the State of Delaware (the "DGCL"), hereby certifies as follows:

1. The Corporation filed its original certificate of incorporation with the Secretary of State of the State of Delaware on April 28, 2010, under the name Games Acquisition Corp., filed an Amended and Restated Certificate of Incorporation on June 1, 2010 to change the name of the Corporation to Dave & Buster's Parent, Inc., filed a Certificate of Amendment on July 14, 2011 to change the name of the Corporation to Dave & Buster's Entertainment, Inc., and filed a Second Certificate of Amendment on [\times], 2014 to effectuate a stock split of its Common Stock.

2. This Second Amended and Restated Certificate of Incorporation, which restates, integrates and further amends the certificate of incorporation of the Corporation as heretofore amended and restated, has been duly adopted by the Corporation in accordance with Sections 242 and 245 of the DGCL and has been adopted by the requisite vote of the stockholders of the corporation, acting by written consent in lieu of a meeting in accordance with Section 228 of the DGCL.

3. The date of filing of this Second Amended and Restated Certificate of Incorporation is [X], 2014 (the "Effective Time").

4. The certificate of incorporation is hereby amended and restated in its entirety to read as follows:

ARTICLE I NAME

The name of the corporation is "Dave & Buster's Entertainment, Inc." (hereinafter called the "Corporation").

ARTICLE II REGISTERED OFFICE AND AGENT

The address of the registered office of the Corporation in the State of Delaware is 2711 Centerville Road, Suite 400, Wilmington, County of New Castle, State of Delaware, 19808. The name of the registered agent of the Corporation in the State of Delaware at such address is Corporation Service Company.

ARTICLE III PURPOSE

The purpose of the Corporation is to engage in any lawful act or activity for which corporations may be organized under the DGCL.

ARTICLE IV CAPITAL STOCK

(A) <u>Classes of Stock</u>. The total number of shares of all classes of capital stock that the Corporation is authorized to issue is 450,000,000 shares, which shall be divided into two classes of stock to be designated "<u>Common Stock</u>" and "<u>Preferred Stock</u>". The total number of shares of Common Stock that the Corporation is authorized to issue is 400,000,000 shares, par value \$0.01 per share. The total number of shares of Preferred Stock that the Corporation is authorized to issue is 50,000,000 shares, par value \$0.01 per share. The total number of shares of Preferred Stock, the number of authorized to issue is 50,000,000 shares, par value \$0.01 per share. Subject to the rights of the holders of any series of Preferred Stock, the number of authorized shares of either the Common Stock or Preferred Stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of a majority in voting power of the stock of the Corporation entitled to vote thereon irrespective of the provisions of Section 242(b)(2) of the DGCL, and no vote of the holders of either the Common Stock or Preferred Stock or Preferred Stock voting separately as a class shall be required therefor.

(B) <u>Common Stock</u>. The powers, preferences and relative, participating, optional or other special rights, and the qualifications, limitations and restrictions of the Common Stock, are as follows:

1. <u>Ranking</u>. The voting, dividend and liquidation rights of the holders of the Common Stock are subject to and qualified by such rights of the holders of the Preferred Stock of any series as may be designated by the Board of Directors of the Corporation (the "<u>Board</u>") upon any issuance of the Preferred Stock of any series.

2. Voting. Except as otherwise provided by law or by the resolution or resolutions providing for the issue of any series of Preferred Stock, the holders of outstanding shares of Common Stock shall have the exclusive right to vote for the election of directors and for all other purposes. Notwithstanding any other provision of this Second Amended and Restated Certificate of Incorporation (as the same may be further amended and/or restated from time to time, including the terms of any Preferred Stock Designation (as defined below), this "Certificate of Incorporation") to the contrary, the holders of Common Stock shall not be entitled to vote on any amendment to this Certificate of Incorporation (including any Preferred Stock Designation) that relates solely to the terms of one or more outstanding series of Preferred Stock if the holders of such affected series are entitled, either separately or together as a class with the holders of one or more other such series, to vote thereon pursuant to this Certificate of Incorporation (including any Preferred Stock Designation) or the DGCL. On each matter on which they are entitled to vote, the holders of the outstanding shares of Common Stock shall be entitled to one (1) vote for each share of Common Stock held by such stockholder.

3. <u>Dividends</u>. Subject to the rights of the holders of Preferred Stock, holders of shares of Common Stock shall be entitled to receive such dividends and other distributions in cash, stock or property of the Corporation when, as and if declared thereon by the Board from time to time out of assets or funds of the Corporation legally available therefor.

4. <u>Liquidation</u>. Subject to the rights of the holders of Preferred Stock, holders of shares of Common Stock shall be entitled to receive the assets and funds of the Corporation available for distribution to stockholders in the event of any liquidation, dissolution or winding up of the affairs of the Corporation, whether voluntary or involuntary. A liquidation, dissolution or winding up of the affairs of the Corporation, as such terms are used in this <u>Section (B)(4)</u>, shall not be deemed to be occasioned by or to include any consolidation or merger of the Corporation with or into any other person or a sale, lease, exchange or conveyance of all or a part of its assets.

(C) Preferred Stock.

Shares of Preferred Stock may be issued from time to time in one or more series. The Board is hereby authorized to provide by resolution or resolutions from time to time for the issuance, out of the unissued shares of Preferred Stock, of one or more series of Preferred Stock by filing a certificate pursuant to the DGCL (the "Preferred Stock Designation"), setting forth such resolution or resolutions and, with respect to each such series, establishing the number of shares to be included in such series, and fixing the voting powers, full or limited, or no voting power of the shares of such series, and the designation, preferences and relative, participating, optional or other special rights, if any, of the shares of each such series and any qualifications, limitations or restrictions thereof. The powers, designation, preferences and relative, participating, optional and other special rights of each series of Preferred Stock, and the qualifications, limitations and restrictions thereof, if any, may differ from those of any and all other series at any time outstanding. The authority of the Board with respect to each series of Preferred Stock shall include, but not be limited to, the determination of the following:

1. the designation of the series, which may be by distinguishing name, number, letter or title;

2. the number of shares of the series, which number the Board may thereafter (except where otherwise provided in the Preferred Stock Designation) increase or decrease (but not below the number of shares thereof then outstanding);

3. the rights in respect of any dividends (or methods of determining the dividends), if any, payable to the holders of the shares of such series, any conditions upon which such dividends shall be paid, the amounts or rates at which dividends, if any, will be payable on, and the preferences, if any, of shares of such series in respect of dividends, whether such dividends, if any, shall be cumulative or noncumulative and the date or dates upon which such dividends shall be payable;

4. the redemption rights and price or prices, if any, for shares of the series, the form of payment of such price or prices (which may be cash, property or rights, including securities of the Corporation or another corporation or entity) for which, the period or periods within which and the other terms and conditions upon which the shares of such series may be redeemed, in whole or in part, at the option of the Corporation or at the option of the holder or holders thereof or upon the happening of a specified event or events, if any, including the obligation, if any, of the Corporation to purchase or redeem shares of such series pursuant to a sinking fund or otherwise;

5. the amounts payable out of the assets of the Corporation on, and the preferences, if any, of shares of the series in the event of any voluntary or involuntary liquidation, dissolution or winding up of the affairs of the Corporation;

6. whether the shares of the series shall be convertible into or exchangeable for, shares of any other class or series, or any other security, of the Corporation or any other corporation, and, if so, the specification of such other class or series or such other security, the conversion or exchange price or prices or rate or rates, any adjustments thereof, the date or dates at which such shares shall be convertible or exchangeable and all other terms and conditions upon which such conversion or exchange may be made;

7. any restrictions on the issuance of shares of the same series or any other class or series;

8. the voting rights, if any, of the holders of shares of the series generally or upon specified events; and

9. any other powers, preferences and relative, participating, optional or other special rights of each series of Preferred Stock, and any qualifications, limitations or restrictions thereof, all as may be determined from time to time by the Board and stated in the resolution or resolutions providing for the issuance of such series of Preferred Stock.

Without limiting the generality of the foregoing, the resolutions providing for issuance of any series of Preferred Stock may provide that such series shall be superior or rank equally or be junior to any other series of Preferred Stock to the extent permitted by law.

ARTICLE V MANAGEMENT

This Article V is inserted for the management of the business and for the conduct of the affairs of the Corporation.

(A) General Powers. The business and affairs of the Corporation shall be managed by or under the direction of the Board, except as otherwise provided by law.

(B) Election of Directors; Number of Directors. At each annual meeting, directors shall be elected to hold office until the next annual meeting and until their successors have been duly elected and qualified; except that if any such election shall be not so held, such election shall take place at a stockholders' meeting called and held in accordance with the DGCL. At any meeting of stockholders at which directors are to be elected, directors shall be elected by the plurality vote of the votes cast by the holders of shares present or represented at the meeting and entitled to vote thereon. Subject to the rights of holders of any series of Preferred Stock to elect additional directors, the number of the directors of the Corporation shall be fixed from time to time by resolution of the Board.

(C) <u>Vacancies</u>. Subject to the rights of holders of any series of Preferred Stock to elect directors, any newly created directorship that results from an increase in the number of directors or any vacancy on the Board that results from the death, disability, resignation, disqualification or removal of any director or from any other cause shall be filled solely by the affirmative vote of a majority of the total number of directors then in office, even if less than a quorum, or by a sole remaining director and shall not be filled by the stockholders. Any director elected to fill a vacancy shall hold office for the remaining term of his or her predecessor.

(D) <u>Removal</u>. At any meeting called expressly for that purpose, any director may be removed, but only with Cause (defined below) by the affirmative vote of a majority of the remaining members of the Board or the holders of at least sixty-six and two-thirds percent (662/3%) of the then outstanding voting stock of the Corporation then entitled to vote on the election of directors, voting together as a single class.

"<u>Cause</u>" shall be deemed to exist only if: (i) such director has been indicted for or convicted of, has pleaded guilty or *nolo contendere* to, or such director is granted immunity to testify where another has been convicted of, a felony, (ii) such director has willfully failed to perform his duties, has been grossly negligent in the performance of his duties or has engaged in willful or serious misconduct in a matter that is injurious to the Corporation, in each case as determined by a court of competent jurisdiction or by the affirmative vote of at least a majority of the other members of the Board at any regular or special meeting of the Board called for such purpose, (iii) such director has been adjudicated by a court of competent jurisdiction to be mentally incompetent, which mental incompetency directly affects his ability to perform as a director of the Corporation, or (iv) such director has been found by a court of competent jurisdiction or by the other members of the Board at any regular or special meeting of the Board called for such purpose, (iii) such director of the Corporation, or (iv) such director has been found by a court of competent jurisdiction or by the affirmative vote of at least a majority of the other members of the Board at any regular or special meeting of the Board called for such purpose to have breached such director's duty of loyalty to the Corporation or its stockholders or to have engaged in any transaction with the Corporation from which such director derived an improper personal benefit. No director of the Corporation so removed may be nominated, re-elected or reinstated as a director of the Corporation so long as the cause for removal continues to exist.

(F) <u>Committees</u>. Pursuant to the Amended and Restated Bylaws of the Corporation (as the same may be amended and/or restated from time to time, the "<u>Bylaws</u>"), the Board may establish one or more committees to which may be delegated any or all of the powers and duties of the Board to the full extent permitted by law.

ARTICLE VI ELECTION OF DIRECTORS

5

Unless and except to the extent that the Bylaws shall so require, the election of directors of the Corporation need not be by written ballot.

ARTICLE VII EXCULPATION AND INDEMNIFICATION OF DIRECTORS

(A) <u>Limited Liability</u>. To the fullest extent permitted by the DGCL as the same exists or as may hereafter be amended, a director of the Corporation shall not be personally liable to the Corporation or its stockholders for monetary damages for breach of fiduciary duty as a director. No repeal or modification of this <u>Article VII</u> shall apply to or have any adverse effect on any right or protection of, or any limitation of the liability of, a director of the Corporation existing at the time of such repeal or modification.

(B) <u>Right to Indemnification</u>. The Corporation shall indemnify and hold harmless, to the fullest extent permitted by applicable law as it presently exists or may hereafter be amended, any person (a "<u>Covered Person</u>") who was or is made or is threatened to be made a party or is otherwise involved in any action, suit or proceeding, whether civil, criminal, administrative or investigative (a "<u>proceeding</u>"), by reason of the fact that he or she, or a person for whom he or she is the legal representative, is or was a director or officer of the Corporation, or has or had agreed to become a director or officer of the Corporation, or, while a director or officer of the Corporation, is or was serving at the request of the Corporation as a director, employee or agent of another corporation or of a limited liability company, partnership, joint venture, trust, enterprise or nonprofit entity, including service with respect to employee benefit plans, against all liability and loss suffered and expenses (including attorneys' fees) reasonably incurred by such Covered Person. Notwithstanding the preceding (or part thereof) commenced by such Covered Person only if the commencement of such proceeding (or part thereof) by the Covered Person only if the commencement of such proceeding (or part thereof) by the Covered Person only if the commencement of such proceeding (or part thereof) by the Covered Person only if the commencement of such proceeding (or part thereof) by the Board.

(C) <u>Prepayment of Expenses</u>. The Corporation shall to the fullest extent not prohibited by applicable law, as the same exists or may hereafter be amended, pay the expenses (including attorneys' fees) incurred by a Covered Person in defending any proceeding in advance of its final disposition, provided, however, that, to the extent required by law, such payment of expenses in advance of the final disposition of the proceeding shall be made only upon receipt of an undertaking by or on behalf of the Covered Person to repay all amounts advanced if it should be ultimately determined that the Covered Person is not entitled to be indemnified under this <u>Article VII</u> or otherwise.

(D) <u>Claims</u>. If a claim for indemnification (following the final disposition of such proceeding) or advancement of expenses under this <u>Article VII</u> is not paid in full within thirty (30) days after a written claim therefor by the Covered Person has been received by the Corporation, the Covered Person may file suit to recover the unpaid amount of such claim and, if successful in whole or in part, shall be entitled to be paid the expense (including attorneys' fees) of prosecuting such claim. In any such proceeding the Corporation shall have the burden of proving that the Covered Person is not entitled to the requested indemnification or advancement of expenses under applicable law.

(E) <u>Nonexclusivity of Rights</u>. The rights conferred on any Covered Person by this <u>Article VII</u> shall not be exclusive of any other rights which such Covered Person may have or hereafter acquire under any statute, provision of this Certificate of Incorporation, the Bylaws, written agreement approved by the Board, vote of stockholders or disinterested directors or otherwise.

(F) <u>Amendment or Repeal</u>. Any repeal or modification of the foregoing provisions of this <u>Article VII</u> shall not adversely affect any right or protection hereunder of any Covered Person in respect of any act or omission occurring prior to the time of such repeal or modification.

(G) <u>Other Indemnification and Prepayment of Expenses</u>. This <u>Article VII</u> shall not limit the right of the Corporation, to the extent and in the manner permitted by law, to indemnify and to advance expenses to persons other than Covered Persons when and as authorized by appropriate corporate action.

(H) The Corporation hereby acknowledges that any Covered Persons that are employees of the Oak Hill Funds, Oak Hill Capital Management, LLC or any of their respective affiliates (the "Oak Hill Covered Persons") have, or may in the future have, certain rights to indemnification, advancement of expenses and/or insurance provided by the Oak Hill Funds or other persons or entities that, directly or indirectly, (i) are controlled by, (ii) control, or (iii) are under common control with, the Oak Hill Funds (collectively, the "Other Indemnitors"). The Corporation hereby agrees (i) that it is the indemnitor of first resort (i.e., its obligations to the Oak Hill Covered Persons are primary and any obligation of the Other Indemnitors to advance expenses or to provide indemnification or to provide insurance for the same expenses or liabilities incurred by the Oak Hill Covered Persons are secondary), (ii) that it shall be required to advance the full amount of expenses incurred by the Oak Hill Covered Persons and shall be liable for the full amount of all expenses, judgments, penalties, fines and amounts paid in settlement to the extent legally permitted and as required by the terms of this Certificate of Incorporation and the Bylaws (or any other agreement between the Corporation and the Oak Hill Covered Persons), without regard to any rights the Oak Hill Covered Persons may have against the Other Indemnitors (whether pursuant to this Certificate of Incorporation, the Bylaws or any other agreement between the Corporation and the Oak Hill Covered Persons, or law), and (iii) that it irrevocably waives, relinquishes and releases the Other Indemnitors from any and all rights and claims against the Other Indemnitors for contribution, subrogation or any other recovery of any kind in respect thereof. Any insurance coverage provided, obtained or paid for by the Corporation, on the one hand, and any Other Indemnitor, on the other hand, shall be subject to the same primary and secondary liability hierarchy set forth in this section (H) of this Article VII. The Corporation shall use its reasonable best efforts to cause any insurance coverage policy contemplated by this section (H) of this Article VII and obtained by the Corporation to contain the same primary and secondary liability hierarchy as set forth in this section (H) of this Article VII. The Corporation further agrees that no advancement or payment by the Other Indemnitors on behalf of the Oak Hill Covered Persons with respect to any claim for which the Oak Hill Covered Persons have sought indemnification from the Corporation shall affect the foregoing and the Other Indemnitors shall have a right of contribution and/or be subrogated to the extent of such advancement or payment to all of the rights of recovery of the Oak Hill

Covered Persons against the Corporation. The Corporation and the Oak Hill Covered Persons agree that the Other Indemnitors are express third party beneficiaries of the terms of this section (H) of this Article VII.

For purposes of this Certificate of Incorporation, "<u>Oak Hill Funds</u>" shall mean, collectively, Oak Hill Capital Partners III, L.P., Oak Hill Capital Management Partners III, L.P., funds managed or advised by Oak Hill Capital Management, LLC and each of their respective affiliates (within the meaning of Section 12b-2 under the Securities Exchange Act of 1934, as amended (the "<u>Exchange Act</u>").

ARTICLE VIII STOCKHOLDER ACTION

(A) Any action required or permitted to be taken by the stockholders of the Corporation must be effected (i) at a duly called annual or special meeting of such holders or (ii) without a meeting by the written consent of such holders. Notwithstanding the foregoing, in the event the Oak Hill Funds collectively own less than forty percent (40.0%) of the then outstanding shares of the Common Stock, then any action required or permitted to be taken by the stockholders of the Corporation may not be effected by any written consent in lieu of a meeting by such stockholders; provided, however, that any action required or permitted to be taken by the holders of Preferred Stock, voting separately as a series or separately as a class with one or more other such series, may be taken without a meeting, without prior notice and without a vote, to the extent expressly so provided by the applicable Preferred Stock Designation.

(B) Except as otherwise required by law and subject to the rights of the holders of any series of Preferred Stock, special meetings of the stockholders of the Corporation for any purpose or purposes may be called at any time only (1) by or at the direction of the Board, (2) by or at the direction of the Chief Executive Officer of the Corporation or (3) by the Secretary of the Corporation upon the written request of the Oak Hill Funds or any person that acquires (other than through a registered public offering or through a broker's transaction executed on any securities exchange or other over-the-counter market) from the Oak Hill Funds ten percent (10%) or more of the issued and outstanding Common Stock, as applicable, provided that the Oak Hill Funds or such person that acquires (other than through a registered public offering or through a broker's transaction executed on any securities exchange or other over-the-counter market) from the Oak Hill Funds ten percent (10%) or more of the issued and outstanding Common Stock is a holder of ten percent (10%) or more of the issued and outstanding Common Stock is a holder of ten percent (10%) or more of the issued and outstanding or through a broker's transaction executed on any securities exchange or other over-the-counter market) from the Oak Hill Funds ten percent (10%) or more of the issued and outstanding Common Stock is a holder of ten percent (10%) or more of the issued and outstanding or through a broker's transaction executed on any securities exchange or other over-the-counter market). Except as provided in the preceding sentence, special meetings of the stockholders of the Corporation may not be called by any person or persons.

(C) Advance notice of stockholder nominations for election of directors and other business to be brought by stockholders before a meeting of stockholders shall be given in the manner provided by the Bylaws.

ARTICLE IX SECTION 203 OF THE DGCL

(A) The Corporation shall not be governed by Section 203 of the DGCL.

(B) Notwithstanding the foregoing, the Corporation shall not engage in any business combination (as defined below), at any point in time at which the Corporation's Common Stock is registered under Section 12(b) or 12(g) of the Exchange Act, with any interested stockholder (as defined below) for a period of three (3) years following the time that such stockholder became an interested stockholder, unless:

(1) prior to such time, the Board approved either the business combination or the transaction which resulted in the stockholder becoming an interested stockholder; or

(2) upon consummation of the transaction which resulted in the stockholder becoming an interested stockholder, the interested stockholder owned at least eighty-five percent (85%) of the voting stock (as defined below) of the Corporation outstanding at the time the transaction commenced, excluding for purposes of determining the voting stock outstanding (but not the outstanding voting stock owned by the interested stockholder) those shares owned by (1) persons who are directors and also officers and (2) employee stock plans in which employee participants do not have the right to determine confidentially whether shares held subject to the plan will be tendered in a tender or exchange offer; or

(3) at or subsequent to such time, the business combination is approved by the Board and authorized at an annual or special meeting of stockholders, and not by written consent, by the affirmative vote of at least sixty-six and two-thirds percent (66 2/3%) of the outstanding voting stock of the Corporation which is not owned by the interested stockholder.

(C) For purposes of this Article IX, references to:

(1) "affiliate" means a person that directly, or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, another person.

(2) "associate" when used to indicate a relationship with any person, means: (1) any corporation, partnership, unincorporated association or other entity of which such person is a director, officer or partner or is, directly or indirectly, the owner of twenty percent (20%) or more of any class of voting stock; (2) any trust or other estate in which such person has at least a twenty percent (20%) beneficial interest or as to which such person serves as trustee or in a similar fiduciary capacity; and (3) any relative or spouse of such person, or any relative of such spouse, who has the same residence as such person.

(3) "business combination" when used in reference to the Corporation and any interested stockholder of the Corporation, means:

(i) any merger or consolidation of the Corporation or any direct or indirect majority-owned subsidiary of the Corporation (i) with the interested stockholder, or (ii) with any other corporation, partnership, unincorporated association or other entity if the merger or consolidation is caused by the interested stockholder and as a result of such merger or consolidation the first paragraph of this <u>Article IX</u> is not applicable to the surviving entity;

(ii) any sale, lease, exchange, mortgage, pledge, transfer or other disposition (in one transaction or a series of transactions), except proportionately as a stockholder of the Corporation, to or with the interested stockholder, whether as part of a dissolution or otherwise, of assets of the Corporation or of any direct or indirect majority-owned subsidiary of the Corporation which assets have an aggregate market value equal to ten percent (10%) or more of either the aggregate market value of all the assets of the Corporation determined on a consolidated basis or the aggregate market value of all the outstanding stock of the Corporation;

(iii) any transaction which results in the issuance or transfer by the Corporation or by any direct or indirect majority-owned subsidiary of the Corporation of any stock of the Corporation or of such subsidiary to the interested stockholder, except: (a) pursuant to the exercise, exchange or conversion of the securities exercisable for, exchangeable for or convertible into stock of the Corporation or any subsidiary which securities were outstanding prior to the time that the interested stockholder became such; (b) pursuant to a merger under Section 251(g) of the DGCL; (c) pursuant to a dividend or distribution paid or made, or the exercise, exchange or conversion of securities exercisable for, exchangeable for or convertible for, exchangeable for or convertible into stock of the Corporation subsequent to a dividend or any such subsidiary which security is distributed, pro rata to all holders of a class or series of stock of the Corporation subsequent to the time the interested stockholder became such; (d) pursuant to an exchange offer by the Corporation to purchase stock made on the same terms to all holders of said stock; or (e) any issuance or transfer of stock by the Corporation; provided, however, that in no case under items (c)-(e) of this section (C)(3)(iii) of this <u>Article IX</u> shall there be an increase in the interested stockholder's proportionate share of the stock of any class or series of the Corporation or the voting stock of the Corporation (except as a result of immaterial changes due to fractional share adjustments);

(iv) any transaction involving the Corporation or any direct or indirect majority-owned subsidiary of the Corporation which has the effect, directly or indirectly, of increasing the proportionate share of the stock of any class or series, or securities convertible into the stock of any class or series, of the Corporation or of any such subsidiary which is owned by the interested stockholder, except as a result of immaterial changes due to fractional share adjustments or as a result of any purchase or redemption of any shares of stock not caused, directly or indirectly, by the interested stockholder; or

(v) any receipt by the interested stockholder of the benefit, directly or indirectly (except proportionately as a stockholder of the Corporation), of any loans, advances, guarantees, pledges, or other financial benefits (other than those expressly permitted in section (C)(3)(i)-(iv) above of this <u>Article IX</u>) provided by or through the Corporation or any direct or indirect majority-owned subsidiary.

(4) "<u>control</u>," including the terms "<u>controlling</u>," "controlled by" and "<u>under common control with</u>" means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a person, whether through the ownership of voting stock, by contract, or otherwise. A person who is the owner of twenty percent (20%) or more of the outstanding voting stock of any corporation, partnership, unincorporated association or other entity shall be presumed to have control of such entity, in the absence of proof by a preponderance of the evidence to the contrary. Notwithstanding the foregoing, a presumption of control shall not apply where such person holds voting stock, in good faith and not for the purpose of circumventing this <u>Article IX</u>, as an agent, bank, broker, nominee, custodian or trustee for one or more owners who do not individually or as a group have control of such entity.

(5) "interested stockholder" means any person (other than the Corporation or any direct or indirect majority-owned subsidiary of the Corporation) that (1) is the owner of fifteen percent (15%) or more of the outstanding voting stock of the Corporation, or (2) is an affiliate or associate of the Corporation and was the owner of fifteen percent (15%) or more of the outstanding voting stock of the Corporation at any time within the three (3) year period immediately prior to the date on which it is sought to be determined whether such person is an interested stockholder, and the affiliates and associates of such person; provided, however, that the term "interested stockholder" shall in no case include or be deemed to include, either of the Oak Hill Funds, any Oak Hill Direct Transferee, any Oak Hill Indirect Transferee or any of their respective affiliates or successors or any group, or any member of any such group, to which such persons are a party under Rule 13d-5 of the Exchange Act, or (2) any person whose ownership of share in excess of the fifteen percent (15%) limitation set forth herein is the result of any action taken solely by the Corporation; provided that such person specified in this clause (2) shall be an interested stockholder if thereafter such person. For the purpose of determining whether a person is an interested stockholder, the voting stock of the Corporation deemed to be outstanding shall include voting stock deemed to be owned by the person through application of the definition of "owner" below but shall not include any other unissued stock of the Corporation may be issuable pursuant to any agreement, arrangement or understanding, or upon exercise of conversion rights, warrants or options, or otherwise.

(6) "<u>Oak Hill Direct Transferee</u>" means any person that acquires (other than through a registered public offering or through a broker's transaction executed on any securities exchange or other over-the-counter market) directly from the Oak Hill Funds or any of their respective affiliates or successors or a "group", or any member of any such group, of which such persons are a party under Rule 13d-5 of the Exchange Act, beneficial ownership of five percent (5%) or more of the then outstanding voting stock of the Corporation; <u>provided</u> that such person was not an "interested stockholder" prior to such acquisition.

(7) "Oak Hill Indirect Transferee" means any person that acquires (other than through a registered public offering or through a broker's transaction executed on any securities exchange or other over-the-counter market) directly from any Oak Hill Direct Transferee or any other Oak Hill Indirect Transferee, beneficial ownership of five percent (5%) or more of the then outstanding voting stock of the Corporation; provided that such person was not an "interested stockholder" prior to such acquisition.

(8) "owner," including the terms "own" and "owned," when used with respect to any stock, means a person that individually or with or through any of its affiliates or associates:

(i) beneficially owns such stock, directly or indirectly; or

(ii) has (a) the right to acquire such stock (whether such right is exercisable immediately or only after the passage of time) pursuant to any agreement, arrangement or understanding, or upon the exercise of conversion rights, exchange rights, warrants, options, or

otherwise; provided, however, that a person shall not be deemed the owner of stock tendered pursuant to a tender or exchange offer made by such person or any of such person's affiliates or associates until such tendered stock is accepted for purchase or exchange; or (b) the right to vote such stock pursuant to any agreement, arrangement or understanding; provided, however, that a person shall not be deemed the owner of any stock because of such person's right to vote such stock if the agreement, arrangement or understanding to vote such stock arises solely from a revocable proxy or consent given in response to a proxy or consent solicitation made to ten (10) or more persons; or

(iii) has any agreement, arrangement or understanding for the purpose of acquiring, holding, voting (except voting pursuant to a revocable proxy or consent as described in item (b) of section (C)(8)(ii) above of this <u>Article IX</u>), or disposing of such stock with any other person that beneficially owns, or whose affiliates or associates beneficially own, directly or indirectly, such stock.

(9) "person" means any individual, corporation, partnership, or unincorporated association or other entity.

(10) "stock" means, with respect to any corporation, capital stock and, with respect to any other entity, any equity interest.

(11) "voting stock" means stock of any class or series entitled to vote generally in the election of directors.

ARTICLE X SEVERABILITY

If any provision or provisions (or any part thereof) of this Certificate of Incorporation shall be held to be invalid, illegal or unenforceable as applied to any circumstance for any reason whatsoever: (i) the validity, legality and enforceability of such provisions in any other circumstance and of the remaining provisions of this Certificate of Incorporation (including, without limitation, each portion of any paragraph of this Certificate of Incorporation containing any such provision held to be invalid, illegal or unenforceable that is not itself held to be invalid, illegal or unenforceable) shall not in any way be affected or impaired thereby and (ii) to the fullest extent possible, the provisions of this Certificate of Incorporation containing any such provision held to be invalid, illegal or unenforceable) shall be construed so as to permit the Corporation to protect its directors, officers, employees and agents from personal liability in respect of their good faith service or for the benefit of the Corporation to the fullest extent permitted by law.

ARTICLE XI AMENDMENT OF CERTIFICATE OF INCORPORATION

The Corporation reserves the right at any time and from time to time to amend, alter, change or repeal any provision contained in this Certificate of Incorporation, and any other provisions authorized by the DGCL may be added or inserted, in the manner now or hereafter prescribed by law; and all rights, preferences and privileges of whatsoever nature conferred upon

stockholders, directors or any other persons whomsoever by and pursuant to this Certificate of Incorporation in its present form or as hereafter amended are granted subject to the right reserved in this <u>Article XI</u>. Subject to applicable law, and subject to the rights of the holders of any series of Preferred Stock pursuant to any Preferred Stock Designation, the affirmative vote of the holders of a majority in voting power of the stock of the Corporation entitled to vote thereon shall be required to amend, alter, change or repeal any provision of this Certificate of Incorporation, or to adopt any new provision of this Certificate of Incorporation <u>provided</u>, <u>however</u>, notwithstanding anything to the contrary contained herein, <u>Article V</u> (Management), <u>Article VI</u> (Election of Directors), <u>Article VII</u> (Exculpation and Indemnification of Directors), <u>Article XII</u> (Stockholder Action), <u>Article IX</u> (Section 203 of the DGCL) <u>Article X</u> (Severability) <u>Article XII</u> (Amendment of Bylaws), <u>Article XIII</u> (Forum), <u>Article XIV</u> (Corporate Opportunities) and this <u>Article XI</u> (Amendment of Certificate of Incorporation) may not be amended, modified or repealed (or any provision adopted inconsistent with such provisions) by the stockholders without the affirmative vote of the holders of at least sixty-six and two-thirds percent (66 2/3%) of the voting power of the Corporation's then outstanding shares of stock entitled to vote thereon, voting together as a single class at any annual or special meeting of the stockholders in which notice of such amendment, (10%) or more of the issued and outstanding Common Stock is a holder of ten percent (10%) or more of the issued and Common Stock, any alteration, amendment or repeal, in whole or in part, of section (H) of <u>Article VII</u> (Exculpation and Indemnification of Directors) and Article VIII (Stockholder Action) shall require the affirmative approval of the Oak Hill Funds or any such acquirer, as applicable.

ARTICLE XII AMENDMENT OF BYLAWS

In furtherance and not in limitation of the powers conferred upon it by law, the Board is expressly authorized and empowered to adopt, amend and repeal the Bylaws by the affirmative vote of a majority of the total number of directors present at a regular or special meeting of the Board at which there is a quorum or by unanimous written consent. The Bylaws may also be amended, altered or repealed and new Bylaws may be adopted by the affirmative vote of the holders of at least sixty-six and two-thirds percent (66 2/3%) of the voting power of the Corporation's then outstanding shares entitled to vote generally in the election of directors, voting together as a single class.

ARTICLE XIII FORUM

Unless the Corporation consents in writing to the selection of an alternative forum, the Court of Chancery of the State of Delaware shall, to the fullest extent permitted by law, be the sole and exclusive forum for (1) any derivative action or proceeding brought on behalf of the Corporation, (2) any action asserting a claim of breach of a fiduciary duty owed by any director, officer or employee of the Corporation to the Corporation or the Corporation's stockholders, (3) any action asserting a claim arising pursuant to any provision of the DGCL, or (4) any action

asserting a claim governed by the internal affairs doctrine. To the fullest extent permitted by law, any person or entity purchasing or otherwise acquiring or holding any interest in shares of capital stock of the Corporation shall be deemed to have notice of and consented to the provisions of this <u>Article XIII</u>.

ARTICLE XIV CORPORATE OPPORTUNITIES

To the fullest extent permitted by Section 122(17) of the DGCL and except as may be otherwise expressly agreed in writing by the Corporation and the Oak Hill Funds, the Corporation, on behalf of itself and its subsidiaries, renounces any interest or expectancy of the Corporation and its subsidiaries in, or in being offered an opportunity to participate in, business opportunities, which are from time to time presented to the Oak Hill Funds or any of their respective managers, officers, directors, agents, stockholders, members, partners, affiliates and subsidiaries (other than the Corporation and its subsidiaries), even if the opportunity is one that the Corporation or its subsidiaries might reasonably be deemed to have pursued or had the ability or desire to pursue if granted the opportunity to do so, and no such person or entity shall be liable to the Corporation or any of its subsidiaries for breach of any fiduciary or other duty, as a director or officer or otherwise, by reason of the fact that such person or entity pursues or acquires such business opportunity, directs such business opportunity to another person or entity of fails to present such business opportunity, or information regarding such business opportunity, to the Corporation or its subsidiaries unless, in the case of any such person who is a director or officer of the Corporation. Any person or entity purchasing or otherwise expression or otherwise acquiring any interest in any shares of stock of the Corporation shall be deemed to have notice of and consented to the provisions of this <u>Article XIV</u>. Neither the alteration, amendment, addition to or repeal of this <u>Article XIV</u>, nor the adoption of any provision of this <u>Article XIV</u>, shall eliminate or reduce the effect of this <u>Article XIV</u>, would accrue or arise, prior to such alteration, amendment, addition, repeal or any other matter occurring, or any cause of action, suit or claim that, but for this <u>Article XIV</u>, would accrue or arise, prior to such alteration, amendment, addition, repeal

[remainder of page intentionally left blank]

IN WITNESS WHEREOF, this Second Amended and Restated Certificate of Incorporation has been executed on this [×] day of [×], 2014.

DAVE & BUSTER'S ENTERTAINMENT, INC.

By:

Name: Stephen M. King Title: Chief Executive Officer

SECOND AMENDED AND RESTATED

BYLAWS

OF

DAVE & BUSTER'S ENTERTAINMENT, INC.

		Page
ARTICLE I	STOCKHOLDERS	1
1.1	Place of Meetings	1
1.2	Annual Meeting	1
1.3	Special Meetings	1
1.4	Stockholder Action by Written Consent	1
1.5	Notice of Meetings	2
1.6	Voting List	2
1.7	Quorum	3
1.8	Adjournments	3
1.9	Proxies	4
1.10	Action at Meeting	4
1.11	Notice of Stockholder Business and Nominations	5
1.12	Conduct of Meetings	17
ARTICLE II	DIRECTORS	19
2.1	General Powers	19
2.2	Election, Number and Qualification	19
2.3	Chairman of the Board; Vice Chairman of the Board	20
2.4	Quorum	20
2.5	Action at Meeting	20
2.6	Removal	20
2.7	Vacancies	20
2.8	Resignation	21
2.9	Regular Meetings	21
2.10	Special Meetings	21
2.11	Notice of Special Meetings	21
2.12	Meetings by Conference Communications Equipment	21
2.13	Action by Written Consent	22
2.14	Committees	22
2.15	Compensation of Directors	22
ARTICLE III	OFFICERS	23
3.1	Titles	23

i

Table of Contents (continued)

		Page
3.2	Election	23
3.3	Qualification	23
3.4	Tenure	23
3.5	Resignation and Removal	23
3.6	Vacancies	23
3.7	Chief Executive Officer	24
3.8	President	24
3.9	Chief Financial Officer	24
3.10	Vice Presidents	24
3.11	Secretary and Assistant Secretaries	24
3.12	Treasurer and Assistant Treasurers	25
3.13	Delegation of Authority	25
ARTICLE IV	CAPITAL STOCK	25
4.1	Issuance of Stock	25
4.2	Stock Certificates; Uncertificated Shares	26
4.3	Transfers	27
4.4	Lost, Stolen or Destroyed Certificates	27
ARTICLE V	GENERAL PROVISIONS	27
5.1	Fiscal Year	27
5.2	Corporate Seal	27
5.3	Waiver of Notice	28
5.4	Voting of Securities	28
5.5	Evidence of Authority	28
5.6	Certificate of Incorporation	28
5.7	Severability	28
5.8	Pronouns	29
5.9	Electronic Transmission	29
ARTICLE VI	AMENDMENTS	29
ARTICLE VII	INDEMNIFICATION AND ADVANCEMENT	30
7.1	Right to Indemnification	30
7.2	Prepayment of Expenses	30

ii

Table of Contents (continued)

		Page
7.3	Authorization of Indemnification	30
7.4	Good Faith Defined	31
7.5	Right of Claimant to Bring Suit	32
7.6	Nonexclusivity of Indemnification and Advancement of Expenses	32
7.7	Insurance	32
7.8	Certain Definitions	33
7.9	Survival of Indemnification and Advancement of Expenses	33
7.10	Contract Rights	34
7.11	Primary Obligation	34

iii

ARTICLE I

STOCKHOLDERS

1.1 <u>Place of Meetings</u>. All meetings of stockholders shall be held at such place, if any, as may be designated from time to time by the Board of Directors (the "<u>Board</u>") of Dave & Buster's Entertainment, Inc. (the "<u>Corporation</u>"), the Chairman of the Board or the Chief Executive Officer or, if not so designated, at the principal office of the Corporation. The Board may, in its sole discretion, determine that a meeting shall not be held at any place, but may instead be held solely by means of remote communication in accordance with Section 211(a) of the General Corporation Law of the State of Delaware (the "<u>DGCL</u>").

1.2 <u>Annual Meeting</u>. The annual meeting of stockholders for the election of directors and for the transaction of such other business as may properly be brought before the meeting shall be held on a date and at a time designated by the Board (which date shall not be a legal holiday in the place, if any, where the meeting is to be held). The Board may postpone, reschedule or cancel any previously scheduled annual meeting of stockholders.

1.3 Special Meetings. Special meetings of stockholders for any purpose or purposes may be called as (and only as) provided in the Certificate of Incorporation. The Board may postpone, reschedule or cancel any previously scheduled special meeting of stockholders, other than one called at the request of Oak Hill Capital Partners III, L.P., Oak Hill Capital Management Partners III, L.P., funds managed or advised by Oak Hill Capital Management, LLC and each of their respective affiliates (collectively, the "Oak Hill Funds"), or any person that acquires (other than through a registered public offering or through a broker's transaction executed on any securities exchange or over-the-counter market) from the Oak Hill Funds ten percent (10%) or more of the issued and outstanding Common Stock, as applicable. Business transacted at any special meeting of stockholders shall be limited to matters relating to the purpose or purposes stated in the notice of meeting.

1.4 <u>Stockholder Action by Written Consent</u>. Any action required or permitted to be taken by the stockholders of the Corporation must be effected (i) at a duly called annual or special meeting of such holders or (ii) without a meeting by the written consent of such holders. Notwithstanding the foregoing, in the event the Oak Hill Funds collectively own less than forty

percent (40.0%) of the then outstanding shares of the Common Stock, then any action required or permitted to be taken by the stockholders of the Corporation may not be effected by any written consent in lieu of a meeting by such stockholders; provided, however, that any action required or permitted to be taken by the holders of Preferred Stock, voting separately as a series or separately as a class with one or more other such series, may be taken without a meeting, without prior notice and without a vote, to the extent expressly so provided by the applicable Preferred Stock Designation

1.5 Notice of Meetings. Except as otherwise provided by the DGCL, notice of each meeting of stockholders, whether annual or special, shall be given not less than ten (10) nor more than sixty (60) days before the date of the meeting to each stockholder entitled to vote at such meeting as of the record date for determining the stockholders entitled to notice of the meeting. Without limiting the manner by which notice otherwise may be given to stockholders, any notice shall be effective if given by a form of electronic transmission consented to (in a manner consistent with the DGCL) by the stockholder to whom the notice is given. The notices of all meetings shall state the place, if any, date and time of the meeting, the means of remote communications, if any, by which stockholders and proxyholders may be deemed to be present in person and vote at such meeting, and the record date for determining that is different from the record date for stockholders entitled to notice of a special meeting shall state, in addition, the purpose or purposes for which the meeting is called. If notice is given by mail, such notice shall be deemed given when deposited in the United States mail, postage prepaid, directed to the stockholder at such stockholder's address as it appears on the records of the Corporation. If notice is given by electronic transmission, such notice shall be deemed given at the time specified in Section 232 of the DGCL.

1.6 <u>Voting List</u>. The Secretary shall prepare, at least ten (10) days before every meeting of stockholders, a complete list of the stockholders entitled to vote at the meeting (<u>provided</u>, <u>however</u>, if the record date for determining the stockholders entitled to vote is less than ten (10) days before the date of the meeting, the list shall reflect the stockholders entitled to vote as of the tenth (10th) day before the meeting date), arranged in alphabetical order, and showing the address of each stockholder and the number of shares registered in the name of each

stockholder. Such list shall be open to the examination of any stockholder, for any purpose germane to the meeting, for a period of at least ten (10) days prior to the meeting: (a) on a reasonably accessible electronic network, <u>provided</u> that the information required to gain access to such list is provided with the notice of the meeting, or (b) during ordinary business hours, at the principal place of business of the Corporation. If the meeting is to be held at a place, then the list shall also be produced and kept at the time and place of the meeting during the whole time thereof, and may be inspected by any stockholder who is present. If the meeting is to be held solely by means of remote communication, then the list shall also be open to the examination of any stockholder during the whole time of the meeting on a reasonably accessible electronic network, and the information required to access such list shall be provided with the notice of the meeting. Except as otherwise provided by law, the list shall presumptively determine the identity of the stockholders entitled to vote at the meeting and the number of shares held by each of them.

1.7 Quorum. Except as otherwise provided by law, the Certificate of Incorporation or these Bylaws, the holders of a majority in voting power of the shares of the capital stock of the Corporation issued and outstanding and entitled to vote at the meeting, present in person, present by means of remote communication in a manner, if any, authorized by the Board in its sole discretion, or represented by proxy, shall constitute a quorum for the transaction of business; provided, however, that where a separate vote by a class or classes or series of capital stock is required by law or the Certificate of Incorporation, the holders of a majority in voting power of the shares of such class or classes or series of the capital stock of the Corporation issued and outstanding and entitled to vote on such matter, present in person, present by means of remote communication in a manner, if any, authorized by the Board in its sole discretion, or represented by proxy, shall constitute a quorum entitled to take action with respect to the vote on such matter. A quorum, once established at a meeting, shall not be broken by the withdrawal of enough votes to leave less than a quorum.

1.8 <u>Adjournments</u>. Any meeting of stockholders, annual or special, may be adjourned from time to time to any other time and to any other place at which a meeting of stockholders may be held under these Bylaws by the chairman of the meeting or, if directed to be voted on by the chairman of the meeting, by the stockholders present or represented at the meeting and

entitled to vote thereon, although less than a quorum. If the adjournment is for more than thirty (30) days, a notice of the adjourned meeting shall be given to each stockholder of record entitled to vote at the meeting. If after the adjournment a new record date for determination of stockholders entitled to vote is fixed for the adjourned meeting, the Board shall fix as the record date for determining stockholders entitled to notice of such adjourned meeting the same or an earlier date as that fixed for determination of stockholders entitled to vote at the adjourned meeting, and shall give notice of the adjourned meeting to each stockholder of record as of the record date so fixed for notice of such adjourned meeting. At the adjourned meeting, the Corporation may transact any business which might have been transacted at the original meeting.

1.9 <u>Proxies</u>. Each stockholder of record entitled to vote at a meeting of stockholders may vote in person (including by means of remote communication in a manner, if any, authorized by the Board in its sole discretion by which stockholders may be deemed to be present in person and vote at such meeting) or may authorize another person or persons to vote for such stockholder by a proxy executed or transmitted in a manner permitted by applicable law. No such proxy shall be voted upon after three (3) years from the date of its execution, unless the proxy expressly provides for a longer period.

1.10 Action at Meeting. When a quorum is present at any meeting, any matter to be voted upon by the stockholders at such meeting shall, except as set forth in Section 2.2, be decided by the vote of the holders of shares of stock having a majority in voting power of the votes cast by the holders of all of the shares of stock present or represented at the meeting and voting affirmatively or negatively on such matter (or if there are two (2) or more classes or series of stock entitled to vote as separate classes, then in the case of each such class or series, the holders of shares of stock having a majority in voting power of the votes cast by the holders of all of the shares of stock of that class or series present or represented at the meeting and voting affirmatively or negatively on such matter), except when a different vote is required by applicable law, regulation applicable to the Corporation or its securities, the rules or regulations of any stock exchange applicable to the Corporation, the Certificate of Incorporation or these Bylaws. Voting at meetings of stockholders need not be by written ballot.

1.11 Notice of Stockholder Business and Nominations.

(A) Annual Meetings of Stockholders.

(1) Nominations of persons for election to the Board and the proposal of other business to be considered by the stockholders may be made at an annual meeting of stockholders only (a) pursuant to the Corporation's notice of meeting (or any supplement thereto), (b) by or at the direction of the Board or any committee thereof or (c) by any stockholder of the Corporation who was a stockholder of record of the Corporation at the time the notice provided for in this Section 1.11 is delivered to the Secretary of the Corporation, who is entitled to vote at the meeting and who complies with the notice procedures set forth in this Section 1.11. For the avoidance of doubt, the procedures set forth in this Section 1.11 shall be the exclusive means for a stockholder to make nominations or submit proposals for other business (other than matters properly brought under Rule 14a-8 under the Securities Exchange Act of 1934, as amended (the "Exchange Act") or any successor rule thereto and included in the Corporation's proxy statement that has been prepared to solicit proxies for such annual meeting) for an annual meeting of stockholders.

(2) For any nominations or any other business to be properly brought before an annual meeting by a stockholder, the stockholder must have given timely notice thereof in writing to the Secretary of the Corporation and any such proposed business (other than the nominations of persons for election to the Board) must constitute a proper matter for stockholder action. To be timely, a stockholder's notice shall be delivered to the Secretary at the principal executive offices of the Corporation not later than the close of business on the ninetieth (90th) day, nor earlier than the close of business on the one hundred and twentieth (120th) day, prior to the first anniversary of the preceding year's annual meeting (provided, however, that (a) in the case of the annual meeting of stockholders of the Corporation to be held in 2015, or (b) in the event that the date of the annual meeting is more than thirty (30) days before or more than ninety (90) days after such anniversary date, notice by the stockholder to be timely must be so delivered not earlier than the close of business on the one hundred twentieth (120th) day prior to such annual meeting and not later than the close of business on the later of the ninetieth (90th) day prior to such annual meeting and not later than the close of business on the later of the ninetieth (90th) day prior to such annual meeting and not later than the close of business on the later of the ninetieth (90th) day prior to such annual meeting and not later than the close of business on the later of the ninetieth (90th) day prior to such annual meeting and not later than the close of business on the later of the ninetieth (90th) day prior to such annual meeting and not later than the close of business on the later of the ninetieth (90th) day prior to such annual meeting and not later than the close of business on the later of the ninetieth (90th) day following the day on which public

announcement of the date of such annual meeting is first made by the Corporation). In no event shall the adjournment or postponement of an annual meeting (or any public announcement thereof) commence a new time period (or extend any time period) for the giving of a stockholder's notice as described above. To be in proper form, such stockholder's notice (whether given pursuant to this paragraph (A)(2) of Section 1.11 or paragraph (B) of Section 1.11) to the Secretary of the Corporation shall set forth:

- (a) as to each person, if any, whom the stockholder proposes to nominate for election or reelection as a director to the Board
 - all information relating to such person that is required to be disclosed, whether in a proxy statement, other filings required to be made in connection with solicitations of proxies for election of directors in a contested election contest, or as otherwise required, in each case pursuant to and in accordance with Section 14(a) of the Exchange Act, and the rules and regulations promulgated thereunder;
 - (ii) such person's written consent to being named in the proxy statement as a nominee and to serving as a director if elected; a written questionnaire with respect to the background and qualification of such person and the background of any other person or entity on whose behalf the nomination is being made (which questionnaire shall be provided by the Secretary of the Corporation upon written request); such person's written representation and agreement (in the form provided by the Secretary of the Corporation upon written request), (A) that such person is not and will not become party to any agreement, arrangement or understanding with, and has not given any commitment or assurance to, any person or entity as to how such person, if elected as a director of the Corporation, will

act or vote on any issue or question (a "<u>Voting Commitment</u>") that has not been disclosed to the Corporation or any Voting Commitment that could limit or interfere with such person's ability to comply, if elected as a director of the Corporation, with such person's fiduciary duties under applicable law, (B) that such person is not and will not become a party to any agreement, arrangement, or understanding with any person or entity other than the Corporation with respect to any direct or indirect compensation, reimbursement, or indemnification in connection with service or action as a director that has not been disclosed to the Corporation, and (C) that, in such person's individual capacity and on behalf of any person or entity on whose behalf the nomination is being made, such person would, if elected as a director, comply with all of the Corporation's corporate governance, ethics, conflict of interest, confidentiality and stock ownership and trading policies and guidelines applicable generally to the Corporation's directors and, if elected as a director of the Corporation, such person currently would be in compliance with any such policies and guidelines that have been publicly disclosed;

(iii) a description of all direct and indirect compensation and other material monetary agreements, arrangements and understandings during the past three (3) years, and any other material relationships, between or among the stockholder making the nomination and any beneficial owner on whose behalf the nomination is made, and their respective affiliates and associates, or any other person or persons (including their names) acting in concert therewith, on the one hand, and each proposed nominee, and his or her respective affiliates or associates, or any other person or persons (including their names) acting in concert therewith, on the other hand, including, without limitation, all information that would be required to be disclosed pursuant to Item 404 promulgated under Regulation S-K if the stockholder making the nomination and any beneficial owner on whose behalf the nomination is made, if any, or any affiliate or associate thereof or person acting in concert therewith, were the "registrant" for purposes of such rule and the nominee were a director or executive officer of such registrant;

- (iv) any information that such person would be required to disclose pursuant to clauses (ii) (ix) of clause (c) of this paragraph (A)(2) of Section 1.11 if such person were a stockholder purporting to make a nomination or propose business pursuant thereto; and
- (v) an undertaking to notify the Corporation in writing of any change in the information called for by clauses (i) (iv) as of the record date for notice of such meeting, by notice received by the Secretary of the Corporation at the principal executive offices of the Corporation not later than the tenth (10th) day following such record date;
- (b) as to any other business that the stockholder proposes to bring before the meeting,
 - (i) a brief description of the business desired to be brought before the annual meeting, the text of the proposal or business (including the complete text of any resolutions proposed for consideration and in the event that such business includes a proposal to amend any Corporation

document, the language of the proposed amendment), the reasons for conducting such business at the annual meeting and any material interest in such business of such stockholder and the beneficial owner, if any, on whose behalf the proposal is made; and

- (ii) a description of all agreements, arrangements and understandings between such stockholder and beneficial owner, if any, and their respective affiliate and associates, and any other person or persons (including their names) acting in concert therewith in connection with the proposal of such business by such stockholder; and
- (c) as to the stockholder giving the notice, the beneficial owner, if any, on whose behalf the nomination or proposal for other business is made, any of their respective affiliates or associates (including, if such stockholder or beneficial owner is an entity, as to each director, executive, managing member or control person of such entity), and any others acting in concert with any of the following:
 - (i) the name and address of such stockholder, as they appear on the Corporation's books, of such beneficial owner, if any, any of their respective affiliates or associates, and any others acting in concert with any of the foregoing;
 - the class or series and number of shares of capital stock of the Corporation which are, directly or indirectly, owned beneficially and of record by such stockholder, such beneficial owner, if any, any of their respective affiliates or associates, and any others acting in concert with any of the foregoing;

- (iii) a description of any agreement, arrangement or understanding with respect to the nomination or proposal between or among such stockholder and/or such beneficial owner, any of their respective affiliates or associates, and any others acting in concert with any of the foregoing, including, in the case of a nomination, the nominee;
- (iv) a description of any agreement, arrangement or understanding (including any derivative or short positions, profit interests, options, warrants, convertible securities, stock appreciation or similar rights with an exercise or conversion privilege or a settlement payment or mechanism at a price related to any class or series of capital stock of the Corporation or with a value derived in whole or in part from the value of any class or series of capital stock of the Corporation, hedging transactions, and borrowed or loaned shares) that has been entered into as of the date of the stockholder's notice by, or on behalf of, such stockholder and such beneficial owners, if any, any of their respective affiliates or associates, and any others acting in concert with any of the foregoing, whether or not such instrument or right shall be subject to settlement in underlying shares of capital stock of the Corporation (a "Derivative Instrument");
- a description of any proxy, contract, arrangement, understanding or relationship pursuant to which such stockholder and such beneficial owner, if any, any of their respective affiliates or associates, and any others acting in concert with any of the foregoing, has the right to vote any shares of any security of the Corporation;

¹⁰

- (vi) any short interest of such stockholder and such beneficial owner, if any, any of their respective affiliates or associates, and any others acting in concert with any of the foregoing, in any security of the Corporation (for purposes of these Bylaws, a person shall be deemed to have a short interest in a security if such person directly or indirectly, through any contract, arrangement, understanding, relationship or otherwise, has the opportunity to profit or share in any profit derived from any decrease in the value of the subject security);
- (vii) any rights to dividends on the shares of the Corporation owned beneficially by such stockholder and such beneficial owner, if any, any of their respective affiliates or associates, and any others acting in concert with any of the foregoing, that are separated or separable from the underlying shares of capital stock of the Corporation;
- (viii) any proportionate interest in shares of capital stock of the Corporation or Derivative Instruments, held, directly or indirectly, by a general or limited partnership or similar entity in which such stockholder or such beneficial owner, if any, any of their respective affiliates or associates, and any others acting in concert with any of the foregoing, is a general partner or, directly or indirectly, beneficially owns an interest in a general partner, is the manager, managing member or directly or indirectly beneficially owns an interest in the manager or managing member of a limited liability company or similar entity;
- (ix) any performance related fees (other than an asset-based fee) that such stockholder and such beneficial owner, if any, any

of their respective affiliates or associates, and any others acting in concert with any of the foregoing, is entitled to based on any increase or decrease in the value of shares of capital stock of the Corporation or Derivative Instruments, if any;

- a representation that the stockholder is a holder of record of stock of the Corporation entitled to vote at such annual meeting and intends to appear in person or by proxy at the annual meeting to propose such business or nomination;
- (xi) a representation whether the stockholder or the beneficial owner, if any, any of their respective affiliates or associates, and any others acting in concert with any of the foregoing, intends or is part of a group which intends (a) to deliver a proxy statement and/or form of proxy to holders of at least the percentage of the Corporation's outstanding capital stock required to approve or adopt the proposal or elect the nominee and/or (b) otherwise to solicit proxies or votes from stockholders in support of such proposal or nomination;
- (xii) any other information relating to such stockholder and beneficial owner, if any, any of their respective affiliates or associates, and any others acting in concert with any of the foregoing, required to be disclosed under the DGCL or in a proxy statement or other filings required to be made in connection with solicitations of proxies for, as applicable, the proposal of other business and/or for the election of directors in an election contest pursuant to and in accordance with Section 14(a) of the Exchange Act and the rules and regulations promulgated thereunder; and

(xiii) an undertaking by the stockholder and beneficial owner, if any, to notify the Corporation in writing of any change in the information called for by clauses (i) – (xii) above as of the record date for such meeting, by notice received by the Secretary of the Corporation at the principal executive offices of the Corporation not later than the tenth (10th) day following such record date.

The Corporation may, as a condition of any such nomination being deemed properly brought before an annual meeting, require any proposed nominee to furnish (i) any information required pursuant to any undertaking delivered pursuant to this paragraph (A)(2) of Section 1.11, and (ii) such other information as the Corporation may require to determine the eligibility of such proposed nominee to serve as a director, or independent director of the Corporation may request. The foregoing notice requirements of this paragraph (A) of this Section 1.11 shall be deemed satisfied by a stockholder with respect to business (other than any purported nomination) if the stockholder has notified the Corporation of his, her or its intention to present a proposal at an annual meeting in compliance with Rule 14a-8 promulgated under the Exchange Act or any successor rule thereto and such stockholder's proposal has been included in a proxy statement that has been prepared by the Corporation to solicit proxies for such annual meeting.

(3) Notwithstanding anything in the second sentence of paragraph (A)(2) of this Section 1.11 to the contrary, in the event that the number of directors to be elected to the Board at the annual meeting is increased effective after the time period for which nominations would otherwise be due under paragraph (A)(2) of this Section 1.11 and there is no public announcement by the Corporation naming the nominees for the additional directorships at least one hundred (100) days prior to the first anniversary of the preceding year's annual meeting, a stockholder's notice required by this Section 1.11 shall also be considered timely, but only with respect to nominees for the additional directorship positions, if it shall be delivered to the Secretary of the Corporation at the principal executive offices of the Corporation not later than the close of business on the tenth (10th) day following the day on which such public announcement is first made by the Corporation.

(B) Special Meetings of Stockholders. Only such business shall be conducted at a special meeting of stockholders as shall have been brought before the special meeting pursuant to the Corporation's notice of meeting. Nominations of persons for election to the Board may be made at a special meeting of stockholders at which directors are to be elected pursuant to the Corporation's notice of meeting (1) by or at the direction of the Board or any committee thereof or (2) provided that the Board pursuant to Section 1.3 hereof has determined that directors shall be elected at such meeting, by any stockholder of the Corporation who (a) is a stockholder of record at the time the notice provided for in this Section 1.11 is delivered to the Secretary of the Corporation and at the time of the special meeting, (b) is entitled to vote at the special meeting and (iii) complies with the notice procedures and conditions set forth in this Section 1.11 (including the information requirements in paragraph (A)(2) of Section 1.11) as to such nomination. For the avoidance of doubt, the foregoing clause (2) of this paragraph (B) of Section 1.11 shall be the exclusive means for a stockholder to propose nominations of persons for election to the Board at a special meeting of stockholders at which directors are to be elected. In the event the Corporation calls a special meeting of stockholders for the purpose of electing one or more directors to the Board, any such stockholder entitled to vote in such election of directors may nominate a person or persons (as the case may be) for election to such position(s) as specified in the Corporation's notice of meeting, if the stockholder's notice required by paragraph (A) (2) of this Section 1.11 shall be delivered to the Secretary of the Corporation at the principal executive offices of the Corporation not earlier than the close of business on the one hundred twentieth (120th) day prior to such special meeting and not later than the close of business on the later of the ninetieth (90th) day prior to such special meeting and the tenth (10th) day following the day on which public announcement is first made of the date of the special meeting and of the nominees proposed by the Board to be elected at such meeting. In no event shall the adjournment or postponement of a special meeting as to which notice has been sent to stockholders, or any public announcement with respect thereto, commence a new time period (or extend any time period) for the giving of a stockholder's notice as described above.

(C) <u>General</u>. (1) Except as otherwise expressly provided in any applicable rule or regulation promulgated under the Exchange Act, only such persons who are nominated in accordance with the procedures set forth in this Section 1.11 shall be eligible to be properly elected at an annual or special meeting of stockholders of the Corporation to serve as directors

and only such business shall be conducted at a meeting of stockholders as shall have been brought before the meeting in accordance with the procedures set forth in this Section 1.11. Except as otherwise provided by law, the Certificate of Incorporation or the Bylaws, the chairman of the meeting shall have the power and duty (a) to determine whether a nomination or any other business proposed to be brought before the meeting was made or proposed, as the case may be, in accordance with the procedures set forth in this Section 1.11 (including whether the stockholder or beneficial owner, if any, on whose behalf the nomination or proposal is made, solicited (or is part of a group which solicited) or did not so solicit, as the case may be, proxies or votes in support of such stockholder's nominee or proposal in compliance with such stockholder's representation as required by clause (A)(2)(c)(xi) of this Section 1.11) and (b) if any proposed nomination or business was not made or proposed in compliance with this Section 1.11, to declare that such nomination shall be disregarded or that such proposed business shall not be transacted. Notwithstanding the foregoing provisions of this Section 1.11, unless otherwise required by law, if the stockholder (or a qualified representative of the stockholder) does not appear at the annual or special meeting of stockholders of the Corporation to present a nomination or proposed business, such nomination shall be disregarded and such proposed business shall not be transacted. Notwithstanding the fore purposes of this Section 1.11, to be considered a qualified representative of the stockholder, a person must be a duly authorized officer, manager or partner of such stockholder or must be authorized by a writing executed by such stockholder or an electronic transmission delivered by such stockholder, to act for such stockholder as proxy at the annual or special meeting of stockholders and such person must produce such writing or electronic transmission, or a reliable reproduction of th

(2) For purposes of this Section 1.11, "public announcement" shall include disclosure in a press release reported by the Dow Jones News Service, Associated Press or other national news service or in a document publicly filed by the Corporation with the Securities and Exchange Commission pursuant to Section 13, 14 or 15(d) of the Exchange Act and the rules and regulations promulgated thereunder.

(3) Notwithstanding the foregoing provisions of this Section 1.11, a stockholder shall also comply with all applicable requirements of the Exchange Act and the rules and regulations promulgated thereunder with respect to the matters set forth in this Section 1.11; provided however, that any references in these Bylaws to the Exchange Act or the rules and regulations promulgated thereunder are not intended to and shall not limit any requirements applicable to nominations or proposals as to any other business to be considered pursuant to this Section 1.11 (including paragraphs (A)(2) and (B) hereof), and compliance with paragraphs (A)(2) and (B) of this Section 1.11 shall be the exclusive means for a stockholder to make nominations or submit other business (other than, as provided in the penultimate sentence of (A)(2), business other than nominations brought properly under and in compliance with Rule 14a-8 of the Exchange Act or any successor rule thereto, as may be amended from time to time). Nothing in this Section 1.11 shall be deemed to affect any rights (a) of stockholders to request inclusion of proposals or nominations in the Corporation's proxy statement pursuant to applicable Rule 14a-8 promulgated under the Exchange Act or (b) of the holders of any series of Preferred Stock to elect directors if and to the extent provided for under any applicable provisions of the Certificate of Incorporation.

(D) <u>Record Date</u>. In order that the Corporation may determine the stockholders entitled to notice of any meeting of stockholders or any adjournment thereof, the Board may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted by the Board, and which record date shall, unless otherwise required by law, not be more than sixty (60) nor less than ten (10) days before the date of such meeting. If the Board so fixes a date, such date shall also be the record date for determining the stockholders entitled to vote at such meeting unless the Board determines, at the time it fixes such record date for determining stockholders entitled to notice of or to vote at a meeting of stockholders shall be at the close of business on the day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the day next preceding the day on which the meeting is held. A determination of stockholders of record entitled to notice of or to vote at a meeting of stockholders shall apply to any adjournment of the meeting; provided, however, that the Board may fix a new record date for determination of stockholders entitled to vote at the adjourned

meeting, and in such case shall also fix as the record date for stockholders entitled to notice of such adjourned meeting the same or an earlier date as that fixed for determination of stockholders entitled to vote in accordance herewith at the adjourned meeting.

In order that the Corporation may determine the stockholders entitled to receive payment of any dividend or other distribution or allotment of any rights, or entitled to exercise any rights in respect of any change, conversion or exchange of stock or for the purpose of any other lawful action, the Board may fix a record date, which shall not be more than sixty (60) days prior to such action. If no such record date is fixed, the record date for determining stockholders for any such purpose shall be at the close of business on the day on which the Board adopts the resolution relating thereto.

In order that the Corporation may determine the stockholders entitled to consent to corporate action in writing without a meeting, the Board may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted by the Board, and which date shall not be more than ten (10) days after the date upon which the resolution fixing the record date is adopted by the Board. If no record date has been fixed by the Board, the record date for determining the stockholders entitled to consent to corporate action in writing without a meeting, when no prior action by the Board is required under the DGCL, shall be the first date on which a signed written consent setting forth the action taken or proposed to be taken is delivered to the Corporation by delivery to its registered office in the State of Delaware, its principal place of business or an officer or agent of the Corporation having custody of the book in which proceedings of meetings of stockholders are recorded. Delivery made to the Corporation's registered office shall be by hand or certified or registered mail, return receipt requested. If no record date has been fixed by the Board adopts the resolution taking such prior action.

1.12 Conduct of Meetings.

(A) Meetings of stockholders shall be presided over by the Chairman of the Board, if any, or in the Chairman's absence by the Vice Chairman of the Board, if any, or in the

Vice Chairman's absence by the Chief Executive Officer, or in the Chief Executive Officer's absence, by the President, or in the President's absence by a Vice President, or in the absence of all of the foregoing persons by a chairman designated by the Board. The Secretary shall act as secretary of the meeting, but in the Secretary's absence the chairman of the meeting may appoint any person to act as secretary of the meeting.

(B) The Board may adopt by resolution such rules, regulations and procedures for the conduct of any meeting of stockholders of the Corporation as it shall deem appropriate including, without limitation, such guidelines and procedures as it may deem appropriate in its sole discretion regarding the participation by means of remote communication of stockholders and proxyholders not physically present at a meeting. Except to the extent inconsistent with such rules, regulations and procedures as adopted by the Board, the chairman of any meeting of stockholders shall have the right and authority to convene and (for any or no reason) to recess and/or adjourn the meeting, to prescribe such rules, regulations and procedures and to do all such acts as, in the judgment of such chairman, are appropriate for the proper conduct of the meeting. Such rules, regulations or procedures, whether adopted by the Board or prescribed by the chairman of the meeting, may include, without limitation, the following: (i) the establishment of an agenda or order of business for the meeting is to stockholders of record of the Corporation, their duly authorized and constituted proxies or such other persons as shall be determined; (iv) restrictions on entry to the meeting after the time fixed for the commencement thereof; and (v) limitations on the time allotted to questions or comments by participants. Unless and to the extent determined by the Board or the chairman of the meeting, meetings of stockholders shall not be required to be held in accordance with the rules of parliamentary procedure.

(C) The chairman of the meeting shall announce at the meeting the date and time of the opening and closing of the polls for each matter voted upon at the meeting. After the polls close, no ballots, proxies or votes or any revocations or changes thereto may be accepted unless the Court of Chancery of the State of Delaware shall determine otherwise.

(D) In advance of any meeting of stockholders, the Board, the Chairman of the Board, the Chief Executive Officer or the President shall appoint one or more inspectors of election to act at the meeting and make a written report thereof. One or more other persons may be designated as alternate inspectors to replace any inspector who fails to act. If no inspector or alternate is present, ready and willing to act at a meeting of stockholders, the chairman of the meeting shall appoint one or more inspectors to act at the meeting. Unless otherwise required by law, inspectors may be officers, employees or agents of the Corporation. Each inspector, before entering upon the discharge of such inspector's duties, shall take and sign an oath faithfully to execute the duties of inspector with strict impartiality and according to the best of such inspector's ability. The inspector shall have the duties prescribed by law and, when the vote is completed, shall make a certificate of the result of the vote taken and of such other facts as may be required by law. Every vote taken by ballots shall be counted by a duly appointed inspector or duly appointed inspectors.

ARTICLE II

DIRECTORS

2.1 <u>General Powers</u>. The business and affairs of the Corporation shall be managed by or under the direction of the Board, which may exercise all of the powers of the Corporation except as otherwise provided by law or the Certificate of Incorporation.

2.2 Election, Number and Qualification. At each annual meeting, directors shall be elected to hold office until the next annual meeting and until their successors have been duly elected and qualified; except that if any such election shall be not so held, such election shall take place at a stockholders' meeting called and held in accordance with the DGCL. At any meeting of stockholders at which directors are to be elected, directors shall be elected by the plurality vote of the votes cast by the holders of shares present or represented at the meeting and entitled to vote thereon. Election of directors need not be by written ballot. Subject to the Certificate of Incorporation, the total number of directors constituting the Board shall be such number as may be fixed from time to time by resolution of the Board. Each director shall be at least 18 years of age. A director need not be a stockholder of the Corporation, a citizen of the United States, or a resident of the State of Delaware.

2.3 <u>Chairman of the Board</u>; Vice Chairman of the Board. The Board may appoint from its members a Chairman of the Board and a Vice Chairman of the Board, neither of whom need be an employee or officer of the Corporation. If the Board appoints a Chairman of the Board, such Chairman shall perform such duties and possess such powers as are assigned by the Board and, if the Chairman of the Board is also designated as the Corporation's Chief Executive Officer, shall have the powers and duties of the Chief Executive Officer prescribed in Section 3.7 of these Bylaws. If the Board appoints a Vice Chairman of the Board, such Vice Chairman shall perform such duties and possess such powers as are assigned by the Board, the Chairman of the Board or, in the Chairman's absence, the Vice Chairman of the Board, if any, shall preside at all meetings of the Board.

2.4 Quorum. A majority of the directors at any time in office shall constitute a quorum of the Board. If at any meeting of the Board there shall be less than a quorum, a majority of the directors present may adjourn the meeting from time to time without further notice other than announcement at the meeting, until a quorum shall be present.

2.5 <u>Action at Meeting</u>. Every act or decision done or made by a majority of the directors present at a meeting duly held at which a quorum is present shall be regarded as the act of the Board, unless a greater number is required by law, by the Certificate of Incorporation or these Bylaws.

2.6 <u>Removal</u>. Subject to the rights of holders of any series of Preferred Stock, directors of the Corporation may be removed as provided in the Certificate of Incorporation.

2.7 Vacancies. Subject to the provisions of the Certificate of Incorporation and the rights of holders of any series of Preferred Stock, any newly created directorship that results from an increase in the number of directors or any vacancy on the Board that results from the death, disability, resignation, disqualification or removal of any director or from any other cause shall be filled solely by the affirmative vote of a majority of the directors then in office, even if less than a quorum, or by a sole remaining director and shall not be filled by the stockholders. Any director elected to fill a vacancy shall hold office for the remaining term of his or her predecessor.

2.8 <u>Resignation</u>. Any director may resign by delivering a resignation in writing or by electronic transmission to the Corporation at its principal office or to the Chairman of the Board, the Chief Executive Officer, the President or the Secretary. Such resignation shall be effective upon delivery unless it is specified to be effective at some later time or upon the happening of some later event.

2.9 <u>Regular Meetings</u>. Regular meetings of the Board may be held without notice at such time and place as shall be determined from time to time by the Board; <u>provided</u> that any director who is absent when such a determination is made shall be given notice of the determination. A regular meeting of the Board may be held without notice immediately after and at the same place as the annual meeting of stockholders.

2.10 <u>Special Meetings</u>. Special meetings of the Board may be called by the Chairman of the Board, by the Chief Executive Officer, or by the affirmative vote of a majority of the directors then in office or by one director in the event that there is only a single director in office.

2.11 <u>Notice of Special Meetings</u>. Notice of the date, place and time of any special meeting of the Board shall be given to each director by the Secretary or by the officer or one of the directors calling the meeting. Notice shall be duly given to each director (a) in person or by telephone at least twenty-four (24) hours in advance of the meeting, (b) by sending written notice by reputable overnight courier, telecopy, facsimile or other means of electronic transmission, or delivering written notice by hand, to such director's last known business, home or means of electronic transmission address at least twenty-four (24) hours in advance of the meeting, or (c) by sending written notice by first-class mail to such director's last known business or home address at least seventy-two (72) hours in advance of the meeting. A notice or waiver of notice of a meeting of the Board need not specify the purposes of the meeting.

2.12 <u>Meetings by Conference Communications Equipment</u>. Directors may participate in meetings of the Board or any committee thereof by means of conference telephone or other communications equipment by means of which all persons participating in the meeting can hear each other, and participation by such means shall constitute presence in person at such meeting.

2.13 <u>Action by Written Consent</u>. Any action required or permitted to be taken at any meeting of the Board or of any committee thereof may be taken without a meeting, if all members of the Board or committee, as the case may be, consent to the action in writing or by electronic transmission, and the writing or writings or electronic transmission or transmissions are filed with the minutes of proceedings of the Board or committee thereof. Such filing shall be in paper form if the minutes are maintained in paper form and shall be in electronic form if the minutes are maintained in electronic form.

2.14 <u>Committees</u>. The Board may designate one or more committees, each committee to consist of one or more of the directors of the Corporation with such lawfully delegable powers and duties as the Board thereby confers, to serve at the pleasure of the Board. The Board may designate one or more directors as alternate members of any committee, who may replace any absent or disqualified member at any meeting of the committee. In the absence or disqualification of a member of a committee, the member or members of the committee present at any meeting and not disqualified from voting, whether or not such member or members constitute a quorum, may unanimously appoint another member of the Board to act at the meeting in the place of any such absent or disqualified member. Any such committee, to the extent provided in the resolution of the Board but subject to the DGCL, shall have and may exercise all the powers and authority of the Board in the management of the business and affairs of the Corporation and may authorize the seal of the Corporation to be affixed to all papers which may require it. Each such committee shall keep minutes and make such reports as the Board may otherwise determine, any committee may make rules for the conduct of its business, but unless otherwise provided by the committee or in such rules, its business shall be conducted as nearly as possible in the same manner as is provided in these Bylaws for the Board. Except as otherwise provided in the Certificate of Incorporation, these Bylaws, or the resolution of the Board designating the committee any create one or more subcommittee, and delegate to a subcommittee any or all of the powers and authority of the consist of one or more members of the committee, and delegate to a subcommittee any or all of the powers and authority of the consist of one or more members of the committee, and delegate to a subcommittee any or all of the powers and authority of the committee.

2.15 <u>Compensation of Directors</u>. Directors may be paid such compensation for their services and such reimbursement for expenses of attendance at meetings as the Board may from

time to time determine. No such payment shall preclude any director from serving the Corporation or any of its parent or subsidiary entities in any other capacity and receiving compensation for such service.

ARTICLE III

OFFICERS

3.1 <u>Titles</u>. The officers of the Corporation may consist of a Chief Executive Officer, a President, a Chief Financial Officer, a Treasurer and a Secretary and such other officers with such other titles as the Board shall from time to time determine. The Board may appoint such other officers, including one or more Vice Presidents and one or more Assistant Treasurers or Assistant Secretaries, as it may deem appropriate from time to time.

3.2 Election. The officers of the Corporation shall be elected annually by the Board at its first meeting following the annual meeting of stockholders.

3.3 <u>Qualification</u>. No officer need be a stockholder. To the extent permitted by the DGCL, any two or more offices may be held by the same person.

3.4 <u>Tenure</u>. Except as otherwise provided by law, by the Certificate of Incorporation or by these Bylaws, each officer shall hold office until such officer's successor is duly elected and qualified, unless a different term is specified in the resolution electing or appointing such officer, or until such officer's earlier death, resignation, disqualification or removal.

3.5 <u>Resignation and Removal</u>. Any officer may resign by delivering a written resignation to the Corporation at its principal office or to the Board, the Chief Executive Officer, the President or the Secretary. Such resignation shall be effective upon receipt unless it is specified to be effective at some later time or upon the happening of some later event. Any officer may be removed at any time, with or without cause, by the affirmative vote of a majority of the directors then in office.

3.6 <u>Vacancies</u>. The Board may fill any vacancy occurring in any office for any reason and may, in its discretion, leave unfilled, for such period as it may determine, any offices. Each such successor shall hold office for the unexpired term of such officer's predecessor and until a successor is duly elected and qualified, or until such officer's earlier death, resignation, disqualification or removal.

3.7 <u>Chief Executive Officer</u>. The Board of Directors shall select a Chief Executive Officer to serve at the pleasure of the Board of Directors. The Chief Executive Officer shall have general charge and supervision of the business of the Corporation subject to the direction of the Board, and shall perform all duties and have all powers that are commonly incident to the office of chief executive or that are delegated to such officer by the Board.

3.8 <u>President</u>. The President shall perform all duties and have all powers that are commonly incident to the office of president or that are delegated to such officer by the Board or by the Chief Executive Officer.

3.9 <u>Chief Financial Officer</u>. The Chief Financial Officer shall perform all duties and have all powers that are commonly incident to the office of chief financial officer or that are delegated to such officer by the Board or by the Chief Executive Officer.

3.10 <u>Vice Presidents</u>. Each Vice President shall perform such duties and possess such powers as the Board or the Chief Executive Officer may from time to time prescribe. The Board may assign to any Vice President the title of Executive Vice President, Senior Vice President or any other title selected by the Board.

3.11 Secretary and Assistant Secretaries. The Secretary shall perform such duties and shall have such powers as the Board or the Chief Executive Officer may from time to time prescribe. In addition, the Secretary shall perform such duties and have such powers as are incident to the office of the secretary, including without limitation the duty and power to give notices of all meetings of stockholders and special meetings of the Board, to attend all meetings of stockholders and of the Board and keep a record of the proceedings thereof, to maintain a stock ledger and prepare lists of stockholders and their addresses as required, to be custodian of corporate records and the corporate seal and to affix and attest to the same on documents.

Any Assistant Secretary shall perform such duties and possess such powers as the Board, the Chief Executive Officer or the Secretary may from time to time prescribe. In the event of the absence, inability or refusal to act of the Secretary, the Assistant Secretary (or if there shall be more than one, the Assistant Secretaries in the order determined by the Board) shall perform the duties and exercise the powers of the Secretary.

The chairman of any meeting of the Board or of stockholders may designate a temporary secretary to keep a record of any meeting.

3.12 <u>Treasurer and Assistant Treasurers</u>. The Treasurer shall perform such duties and shall have such powers as may from time to time be assigned by the Board or the Chief Executive Officer. In addition, the Treasurer shall perform such duties and have such powers as are incident to the office of treasurer, including without limitation the duty and power to keep and be responsible for all funds and securities of the Corporation, to deposit funds of the Corporation in depositories selected in accordance with these Bylaws, to disburse such funds as ordered by the Board, to make proper accounts of such funds, and to render as required by the Board statements of all such transactions and of the financial condition of the Corporation.

The Assistant Treasurers shall perform such duties and possess such powers as the Board, the Chief Executive Officer or the Treasurer may from time to time prescribe. In the event of the absence, inability or refusal to act of the Treasurer, the Assistant Treasurer (or if there shall be more than one, the Assistant Treasurers in the order determined by the Board) shall perform the duties and exercise the powers of the Treasurer.

3.13 <u>Delegation of Authority</u>. The Board may from time to time delegate the powers or duties of any officer to any other officer or agent, notwithstanding any provision hereof.

ARTICLE IV

CAPITAL STOCK

4.1 <u>Issuance of Stock</u>. Subject to the provisions of the Certificate of Incorporation, the whole or any part of any unissued balance of the authorized capital stock of the Corporation or the whole or any part of any shares of the authorized capital stock of the Corporation held in the Corporation's treasury may be issued, sold, transferred or otherwise disposed of by vote of the Board in such manner, for such lawful consideration and on such terms as the Board may determine.

4.2 <u>Stock Certificates; Uncertificated Shares</u>. The shares of the Corporation shall be represented by certificates, <u>provided</u> that the Board may provide by resolution or resolutions that some or all of any or all classes or series of the Corporation's stock shall be uncertificated shares. Every holder of stock of the Corporation represented by certificates shall be entitled to have a certificate, in such form as may be prescribed by law and by the Board, representing the number of shares held by such holder registered in certificate form. Each such certificate shall be signed in a manner that complies with Section 158 of the DGCL.

Each certificate for shares of stock which are subject to any restriction on transfer pursuant to the Certificate of Incorporation, these Bylaws, applicable securities laws or any agreement among any number of stockholders or among such holders and the Corporation shall have conspicuously noted on the face or back of the certificate either the full text of the restriction or a statement of the existence of such restriction.

If the Corporation shall be authorized to issue more than one class of stock or more than one series of any class, the powers, designations, preferences and relative participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences and/or rights shall be set forth in full or summarized on the face or back of each certificate representing shares of such class or series of stock, provided that in lieu of the foregoing requirements there may be set forth on the face or back of each certificate representing shares of such class or series of stock a statement that the Corporation will furnish without charge to each stockholder who so requests the powers, designations, preferences and relative, participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences and/or rights.

Within a reasonable time after the issuance or transfer of uncertificated shares, the Corporation shall send to the registered owner thereof a written notice containing the information required to be set forth or stated on certificates pursuant to Sections 151, 156, 202(a) or 218(a) of the DGCL or, with respect to Section 151 of DGCL, a statement that the Corporation will furnish without charge to each stockholder who so requests the powers, designations, preferences and relative participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences and/or rights.

4.3 <u>Transfers</u>. Shares of stock of the Corporation shall be transferable in the manner prescribed by law, the Certificate of Incorporation and in these Bylaws. Transfers of shares of stock of the Corporation shall be made only on the books of the Corporation or by transfer agents designated to transfer shares of stock of the Corporation. Subject to applicable law, shares of stock represented by certificates shall be transferred only on the books of the Corporation by the surrender to the Corporation or its transfer agent of the certificate representing such shares properly endorsed or accompanied by a written assignment or power of attorney properly executed, and with such proof of authority or the authenticity of signature as the Corporation or its transfer agent may reasonably require. Except as may be otherwise required by law, by the Certificate of Incorporation or by these Bylaws, the Corporation shall be entitled to treat the record holder of stock as shown on its books as the owner of such stock for all purposes, including the payment of dividends and the right to vote with respect to such stock, regardless of any transfer, pledge or other disposition of such stock until the shares have been transferred on the books of the Corporation in accordance with the requirements of these Bylaws.

4.4 Lost, Stolen or Destroyed Certificates. The Corporation may issue a new certificate or uncertificated shares in place of any previously issued certificate alleged to have been lost, stolen or destroyed, upon such terms and conditions as the Board may prescribe, including the presentation of reasonable evidence of such loss, theft or destruction and the giving of such indemnity and posting of such bond as the Board may require for the protection of the Corporation or any transfer agent or registrar.

ARTICLE V

GENERAL PROVISIONS

5.1 <u>Fiscal Year</u>. Except as from time to time otherwise designated by the Board, the fiscal year of the Corporation shall be a 52 or 53 week period that ends on the Sunday after the Saturday closest to January 31.

5.2 Corporate Seal. The corporate seal shall be in such form as shall be approved by the Board.

5.3 <u>Waiver of Notice</u>. Whenever notice is required to be given by law, by the Certificate of Incorporation or by these Bylaws, a written waiver signed by the person entitled to notice, or a waiver by electronic transmission by the person entitled to notice, whether before, at or after the time of the event for which notice is to be given, shall be deemed equivalent to notice required to be given to such person. Neither the business nor the purpose of any meeting need be specified in any such waiver. Attendance of a person at a meeting shall constitute a waiver of notice of such meeting, except when the person attends a meeting for the express purpose of objecting, at the beginning of the meeting, to the transaction of any business because the meeting is not lawfully called or convened.

5.4 <u>Voting of Securities</u>. Except as the Board may otherwise designate, the Chief Executive Officer, the President, the Chief Financial Officer or the Treasurer may waive notice, vote, consent, or appoint any person or persons to waive notice, vote or consent, on behalf of the Corporation, and act as, or appoint any person or persons to act as, proxy or attorney-in-fact for this Corporation (with or without power of substitution), with respect to the securities of any other entity which may be held by this Corporation.

5.5 Evidence of Authority. A certificate by the Secretary, or an Assistant Secretary, or a temporary Secretary, as to any action taken by the stockholders, directors, a committee or any officer or representative of the Corporation shall as to all persons who rely on the certificate in good faith be conclusive evidence of such action.

5.6 <u>Certificate of Incorporation</u>. All references in these Bylaws to the Certificate of Incorporation shall be deemed to refer to the Second Amended and Restated Certificate of Incorporation of the Corporation, dated [] as it may be further amended and/or restated and in effect from time to time.

5.7 <u>Severability</u>. If any provision or provisions (or any part thereof) of these Bylaws shall be held to be invalid, illegal or unenforceable as applied to any circumstance for any reason whatsoever: (i) the validity, legality and enforceability of such provisions in any other circumstance and of the remaining provision of these Bylaws (including, without limitation, each portion of any paragraph of these Bylaws containing any such provision held to be invalid, illegal or unenforceable that is not itself held to be invalid, illegal or unenforceable) shall not in

any way be affected or impaired thereby; and (ii) to the fullest extent possible, the provisions of these Bylaws (including, without limitation, each such portion of any paragraph of these Bylaws containing any such provision held to be invalid, illegal or unenforceable) shall be construed so as to permit the Corporation to protect its directors, officers, employees and agents from personal liability in respect of their good faith service or for the benefit of the Corporation to the fullest extent permitted by law.

5.8 <u>Pronouns</u>. All pronouns used in these Bylaws shall be deemed to refer to the masculine, feminine or neuter, singular or plural, as the identity of the person or persons may require.

5.9 <u>Electronic Transmission</u>. For purposes of these Bylaws, "electronic transmission" means any form of communication, not directly involving the physical transmission of paper, that creates a record that may be retained, retrieved, and reviewed by a recipient thereof, and that may be directly reproduced in paper form by such a recipient through an automated process.

ARTICLE VI

AMENDMENTS

These Bylaws may be altered, amended or repealed, in whole or in part, or new Bylaws may be adopted by the Board or by the affirmative vote of the holders of at least sixty-six and two-thirds percent (66 2/3%) in voting power of the Corporation's then outstanding shares entitled to vote generally in the election of directors, voting together as a single class; provided that, for so long as Oak Hill or any person that has acquired (other than through a registered public offering or through a broker's transaction executed on any securities exchange or over-the-counter market) from the Oak Hill Funds ten percent (10%) or more of the issued and outstanding Common Stock is a holder of ten percent (10%) or more of the issued and Common Stock, any alteration, amendment or repeal, in whole or in part, of Sections 1.3 or 7.11 hereof shall require the affirmative approval of the Oak Hill Funds or any such acquirer, as applicable.

ARTICLE VII

INDEMNIFICATION AND ADVANCEMENT

7.1 <u>Right to Indemnification</u>. The Corporation shall indemnify and hold hamless, to the fullest extent permitted by applicable law as it presently exists or may hereafter be amended, any person (a "<u>Covered Person</u>") who was or is made or is threatened to be made a party or is otherwise involved in any action, suit or proceeding, whether civil, criminal, administrative or investigative (a "<u>proceeding</u>"), by reason of the fact that he or she, or a person for whom he or she is the legal representative, is or was a director or officer of the Corporation, or has or had agreed to become a director or officer of the Corporation, or, while a director or officer of the Corporation, is or was serving at the request of the Corporation as a director, employee or agent of another corporation or of a limited liability company, partnership, joint venture, trust, enterprise or nonprofit entity, including service with respect to employee benefit plans, against all liability and loss suffered and expenses (including attomeys' fees) reasonably incurred by such Covered Person. Notwithstanding the preceding (or part thereof) commenced by such Covered Person only if the commencement of such proceeding (or part thereof) by the Covered Person was authorized in the specific case by the Board.

7.2 <u>Prepayment of Expenses</u>. The Corporation shall, to the fullest extent not prohibited by applicable law, as the same exists or may hereafter be amended, pay the expenses (including attorneys' fees) incurred by a Covered Person in defending any proceeding in advance of its final disposition, provided, however, that, to the extent required by law, such payment of expenses in advance of the final disposition of the proceeding shall be made only upon receipt of an undertaking by or on behalf of the Covered Person to repay all amounts advanced if it should be ultimately determined that the Covered Person is not entitled to be indemnified under this Article VII or otherwise.

7.3 <u>Authorization of Indemnification</u>. Any indemnification under this Article VII (unless ordered by a court) shall be made by the Corporation only as authorized in the specific case upon a determination that indemnification of the director or officer is proper in the

circumstances because such person has met the applicable standard of conduct set forth Section 7.4. Such determination shall be made, with respect to a person who is a director or officer at the time of such determination, (i) by a majority vote of the directors who are not parties to such proceeding, even though less than a quorum, or (ii) by a committee of such directors designated by a majority vote of such directors, even though less than a quorum, or (iii) if there are no such directors so direct, by independent legal counsel in a written opinion or (iv) by the stockholders. Such determination shall be made, with respect to former directors and officers, by any person or persons having the authority to act on the matter on behalf of the Corporation. To the extent, however, that a present or former director or officer of the Corporation has been successful on the merits or otherwise in defense of any action, suit or proceeding set forth in Section 7.1 or in defense of any claim, issue or matter therein, such person shall be indemnified against expenses (including attorneys' fees) actually and reasonably incurred by such person in connection therewith, without the necessity of authorization in the specific case.

7.4 <u>Good Faith Defined</u>. For purposes of any determination under Section 7.3, a person shall be deemed to have acted in good faith and in a manner such person reasonably believed to be in or not opposed to the best interests of the Corporation, or, with respect to any criminal action or proceeding, to have had no reasonable cause to believe such person's conduct was unlawful, if such person's action was based on good faith reliance on the records or books of account of the Corporation or another enterprise, or on information supplied to such person by the officers of the Corporation or another enterprise in the course of their duties, or on the advice of legal counsel for the Corporation or another enterprise or on information or another enterprise by an independent certified public accountant or by an appraiser or other expert selected with reasonable care by the Corporation or another enterprise. The term "another enterprise" as used in this Section 7.4 shall mean any other corporation or any partnership, limited liability company, joint venture, trust, employee benefit plan or other enterprise of which such person is or was serving at the request of the Corporation as a director, officer, employee or agent. The provisions of this Section 7.4 shall not be deemed to be exclusive or to limit in any way the circumstances in which a person may be deemed to have met the applicable standard of conduct set forth in the DGCL.

7.5 <u>Right of Claimant to Bring Suit</u>. If a claim for indemnification (following the final disposition of such action, suit or proceeding) or advancement of expenses under this Article VII is not paid in full within thirty (30) days after a written claim therefor by the Covered Person has been received by the Corporation, the Covered Person may file suit to recover the unpaid amount of such claim and, if successful in whole or in part, shall be entitled to be paid the expense (including attorneys' fees) of prosecuting such proceeding. In any such action the Corporation shall have the burden of proving that the Covered Person is not entitled to the requested indemnification or advancement of expenses under applicable law. It shall be a defense to any such action brought to enforce a right to indemnification (but not in an action brought to enforce a right to an advancement of expenses) that the claimant has not met the standards of conduct which make it permissible under the DGCL (or other applicable law) for the Corporation to indemnify the claimant for the amount claimed, but the burden of proving such defense shall be on the Corporation. Neither a contrary determination in the specific case under Section 7.3 nor the absence of any determination thereunder shall be a defense to such application or create a presumption that the claimant has not met any applicable standard of conduct.

7.6 <u>Nonexclusivity of Indemnification and Advancement of Expenses</u>. The rights to indemnification and advancement of expenses provided by or granted pursuant to this Article VII shall not be deemed exclusive of any other rights to which those seeking indemnification or advancement of expenses may be entitled under the Certificate of Incorporation, any agreement, or pursuant to any vote of stockholders or disinterested directors or otherwise, both as to action in such person's official capacity and as to action in another capacity while holding such office, it being the policy of the Corporation that, subject to the last sentence of Section 7.1, indemnification of the persons specified in Section 7.1 shall be made to the fullest extent permitted by law. The provisions of this Article VII shall not be deemed to preclude the indemnification of or advancement of expenses to any person who is not specified in Section 7.1 but whom the Corporation has the power or obligation to indemnify under the provisions of the DGCL, or otherwise.

7.7 Insurance. The Corporation may purchase and maintain insurance on behalf of any person who is or was a director, officer, employee or agent of the Corporation, or is or was serving at the request of the Corporation as a director, officer, employee or agent of another

corporation, partnership, limited liability company, joint venture, trust, employee benefit plan or other enterprise against any liability asserted against such person and incurred by such person in any such capacity, or arising out of such person's status as such, whether or not the Corporation would have the power or the obligation to indemnify such person against such liability under the provisions of this Article VII.

7.8 Certain Definitions. For purposes of this Article VII, references to "the Corporation" shall include, in addition to the resulting corporation, any constituent corporation (including any constituent of a constituent) absorbed in a consolidation or merger which, if its separate existence had continued, would have had power and authority to indemnify its directors, officers, employees or agents so that any person who is or was a director, officer, employee or agent of such constituent corporation, or is or was serving at the request of such constituent corporation as a director, officer, employee or agent of another corporation, partnership, limited liability company, joint venture, trust, employee benefit plan or other enterprise, shall stand in the same position under the provisions of this Article VII with respect to the resulting or surviving corporation as such person would have with respect to such constituent corporation if its separate existence had continued. For purposes of this Article VII, references to "fines" shall include any service as a director, officer, employee or agent of the Corporation which imposes duties on, or involves services by, such director, officer, employee or agent of the corporation with respect to an employee benefit plan, its participants or beneficiaries; and a person who acted in good faith and in a manner such person reasonably believed to be in the interest of the participants and beneficiaries of an employee benefit plan shall be deemed to have acted in a manner "not opposed to the best interests of the Corporation" as referred to in this Article VII.

7.9 <u>Survival of Indemnification and Advancement of Expenses</u>. The indemnification and advancement of expenses provided by, or granted pursuant to, this Article VII shall, unless otherwise provided when authorized or ratified, continue as to a person who has ceased to be a director or officer and shall inure to the benefit of the heirs, executors and administrators of such a person.

7.10 <u>Contract Rights</u>. The obligations of the Corporation under this Article VII to indemnify, and advance expenses to, a Covered Person shall be considered a contract between the Corporation and such person, and no modification or repeal of any provision of this Article VII shall affect, to the detriment of such person, such obligations of the Corporation in connection with a claim based on any act or failure to act occurring before such modification or repeal.

7.11 Primary Obligation. The Corporation hereby acknowledges that any Covered Persons that are employees of the Oak Hill Funds, Oak Hill Capital Management, LLC or any of their respective affiliates (the "<u>Oak Hill Covered Persons</u>") have, or may in the future have, certain rights to indemnification, advancement of expenses and/or insurance provided by the Oak Hill Funds or other persons or entities that, directly or indirectly, (i) are controlled by, (ii) control, or (iii) are under common control with, the Oak Hill Funds (collectively, the "<u>Other Indemnitors</u>"). The Corporation hereby agrees (i) that it is the indemnitor of first resort (i.e., its obligations to the Oak Hill Covered Persons are primary and any obligation of the Other Indemnitors to advance expenses or to provide indemnification or to provide insurance for the same expenses or liabilities incurred by the Oak Hill Covered Persons are secondary), (ii) that it shall be required to advance the full amount of expenses incurred by the Oak Hill Covered Persons and shall be liable for the full amount of all expenses, judgments, penalties, fines and amounts paid in settlement to the extent legally permitted and as required by the Cartificate of Incorporation and these Bylaws (or any other agreement between the Corporation and the Oak Hill Covered Persons), without regard to any rights the Oak Hill Covered Persons may have against the Other Indemnitors (whether pursuant to the Certificate of Incorporation, these Bylaws or any other agreement between the Corporation and the Oak Hill Covered Persons, or law), and (iii) that it irrevocably waives, relinquishes and releases the Other Indemnitors from any and all rights and claims against the Other Indemnitors for contribution, subrogation or any other recovery of any kind in respect thereof. Any insurance coverage provided, obtained or paid for by the Corporation, on the one hand, and any Other Indemnitor, on the other hand, shall be subject to the same primary and secondary liability hierarchy set forth i

Corporation shall use its reasonable best efforts to cause any insurance coverage policy contemplated by Section 7.7 hereof and obtained by the Corporation to contain the same primary and secondary liability hierarchy as set forth in this Section 7.11. The Corporation further agrees that no advancement or payment by the Other Indemnitors on behalf of the Oak Hill Covered Persons with respect to any claim for which the Oak Hill Covered Persons have sought indemnification from the Corporation shall affect the foregoing and the Other Indemnitors shall have a right of contribution and/or be subrogated to the extent of such advancement or payment to all of the rights of recovery of the Oak Hill Covered Persons against the Corporation. The Corporation and the Oak Hill Covered Persons agree that the Other Indemnitors are express third party beneficiaries of the terms of this Section 7.11.

Exhibit 4.1



DAVE & BUSTER'S ENTERTAINMENT, INC.

Г

THE COMPANY WILL FURNISH WITHOUT CHARGE TO EACH SHAREHOLDER WHO SO REQUESTS, A SUMMARY OF THE POWERS, DESIGNATIONS, PREFERENCES AND RELATIVE, PARTICIPATING, OPTIONAL OR OTHER SPECIAL RIGHTS OF EACH CLASS OF STOCK OF THE COMPANY AND THE QUALIFICATIONS, LIMITATIONS OR RESTRICTIONS OF SUCH PREFERENCES AND RIGHTS, AND THE VARIATIONS IN RIGHTS, PREFERENCES AND LIMITATIONS DETERMINED FOR EACH SERIES, WHICH ARE FIXED BY THE CERTIFICATE OF INCORPORATION OF THE COMPANY, AS AMENDED, AND THE RESOLUTIONS OF THE BOARD OF DIRECTORS OF THE COMPANY, AND THE AUTHORITY OF THE BOARD OF DIRECTORS TO DETERMINE VARIATIONS FOR FUTURE SERIES. SUCH REQUEST MAY BE MADE TO THE OFFICE OF THE SECRETARY OF THE COMPANY OR TO THE TRANSFER AGENT. THE BOARD OF DIRECTORS MAY REQUIRE THE OWNER OF A LOST OR DESTROYED STOCK CERTIFICATE, OR HIS LEGAL REPRESENTATIVES, TO GIVE THE COMPANY A BOND TO INDEMNIFY IT AND ITS TRANSFER AGENTS AND REGISTRARS AGAINST ANY CLAIM THAT MAY BE MADE AGAINST THEM ON ACCOUNT OF THE ALLEGED LOSS OR DESTRUCTION OF ANY SUCH CERTIFICATE.

Т

		previations, when used in the inscrip cable laws or regulations:	tion on the face of this certifica	e, shall	be construed as though they were written	out in full			
TEN CC	DM - as te	enants in common	UNIF GIFT M	N ACT	Custodian				
TEN EN	NT - as te	enants by the entireties			(Cust) under Uniform Gifts to Minors Act	(Minor)			
JT TEN	- as jo	oint tenants with right of survivorship	UNIF TRF MI	N ACT	Custodian (until a	(State) ge			
	and	not as tenants in common) (Cust)	- <u> </u>			
					under Uniform Transfers to Minors Act				
					(Minor)	(State)			
Additi	ional abbr	reviations may also be used though	not in the above list.						
				_	PLEASE INSERT SOCIAL SECURIT IDENTIFYING NUMBER OF A				
For value r	received,		hereby sell, assign and transfer	unto					
DI EASE E		R TYPEWRITE NAME AND ADDRI	ESS INCLUDING POSTAL 71P	CODE	OF ASSIGNEE)				
(FLEASE F	KINT ON	TITEWRITE NAME AND ADDRI	ESS, INCLUDING FOSTAL ZIF	CODE,	OF ASSIGNEE)				
						Shares			
of the com	mon stock	k represented by the within Certifica	ate, and do hereby irrevocably c	onstitut	e and appoint				
to transfer t	the said st	tock on the books of the within-nam	ned Company with full nower of	substitu	ition in the premises	Attorne			
	the sale s	lock on the books of the within-han	ied company with full power of	substitu	ation in the premises.				
Dated:			20		Signature(s) Guaranteed: Medallion Guar	rantee Stamp			
Signature:					SIGNATURE(S) SHOULD BE GUARANT				
G . (GIBLE GUARANTOR INSTITUTION (Ban ngs and Loan Associations and Credit Uni	· · · · · · · · · · · · · · · · · · ·			
Signature:	Notice:	Notice: The signature to this assignment must correspond with the name as written upon the face of the certificate, in every particular, without alteration or enlargement, or any change whatever.			MEMBERSHIP IN AN APPROVED SIGNATURE GUARANTE MEDALLION PROGRAM, PURSUANT TO S.E.C. RULE 17A0 15.				
			acquin	d after Jan	that we report the cost basis of certain shares wary 1, 2011. If your shares were covered by I you have sold or transferred the shares and				
	0000	TV I NOTOLOTI ONO	reques	ted a speci sed as requ	fic cost basis calculation method, we have uested. If you did not specify a cost basis	S			
		TY INSTRUCTIONS WATERMARKED PAPER, DO NOT ACCEPT WITH	(FIFO)	method, Pl	 d, we have defaulted to the first in, first out lease visit our website or consult your tax d additional information about cost basis. 	34			

If you do not keep in contact with us or do not have any activity in your account for the time periods specified by state law, your property could become subject to state unclaimed property laws and transferred to the appropriate state.

STOCKHOLDERS' AGREEMENT

DATED AS OF

[**×**], 2014

AMONG

DAVE & BUSTER'S ENTERTAINMENT, INC.

AND

THE STOCKHOLDERS PARTY HERETO

		Page
ARTICLE 1 DEL	FINITIONS	1
Section 1.01.	Definitions	1
ARTICLE 2 CO	RPORATE GOVERNANCE	4
Section 2.01.	Board of Directors; Board Nominees; Committees	4
Section 2.02.	Vacancies	5
Section 2.03.	Reimbursement of Director Expenses	5
Section 2.04.	Reimbursement of Certain Oak Hill Expenses	5
Section 2.05.	Corporate Opportunities	5
Section 2.06.	Approvals	6
ARTICLE 3 CO	NFIDENTIAL INFORMATION	6
Section 3.01.	Confidentiality	6
ARTICLE 4 MIS	SCELLANEOUS	7
Section 4.01.	Binding Effect; Assignability; Benefit	7
Section 4.02.	Notices	7
Section 4.03.	Waiver; Amendment; Termination	8
Section 4.04.	Fees and Expenses	9
Section 4.05.	Governing Law	9
Section 4.06.	Jurisdiction	9
Section 4.07.	Waiver of Jury Trial	9
Section 4.08.	Specific Enforcement; Cumulative Remedies	9
Section 4.09.	Entire Agreement	10
Section 4.10.	Severability	10
Section 4.11.	Counterparts; Effectiveness	10
Section 4.12.	No Recourse	10

i

THIS STOCKHOLDERS' AGREEMENT (this "Agreement")

dated as of $[\times]$ among:

(i) Dave & Buster's Entertainment, Inc. (f/k/a Dave & Buster's Parent, Inc.), a Delaware corporation (the "Company"); and

(ii) Oak Hill Capital Partners III, L.P. and Oak Hill Capital Management Partners III, L.P. (collectively, with their permitted assignees as contemplated by Section 4.01(b), "Oak Hill").

WITNESSETH:

WHEREAS, in connection with underwritten initial public offering of Common Stock of the Company (the "*Initial Public Offering*"), it is the intention of the parties hereto to enter into this Agreement to govern Oak Hill's rights with respect to the Company.

NOW, THEREFORE, for good and valuable consideration the sufficiency and adequacy of which is hereby acknowledged, the parties hereto agree as follows:

ARTICLE 1 DEFINITIONS

Section 1.01. Definitions.

(a) The following terms, as used herein, have the following meanings:

"Affiliate" means, with respect to any Person, any other Person directly or indirectly controlling, controlled by or under common control with such Person; provided that no securityholder of the Company shall be deemed an Affiliate of any other securityholder solely by reason of an investment in the Company. For the purpose of this definition, the term "control" (including, with correlative meanings, the terms "controlling", "controlled by" and "under common control with"), as used with respect to any Person, shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by contract or otherwise.

"Board" means the board of directors of the Company.

"Business Day" means any day except a Saturday, Sunday or other day on which commercial banks in New York City are authorized by law to close.

"Charter" means the Second Amended and Restated Certificate of Incorporation of the Company, as the same may be amended from time to time.

"Common Shares" means shares of Common Stock.

"Common Stock" means the Company's common stock, par value \$0.01 per share, and any stock into which such Common Stock may thereafter be converted, changed, reclassified or exchanged.

"*Company Securities*" means (i) the Common Stock, (ii) any preferred stock, (iii) any other common stock issued by the Company and (iv) any securities convertible into or exchangeable for, or options, warrants or other rights to acquire, Common Stock or any other common stock issued by the Company.

"Designation Number" means (i) the product of (x) Oak Hill's aggregate ownership interest in the Company <u>multiplied by</u> (y) the then current number of directors on the Board and (ii) then rounded to the next highest number of directors if the product of clause (x) and (y) does not equal a whole number. For illustrative purposes only, if Oak Hill's aggregate ownership percentage is 60% and there are 9 directors, the product of (x) and (y) would equal 5.4 and the Designation Number would be 6 directors.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Oak Hill Nominees" means the members of the Board designated by Oak Hill.

"Person" means an individual, corporation, limited liability company, partnership, association, trust or other entity or organization, including a government or political subdivision or an agency or instrumentality thereof.

"Relevant Stock Exchange" means the primary U.S. stock exchange on which the Company's Common Stock is listed.

"Securities Act" means the Securities Act of 1933, as amended.

"*Stockholder*" means each Person (other than the Company) who, at any relevant determination date, shall be a party to or bound by this Agreement (as may be amended from time to time) so long as such Person shall "beneficially own" (as such term is defined in Rule 13d-3 of the Exchange Act) any Company Securities.

"Subsidiary" means, with respect to any Person, any entity of which securities or other ownership interests having ordinary voting power to elect a majority of the board of directors or other persons performing similar functions are at the time directly or indirectly owned by such Person.

"*Transfer*" means, with respect to any Company Securities, (i) when used as a verb, to sell, assign, dispose of, exchange, pledge, encumber, hypothecate or otherwise transfer such Company Securities or any participation or interest therein, whether directly or indirectly, or agree or commit to do any of the foregoing and (ii) when used as a noun, a direct or indirect sale, assignment, disposition, exchange, pledge, encumbrance, hypothecation, or other transfer of such Company Securities or any participation or interest therein or any agreement or commitment to do any of the foregoing.

(b) Each of the following terms is defined in the Section set forth opposite such term:

TERM	SECTION
Agreement	Preamble
Company	Preamble
Confidential Information	3.01(b)
Initial Public Offering	Recitals
Oak Hill	Preamble
Replacement Nominee	2.02

(c) Other Definitional and Interpretive Matters. Unless otherwise expressly provided, for purposes of this Agreement, the following rules of interpretation shall apply:

<u>Calculation of Time Period</u>. When calculating the period of time before which, within which or following which any act is to be done or step taken pursuant to this Agreement, the date that is the reference date in calculating such period shall be excluded. If the last day of such period is a non-Business Day, the period in question shall end on the next succeeding Business Day.

Dollars. Any reference in this Agreement to \$ shall mean U.S. dollars.

Exhibits/Schedules. The Exhibits and Schedules to this Agreement are hereby incorporated and made a part hereof and are an integral part of this Agreement. All Exhibits, Annexes and Schedules annexed hereto or referred to herein are hereby incorporated in and made a part of this Agreement as if set forth in full herein. Any capitalized terms used in any Schedule, Annex or Exhibit but not otherwise defined therein shall be defined as set forth in this Agreement.

Gender and Number. Any reference in this Agreement to gender shall include all genders, and words imparting the singular number only shall include the plural and vice versa.

<u>Headings</u>. The provision of a Table of Contents, the division of this Agreement into Articles, Sections and other subdivisions and the insertion of headings are for convenience of reference only and shall not affect or be utilized in construing or interpreting this Agreement. All references in this Agreement to any "Section" are to the corresponding Section of this Agreement unless otherwise specified.

Herein. The words such as "*herein,*" "*hereinafter,*" "*hereof,*" and "*hereunder*" refer to this Agreement as a whole and not merely to a subdivision in which such words appear unless the context otherwise requires.

Including. Wherever the word "include," "includes," or "including" is used in this Agreement, it shall be deemed to be followed by the words "without limitation".

ARTICLE 2 CORPORATE GOVERNANCE

Section 2.01. Board of Directors; Board Nominees; Committees.

(a) Concurrently with the Initial Public Offering, (i) the Board of the Company shall consist of the following nine (9) directors: Tyler J. Wolfram, Kevin M. Mailender, Alan J. Lacy, David A. Jones, Kevin M. Sheehan, Jonathan S. Halkyard, Michael J. Griffith, J. Taylor Crandall, and Stephen M. King, (ii) the Chairman of the Board shall be Alan J. Lacy and (iii) the lead independent director shall be Alan J. Lacy.

(b) Subject to applicable law and the Relevant Stock Exchange rules, Oak Hill shall be entitled to designate directors to serve on the Board proportionate to Oak Hill's aggregate ownership interest in the Company, which such number of director designees shall be equal to the Designation Number; *provided* that as long as Oak Hill owns 5% or more of the issued and outstanding Common Stock of the Company, Oak Hill shall be entitled to designate to the Board at least one (1) director. As of the date hereof, Tyler J. Wolfram, Kevin M. Mailender and J. Taylor Crandall are the Oak Hill Nominees.

(c) Subject to applicable law and the Relevant Stock Exchange rules, (1) the Board shall have at least the following committees and (2) subject to Section 2.01(c)(i) and Section (c)(ii), each committee of the Board shall have at least one (1) member designated by Oak Hill.

(i) For so long as Oak Hill owns 20% or more of the issued and outstanding Common Stock of the Company, the Nominating and Corporate Governance Committee shall consist of no more than three (3) members. Oak Hill shall have the right to designate the members of the Nominating and Corporate Governance Committee up to the number of Oak Hill Nominees and the remaining members shall be appointed by the Board. The members of the Nominating and Corporate Governance Committee shall initially be Tyler J. Wolfram, Alan J. Lacy and Kevin M. Mailender.

(ii) Audit Committee which shall consist of members appointed by the Board. The members of the Audit Committee shall initially be Kevin M. Sheehan, Jonathan S. Halkyard and Michael J. Griffith. The Chairman of the Audit Committee shall initially be Kevin M. Sheehan.

(iii) Compensation Committee which shall consist of members appointed by the Board. The members of the Compensation Committee shall initially be Tyler J. Wolfram, David A. Jones, Alan J. Lacy, Michael J. Griffith and Jonathan S. Halkyard. The Chairman of the Compensation Committee shall initially be David A. Jones.

(iv) Plan Subcommittee of the Compensation Committee which shall consist of members appointed by the Board. The members of the Plan Committee shall initially be Michael J. Griffith and Jonathan S. Halkyard.

(d) The Company agrees to cause each individual designated pursuant to this <u>Section 2.01</u> or <u>Section 2.02</u> to be nominated to serve as a director on the Board, or as Chairman of the Board, and to take all other necessary actions (including calling a special meeting of the Board and/or stockholders or expanding the size of the Board) to ensure that the composition of the Board and any committee of the Board, as applicable, is as set forth in this <u>Section 2.01</u> and to

otherwise implement the provisions of this <u>Section 2.01</u> and <u>Section 2.02</u>. For the avoidance of doubt, the Company agrees to expand the size of the Board to ensure that that the composition of the Board is as set forth in <u>Section 2.01(b)</u>.

Section 2.02. <u>Vacancies</u>. If, as a result of death, disability, retirement, resignation, removal or otherwise, there shall exist or occur any vacancy of an Oak Hill Nominee on the Board, Oak Hill may designate another individual (the "*Replacement Nominee*") to fill such vacancy and serve as a director on the Board.

Section 2.03. <u>Reimbursement of Director Expenses</u>. The Company will pay all reasonable out-of-pocket expenses incurred by the Oak Hill Nominees in connection with traveling to and from and attending meetings of the Board (and any committee thereof) and while conducting business at the request of the Company.

Section 2.04. <u>Reimbursement of Certain Oak Hill Expenses</u>. The Company shall reimburse Oak Hill and its Affiliates for the costs of any third party advisors retained by Oak Hill and its Affiliates, and for any out of pocket expenses incurred in connection with (i) counsel retained by Oak Hill to advise the Oak Hill Nominees and/or the Company in connection with any matter related to or arising out of meetings of the Board (or committees thereof) or otherwise raised by management, (ii) any review, amendment and or enforcement of this Agreement, (iii) the agreements entered into in connection with the Initial Public Offering and the transactions contemplated hereby and thereby (including, the enforcement or amendment thereof) and (iv) any regulatory filings of the Company involving Oak Hill and/or its Affiliates (including any liquor license filings or securities filings (13D, Form 4 etc.)) and similar matters.

Section 2.05. <u>Corporate Opportunities</u>. In furtherance of, and without limiting what is set forth in the Charter, each of the parties hereto acknowledges that Oak Hill, its Affiliates and any related investment funds and portfolio companies may review the business plans and related proprietary information of any enterprise, including any enterprise which may have products or services which compete directly or indirectly with those of the Company or any of its Affiliates or Subsidiaries and may trade in the securities of such enterprise. Nothing in this Agreement shall preclude or in any way restrict Oak Hill, its Affiliates or any related investment funds or portfolio companies from investing or participating in any particular enterprise, or trading in the securities thereof whether or not such enterprise has products or services that compete with those of the Company or any of its Affiliates or Subsidiaries. Notwithstanding anything to the contrary herein, the Company expressly acknowledges and agrees that: (a) Oak Hill, members of the Board designated by Oak Hill and Affiliates or portfolio companies of Oak Hill, have the right to, and shall have no duty (contractual or otherwise) not to, directly or indirectly, engage in the same or similar business activities or lines of business as the Company of Oak Hill acquires knowledge of a potential transaction or matter that may be a corporate opportunity for any of the Company or any of its Affiliates or Subsidiaries; and (b) in the event that Oak Hill or Affiliates or portfolio companies of Oak Hill shall have no duty (contractual or otherwise) to be designated by Oak Hill or any Affiliate or portfolio company of Oak Hill, members of the Board designated by Oak Hill or any of its Affiliates or subsidiaries; and (b) in the event that Oak Hill, members of the Board designated by Oak Hill or any of the Company or any of its Affiliates or portfolio companies of Oak Hill or portfolio company of Oak Hill, members of the Board designated by Oak Hill or any Affiliates

officer solely in his or her capacity as a director or officer of the Company), and, notwithstanding any provision of this Agreement to the contrary, shall not be liable to the Company or any of its Affiliates or Subsidiaries or any stockholders for breach of any duty (contractual or otherwise) by reason of the fact that Oak Hill, any Affiliate thereof or related investment fund or portfolio company thereof, directly or indirectly, pursues or acquires such opportunity for itself, directs such opportunity to another Person, or does not present such opportunity to the Company or any of its Affiliates.

Section 2.06. <u>Approvals</u>. For so long as Oak Hill owns 25% or more of the issued and outstanding Common Stock of the Company, the Company and its Subsidiaries shall not take any of the following actions without the prior written consent of Oak Hill:

(a) declare or pay any non-pro rata dividends or other non-pro rata distributions or purchases in respect of the Company Securities; or

(b) amend the Charter or bylaws of the Company in a manner adverse to Oak Hill.

ARTICLE 3 CONFIDENTIAL INFORMATION

Section 3.01. Confidentiality.

(a) The Company acknowledges and agrees that (i) Oak Hill may disclose Confidential Information to its Affiliates, representatives and advisors, (ii) any Oak Hill Nominee may disclose Confidential Information to Oak Hill and its Affiliates, representatives and advisors, (iii) Oak Hill and its Affiliates may disclose Confidential Information if requested or required by law, judicial or governmental order, deposition, interrogatory, subpoena, civil investigation, demand, discovery request or similar process and (iv) Oak Hill and its Affiliates may disclose Confidential Information to any potential purchaser of the Company and/or Oak Hill's Company Securities, that executes a customary confidentiality agreement. Oak Hill shall have no obligation to the Company to disclose discussions with any potential purchaser of its Company Securities. For so long as Oak Hill owns 10% or more of the issued and outstanding Common Stock of the Company, Oak Hill will be granted access to customary non-public information of the Company and its Subsidiaries' management team as reasonably requested by Oak Hill.

(b) "*Confidential Information*" shall mean any confidential or proprietary information relating to the business or affairs of the Company or any of its Affiliates, including, but not limited to, information relating to financial statements, customer identities, potential customers, employees, sales representatives, suppliers, servicing methods, equipment programs, strategies and information, analyses, profit margins or other proprietary information used by the

Company or any of its Affiliates; *provided*, *however*, that Confidential Information does not include any information which is in the public domain or becomes known in the industry through no wrongful act on the part of Oak Hill; *provided* that Confidential Information shall not include information that (i) is or becomes generally known to the public other than as a result of a disclosure by Oak Hill in violation of this Agreement, (ii) is or was available to Oak Hill, or (iii) was or becomes available to Oak Hill on a non-confidential basis prior to its disclosure to Oak Hill, or (iii) was or becomes available to Oak Hill on a non-confidential basis from a source other than the Company, which source is or was (at the time of receipt of the relevant information) not bound by a confidentiality agreement with the Company or another person.

ARTICLE 4 MISCELLANEOUS

Section 4.01. Binding Effect; Assignability; Benefit.

(a) This Agreement shall inure to the benefit of and be binding upon the parties hereto and their respective heirs, successors, legal representatives and permitted assigns.

(b) Neither this Agreement nor any right, remedy, obligation or liability arising hereunder or by reason hereof shall be assignable by any party hereto pursuant to any Transfer of Company Securities or otherwise; *provided, however*, that in connection with any Transfer of Company Securities by Oak Hill (i) to any of its Affiliates or (ii) in a privately negotiated transaction, in each case Oak Hill may assign all or any portion of its rights as set forth in <u>Article 2</u> and <u>Article 3</u> to any transfere who agrees to be bound by this Agreement.

(c) Nothing in this Agreement, expressed or implied, is intended to confer on any Person other than the parties hereto, and their respective heirs, successors, legal representatives and permitted assigns, any rights, remedies, obligations or liabilities under or by reason of this Agreement.

Section 4.02. Notices. All notices, requests and other communications to any party shall be in writing and shall be delivered in person, mailed by certified or registered mail, return receipt requested, or sent by facsimile transmission,

If to the Company, to:

Dave & Buster's Entertainment, Inc. 2481 Manana Drive Dallas, Texas 75220 Attention: Jay L. Tobin, Esq. Fax: (214) 357-1536

With a copy to:

Oak Hill Capital Management, LLC 65 East 55th Street, 32nd Floor New York, NY 10022 Attention: John R. Monsky, Esq. Fax: (212) 527-8450

and

Weil, Gotshal & Manges LLP 767 Fifth Avenue New York, New York 10153 Attention: Douglas P. Warner, Esq. Fax: (212) 310-8007

If to Oak Hill, to:

Oak Hill Capital Management, LLC 65 East 55th Street, 32nd Floor New York, NY 10022 Attention: John R. Monsky, Esq. Fax: (212) 527-8450

With a copy to:

Weil, Gotshal & Manges LLP 767 Fifth Avenue New York, New York 10153 Attention: Douglas P. Warner, Esq. Fax: (212) 310-8007

or, in each case, at such other address or fax number as such party may hereafter specify for the purpose of notices hereunder by written notice to the other parties hereto. All notices, requests and other communications shall be deemed received on the date of receipt by the recipient thereof if received prior to 5:00 p.m. in the place of receipt and such day is a Business Day in the place of receipt. Otherwise, any such notice, request or communication shall be deemed not to have been received until the next succeeding Business Day in the place of receipt. Any notice, request or other written communication sent by facsimile transmission shall be confirmed by certified or registered mail, return receipt requested, posted within one (1) Business Day, or by personal delivery, whether courier or otherwise, made within two (2) Business Days after the date of such facsimile transmissions.

Any Person that hereafter becomes a Stockholder shall provide its address and fax number to the Company, which shall promptly provide such information to each other Stockholder.

Section 4.03. Waiver; Amendment; Termination.

(a) No provision of this Agreement may be waived except by an instrument in writing executed by the party against whom the waiver is to be effective. No provision of this Agreement may be amended or otherwise modified except by an instrument in writing executed by the Company and Oak Hill.

(b) This Agreement shall terminate upon, (i) the written request of Oak Hill or (ii) such time as Oak Hill or any transferee of Oak Hill who agrees to be bound by this Agreement owns less than 5% in the aggregate of the issued and outstanding Common Stock of the Company.

Section 4.04. Fees and Expenses. Except as set forth in Section 2.04, each party shall pay its own costs and expenses incurred in connection with the preparation and execution of this Agreement.

Section 4.05. Governing Law. This Agreement, and all claims or causes of action (whether at law, in equity, in contract, in tort or otherwise) based upon, arising out of, related to or otherwise in connection with this Agreement or the transactions contemplated hereby, shall be exclusively governed by, and construed in accordance with, the internal laws of the State of Delaware, without giving effect to principles or rules of conflict of laws to the extent such principles or rules would require or permit the application of laws of another jurisdiction (including any claim or cause of action based upon, arising out of or related to any representation or warranty made in or in connection with this Agreement or as an inducement to enter into this Agreement).

Section 4.06. Jurisdiction. The parties hereby agree that any suit, action or proceeding (whether at law, in equity, in contract, in tort or otherwise) seeking to enforce any provision of, or based on any matter arising out of or in connection with, this Agreement or the transactions contemplated hereby (including any claim or cause of action based upon, arising out of or related to any representation or warranty made in or in connection with this Agreement or as an inducement to enter into this Agreement) shall be exclusively brought in the United States District Court for the Southern District of New York or any New York State court sitting in New York City, so long as one of such courts shall have subject matter jurisdiction over such suit, action or proceeding, and that any case of action arising out of this Agreement shall be deemed to have arisen from a transaction of business in the State of New York, and each of the parties hereby irrevocably consents to the exclusive jurisdiction of such courts (and of the appropriate appellate courts therefrom) in any such suit, action or proceeding and irrevocably waives, to the fullest extent permitted by law, any objection that it may now or hereafter have to the laying of the venue of any such suit, action or proceeding in any such court or that any such suit, action or proceeding which is brought in an ysuch court has been brought in an jurisdiction of any such court. Without limiting the foregoing, each party agrees that service of process on such party as provided in <u>Section 4.03</u> shall be deemed effective service of process on such party.

Section 4.07. <u>Waiver of Jury Trial</u>. EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING (WHETHER AT LAW, IN EQUITY, IN CONTRACT, IN TORT OR OTHER WISE) ARISING OUT OF OR RELATED TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.

Section 4.08. Specific Enforcement; Cumulative Remedies. The parties hereto acknowledge that money damages may not be an adequate remedy for violations of this

Agreement and that any party, in addition to any other rights and remedies which the parties may have hereunder or at law or in equity, may, in his or its sole discretion, apply to a court of competent jurisdiction for specific performance or injunction or such other relief as such court may deem just and proper in order to enforce this Agreement or prevent any violation hereof and, to the extent permitted by applicable law, each party waives any objection to the imposition of such relief. All rights, powers and remedies provided under this Agreement or otherwise available in respect hereof at law or in equity shall be cumulative and not alternative, and the exercise or beginning of the exercise of any thereof by any party shall not preclude the simultaneous or later exercise of any other such rights, powers or remedies by such party.

Section 4.09. Entire Agreement. This Agreement and any exhibits and other documents referred to herein constitute the entire agreement and understanding among the parties hereto in respect of the subject matter hereof and thereof and supersede all prior and contemporaneous agreements and understandings, both oral and written, among the parties hereto, or between any of them, with respect to the subject matter hereof and thereof.

Section 4.10. Severability. If any term, provision, covenant or restriction of this Agreement is held by a court of competent jurisdiction or other authority to be invalid, void or unenforceable, the remainder of the terms, provisions, covenants and restrictions of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated. Upon such a determination, the parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner so that the transactions contemplated hereby be consummated as originally contemplated to the fullest extent possible.

Section 4.11. Counterparts; Effectiveness. This Agreement may be executed in any number of counterparts, each of which shall be deemed to be an original, with the same effect as if the signatures thereto and hereto were upon the same instrument.

Section 4.12. No Recourse. Notwithstanding anything that may be expressed or implied in this Agreement, and notwithstanding the fact that the entities comprising Oak Hill are two limited partnerships, each party to this Agreement covenants, agrees and acknowledges that no recourse under this Agreement or any documents or instruments delivered in connection with this Agreement shall be had against any of Oak Hill's current or future directors, officers, employees, general or limited partners, members, managers or trustees, or any partner, member, manager or trustee, as such, whether by the enforcement of any assessment or by any legal or equitable proceeding, or by virtue of any statute, regulation or other applicable law, it being expressly agreed and acknowledged that no personal liability whatsoever shall attach to, be imposed on or otherwise be incurred by any of Oak Hill's current or future officers, agents, employees, directors, managers, members, or any Affiliates or assignees thereof, as such, for any obligation of Oak Hill under this Agreement or any documents or instruments delivered in connection with this Agreement for any claim based on, in respect of or by reason of such obligations or their creation.

[The remainder of this page is intentionally left blank.]

¹⁰

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed by their respective authorized officers as of the day and year first above written.

DAVE & BUSTER'S ENTERTAINMENT, INC.

By:

Name: Title:

OAK HILL CAPITAL PARTNERS III, L.P.

- By: OHCP GenPar III, L.P., its General Partner
- By: OHCP MGP Partners III, L.P., its General Partner
- By: OHCP MGP III, Ltd., its General Partner

By:

Name: John R. Monsky Title:

OAK HILL CAPITAL MANAGEMENT PARTNERS III, L.P.

- By: OHCP GenPar III, L.P., its General Partner
- By: OHCP MGP Partners III, L.P., its General Partner
- By: OHCP MGP III, Ltd., its General Partner

By:

Name: John R. Monsky Title:

[SIGNATURE PAGE TO STOCKHOLDERS AGREEMENT]

REGISTRATION RIGHTS AGREEMENT

DATED AS OF

[**×**], 2014

AMONG

DAVE & BUSTER'S ENTERTAINMENT, INC.

AND

THE STOCKHOLDERS PARTY HERETO

n

		Page
ARTICLE 1 R	EGISTRATION RIGHTS	1
Section 1.01.	Demand Registration	1
Section 1.02.	Piggyback Registration	2
Section 1.03.	Automatic Shelf Registrations; Shelf Take-Downs	4
Section 1.04.	Lock-Up Agreements; Transfer Restrictions	5
Section 1.05.	Registration Procedures	5
Section 1.06.	Indemnification by the Company	8
Section 1.07.	Indemnification by the Participating Stockholders	9
Section 1.08.	Conduct of Indemnification Proceedings	9
Section 1.09.	Contribution	10
Section 1.10.	Participation in Public Offering	11
Section 1.11.	Other Indemnification	11
Section 1.12.	Cooperation by the Company	11
Section 1.13.	S-8 Registration Following Initial Public Offering	11
ARTICLE 2 L	EGENDS	12
ARTICLE 3 D	DEFINITIONS	12
ARTICLE 4 M	MISCELLANEOUS	14
Section 4.01.	Termination	14
Section 4.02.	Termination of Stockholders' Agreement	14
Section 4.03.	Amendment and Waiver	15
Section 4.04.	Successors and Assigns	15
Section 4.05.	Severability	15
Section 4.06.	Entire Agreement	15
Section 4.07.	Counterparts; Execution by Facsimile Signature	15
Section 4.08.	Notices	16
Section 4.09.	Governing Law	16
Section 4.10.	Consent to Jurisdiction	16
Section 4.11.	Waiver of Jury Trial	17

i

REGISTRATION RIGHTS AGREEMENT

THIS REGISTRATION RIGHTS AGREEMENT (this "*Agreement*") is made as of [\times], 2014, among Dave & Buster's Entertainment, Inc., a corporation organized under the laws of the State of Delaware (the "*Company*"), and the Persons named on the signature pages hereto (including any additional signatories to this Agreement after the date hereof, the "*Stockholders*").

WITNESSETH:

WHEREAS, the parties hereto desire to enter into this Agreement to establish certain arrangements with respect to the Company Securities owned by the Stockholders, and other related matters; and

NOW, THEREFORE, in consideration of the premises and of the mutual covenants and obligations hereinafter set forth, the parties hereto hereby agree as follows:

ARTICLE 1 REGISTRATION RIGHTS

Section 1.01. Demand Registration.

(a) If, at any time commencing six (6) months after the date of consummation of the Initial Public Offering, the Company shall receive a written request from Oak Hill (the "*Requesting Stockholder*") that the Company effect the registration under the Securities Act of all or any portion of such Requesting Stockholder's Registrable Securities (which shall be effected by a shelf registration if so requested by the Requesting Stockholder), and specifying the intended method of disposition thereof, then the Company shall promptly give notice of such requested registration (each such request shall be referred to herein as a "*Demand Registration*") at least fifteen (15) Business Days prior to the anticipated filing date of the registration statement relating to such Demand Registration to the Other Stockholders holding Registrable Securities and thereupon shall use its best efforts to effect, as expeditiously as possible, the registration under the Securities Act of:

(i) all Registrable Securities for which the Requesting Stockholder has requested registration under this Section 1.01.

(ii) subject to the restrictions set forth in Sections 1.01(f) and 1.02, all other Registrable Securities of the same class as those requested to be registered by the Requesting Stockholders that any Stockholders with rights to request registration under Section 1.02 (all such Stockholders, together with the Requesting Stockholders, the "*Registering Stockholders*") have requested the Company to register by request received by the Company within ten (10) Business Days after such Stockholders receive the Company's notice of the Demand Registration, all to the extent necessary to permit the disposition (in accordance with the intended methods thereof as aforesaid) of the Registrable Securities so to be registered.

(b) Promptly after the expiration of the ten (10) Business Day period referred to in <u>Section 1.01(a)(ii)</u> hereof, the Company will notify all Registering Stockholders of the identities

of the other Registering Stockholders and the number of shares of Registrable Securities requested to be included therein. At any time prior to the effective date of the registration statement relating to such registration, the Requesting Stockholders may revoke such request, without liability to any of the other Registering Stockholders, by providing a notice to the Company revoking such request.

(c) Oak Hill shall have an unlimited number of Demand Registrations.

(d) The Company shall be liable for and pay all Registration Expenses in connection with each Demand Registration, regardless of whether such Registration is effected.

(e) A Demand Registration shall not be deemed to have occurred:

(i) unless the registration statement relating thereto (A) has become effective under the Securities Act and (B) has remained effective for a period of at least 120 days (or such shorter period in which all Registrable Securities of the Registering Stockholders included in such registration have actually been sold thereunder), *provided* that such registration statement shall not be considered a Demand Registration if, after such registration statement becomes effective, (1) such registration statement is interfered with by any stop order, injunction or other order or requirement of the SEC or other governmental agency or court and (2) less than 75% of the Registrable Securities included in such registration statement have been sold thereunder; or

(ii) if the Maximum Offering Size (as defined below) is reduced in accordance with <u>Section 1.01(f)</u> such that less than 50% of the Registrable Securities of the Requesting Stockholders sought to be included in such registration are included.

(f) If a Demand Registration involves a Public Offering and the managing underwriter advises the Company and the Requesting Stockholders that, in its view, the number of Company Securities that the Registering Stockholders and the Company propose to include in such registration exceeds the largest number of shares that can be sold without having an adverse effect on such offering, including the price at which such Company Securities can be sold (the "*Maximum Offering Size*"), the Company shall include in such registration, in the priority listed below, up to the Maximum Offering Size:

(i) first, all Registrable Securities requested to be registered by the Registering Stockholders (allocated, if necessary for the offering not to exceed the Maximum Offering Size, pro rata among such Registering Stockholders on the basis of the relative number of Registrable Securities so requested to be included in such registration by each); and

(ii) second, all Registrable Securities proposed to be registered by the Company.

Section 1.02. Piggyback Registration.

(a) If the Company proposes to register any Company Securities under the Securities Act (whether for itself or in connection with a sale of securities by any Stockholder, but other than a registration on Form S-8 or S-4, or any successor or similar forms, relating to Common Shares issuable upon exercise of employee stock options or in connection with any employee

benefit or similar plan of the Company or in connection with a direct or indirect acquisition by the Company of another Person), the Company shall each such time give prompt written notice at least ten (10) Business Days prior to the anticipated filing date of the registration statement relating to such registration to each Stockholder holding Registrable Securities with rights to require registration of Company Securities hereunder, which notice shall set forth such Stockholder's rights under this Section 1.02 and shall offer such Stockholder the opportunity to include in such registration statement Company Securities of the same class or series of Registrable Securities as proposed to be offered in such registration (a "Piggyback Registration"), subject to the restrictions set forth herein. Upon the written request of any such Stockholder made within five (5) Business Days after the receipt of notice from the Company (which request shall specify the number of Registrable Securities intended to be registered by such Stockholder), the Company shall use its best efforts to effect the registration under the Securities Act of all Registrable Securities that the Company has been so requested to register by all such Stockholders with rights to require registration of Company Securities hereunder, to the extent requisite to permit the disposition of the Registrable Securities so to be registered, provided that (i) if such registration involves a Public Offering, all such Stockholders requesting to be included in the Company's registration must sell their Registrable Securities to the underwriters selected as provided in Section 1.05(f)(i) on the same terms and conditions as apply to the Company or any other selling Stockholders, and (ii) if, at any time after giving notice of its intention to register any Company Securities pursuant to this Section 1.02(a) and prior to the effective date of the registration statement filed in connection with such registration, the Company shall determine for any reason not to register such Company securities, the Company shall give notice to all such Stockholders and, thereupon, shall be relieved of its obligation to register any Registrable Securities in connection with such registration. No registration effected under this Section 1.02 shall relieve the Company of its obligations to effect a Demand Registration to the extent required by Section 1.01. The Company shall be liable for and pay all Registration Expenses in connection with each Piggyback Registration.

(b) If a Piggyback Registration involves a Public Offering (other than any Demand Registration, in which case the provisions with respect to priority of inclusion in such offering set forth in <u>Section 1.01(f)</u> shall apply) and the managing underwriter advises the Company that, in its view, the number of Company Securities that the Company and such selling stockholders propose to include in such registration exceeds the Maximum Offering Size, the Company shall include in such registration, in the following priority, up to the Maximum Offering Size:

(i) with respect to a Public Offering by the Company for its own account:

(A) first, such number of Registrable Securities proposed to be registered for the account of the Company or any Requesting Stockholder on whose account the registration is being made, if any, as would not cause the offering to exceed the Maximum Offering Size, and

(B) second, all Registrable Securities requested to be included in such registration by any Registering Stockholders (allocated, if necessary for the offering not to exceed the Maximum Offering Size, pro rata among such Stockholders based on their relative ownership of Registrable Securities) requested be included in the Piggyback Registration.

(ii) With respect to a Public Offering by the Company for the account of selling stockholders:

(A) first, all Registrable Securities requested to be included in such registration by any Registering Stockholders (allocated, if necessary for the offering not to exceed the Maximum Offering Size, pro rata among such Stockholders based on their relative number of Registrable Securities) requested to be included in the Piggyback Registration; and

(B) second, all Registrable Securities proposed to be registered for the account of the Company.

Section 1.03. Automatic Shelf Registrations; Shelf Take-Downs.

(a) <u>Automatic Shelf Registrations</u>. After the Initial Public Offering, as promptly as practicable following the date upon which the Company becomes a well-known seasoned issuer (as defined in Rule 405 under the Securities Act) (a "*WKST*"), the Company shall file with the SEC a continuous or delayed basis offering pursuant to Rule 415 under the Securities Act (a "*Shelf Registration Statement*"), which, for the avoidance of doubt, would be an automatic shelf registration statement (as defined in Rule 405 under the Securities Act) for a WKSI relating to the offer and sale of all Registrable Securities by the holders of Registrable Securities from time to time in accordance with the methods of distribution elected by such holders of Registrable Securities and set forth in the Shelf Registration Statement and, as promptly as practicable thereafter, shall use its reasonable best efforts to cause such Shelf Registration Statement to become effective under the Securities Act. If, on the date of any such request, the Company does not qualify to file a Shelf Registration Statement under the Securities Act, the provisions of this <u>Section 1.03</u> shall not apply, and the provisions of <u>Section 1.01</u> shall apply instead. In the event the Company is not, or ceases to be, a WKSI, the Company agrees to use reasonable best efforts to facilitate the sale of Registrable Securities on a Form S-3. The Company shall be liable for and pay all Registration Expenses in connection with this <u>Section 1.03</u>.

(b) <u>Shelf Take Downs</u>. Any Requesting Stockholder included in a Shelf Registration Statement (an "*Initiating Shelf Requesting Stockholder*") may initiate an offering or sale of all or part of such Person's Registrable Securities (a "*Shelf Take-Down*"), in which case the provisions of this <u>Section 1.03(b)</u> shall apply.

(i) If an Initiating Shelf Requesting Stockholder so elects in a written request delivered to the Company (an "Underwritten Shelf Take-Down Notice"), a Shelf Take-Down may be in the form of an underwritten offering (an "Underwritten Shelf Take-Down") and the Company shall file and effect an amendment or supplement to its Shelf Registration Statement for such purpose as soon as reasonably practicable; provided, that any such Marketed Underwritten Shelf Take-Down (as defined below) shall be deemed to be, for purposes of <u>Section 1.01</u>, a Demand Registration. Such Initiating Shelf Requesting Stockholder shall indicate in such Underwritten Shelf Take-Down Notice whether it intends for such Underwritten Shelf Take-Down to involve a customary "road show" (including an "electronic road show") or other substantial marketing effort by the underwritters (a "Marketed Underwritten Shelf Take-Down"). Upon receipt of an Underwritten Shelf Take-Down Notice indicating that such

Underwritten Shelf Take-Down will be a Marketed Underwritten Shelf Take-Down, the Company shall promptly give notice of such Marketed Underwritten Shelf Take-Down to all other holders of Registrable Securities included in a Shelf Registration Statement (all such holders, "*Shelf Holders*") and shall permit the participation of all such Shelf Holders that request inclusion in such Marketed Underwritten Shelf Take-Down who respond in writing within five days after the receipt of such notice of their election to participate. The provisions of <u>Section 1.02</u> shall apply with respect to the right of the Initiating Shelf Requesting Stockholder and the other Shelf Holders to participate in any Underwritten Shelf Take-Down.

(ii) If the Initiating Shelf Requesting Stockholder desires to effect a Shelf Take-Down that does not constitute a Marketed Underwritten Shelf Take-Down (a "*Non-Marketed Underwritten Shelf Take-Down*"), the Initiating Shelf Requesting Stockholder shall so indicate in a written request delivered to the Company no later than two Business Days prior to the expected date of such Non-Marketed Underwritten Shelf Take-Down, which request shall include (i) the total number of Registrable Securities expected to be offered and sold in such Non-Marketed Underwritten Shelf Take-Down, (ii) the excepted plan of distribution of such Non-Marketed Underwritten Shelf Take-Down (attention of such Non-Marketed Underwritten Shelf Take-Down (attention of a such Non-Marketed Underwritten Shelf Take-Down (attention or actions required (including the timing thereof) in connection with such Non-Marketed Underwritten Shelf Take-Down (attention of a such Non-Marketed Underwritten Shelf Take-Down), and the Company shall file and effect an amendment or supplement to its Shelf Registration Statement for such purpose as soon as practicable. All determinations as to whether to complete any Non-Marketed Underwritten Shelf Take-Down and as to the timing, manner, price and other terms of any Non-Marketed Underwritten Shelf Take-Down shall be at the discretion of the Initiating Shelf Requesting Stockholder.

Section 1.04. Lock-Up Agreements; Transfer Restrictions. If any registration of Company Securities shall be effected in connection with a Public Offering, each of the Company and each Stockholder shall enter into a customary "lock-up" agreement with the managing underwriter or underwriters (and the Company shall use reasonable best efforts to cause other stockholders, directors and officers to sign commensurate "lock-up" agreements) and neither the Company nor any Stockholder shall effect any public sale or distribution, including any sale pursuant to Rule 144, of any Company Securities or other security of the Company (except as part of such Public Offering) during the period (each such period, a "*Lock-Up Period*") beginning fourteen (14) days prior to the distribution of a preliminary prospectus until, (i) with respect to the Initial Public Offering, the earlier of (x) such time as the Company and the lead managing underwriter shall agree and (y) 180 days after such effective date of the Initial Public Offering or (ii) with respect to any other Public Offering, 90 days after such effective date of the Public Offering (unless the underwriter requires a longer time period, in which case such period shall not exceed 180 days), in each case unless a later date is reasonably required by applicable FINRA rules and regulations.

Section 1.05. <u>Registration Procedures</u>. Whenever any Stockholders request that any Registrable Securities be registered pursuant to <u>Section 1.01</u> or <u>1.02</u> hereof, subject to the provisions of such Sections, the Company shall use its best efforts to effect the registration and the sale of such Registrable Securities in accordance with the intended method of disposition thereof as quickly as practicable, and, in connection with any such request:

(a) The Company shall as expeditiously as possible prepare and file with the SEC a registration statement on any form for which the Company then qualifies or that counsel for the Company shall deem appropriate and which form shall be available for the sale of the Registrable Securities to be registered thereunder in accordance with the intended method of distribution thereof, and use its best efforts to cause such filed registration statement to become and remain effective for a period of not less than 180 days, or in the case of a shelf registration statement, one (1) year (or such shorter period in which all of the Registrable Securities of the Registering Stockholders included in such registration statement shall have actually been sold thereunder).

(b) Prior to filing a registration statement or prospectus or any amendment or supplement thereto, the Company shall, if requested, furnish to each participating Stockholder and each underwriter, if any, of the Registrable Securities covered by such registration statement copies of such registration statement as proposed to be filed, and thereafter the Company shall furnish to such Stockholder and underwriter, if any, such number of copies of such registration statement, each amendment and supplement thereto (in each case including all exhibits thereto and documents incorporated by reference therein), the prospectus included in such registration statement (including each preliminary prospectus and any summary prospectus) and any other prospectus filed under Rule 424 or Rule 430A (or any similar provision then in force) under the Securities Act and such other documents as such Stockholder or underwriter may reasonably request in order to facilitate the disposition of the Registrable Securities owned by such Stockholder.

(c) After the filing of the registration statement, the Company shall (i) cause the related prospectus to be supplemented by any required prospectus supplement, and, as so supplemented, to be filed pursuant to Rule 424 or any similar provision then in force under the Securities Act, (ii) comply with the provisions of the Securities Act and Exchange Act with respect to the disposition of all Registrable Securities covered by such registration statement during the applicable period in accordance with the intended methods of disposition by the Registering Stockholders thereof set forth in such registration statement or supplement to such prospectus and (iii) promptly notify the Registering Stockholders holding Registrable Securities covered by such registration statement of any stop order issued or threatened by the SEC or any state securities commission and take all reasonable actions required to prevent the entry of such stop order or to remove it if entered.

(d) The Company shall use its best efforts to (i) register or qualify the Registrable Securities covered by such registration statement under such other securities or "*blue sky*" laws of such jurisdictions in the United States as the Registering Stockholders holding such Registrable Securities reasonably (in light of such Stockholder's intended plan of distribution) requests and (ii) cause such Registrable Securities to be registered with or approved by such other governmental agencies or authorities as may be necessary by virtue of the business and operations of the Company and do any and all other acts and things that may be reasonably necessary or advisable to enable such Stockholder to consummate the disposition of the Registrable Securities owned by such Stockholder; *provided* that the Company shall not be required to (A) qualify generally to do business in any jurisdiction where it would not otherwise be required to qualify but for this <u>Section 1.05(d)</u>, (B) subject itself to taxation in any such jurisdiction or (C) consent to general service of process in any such jurisdiction.

(e) The Company shall immediately notify each Registering Stockholders holding such Registrable Securities covered by such registration statement, at any time when a prospectus relating thereto is required to be delivered under the Securities Act, of the occurrence of an event requiring the preparation of a supplement or amendment to such prospectus so that, as thereafter delivered to the purchasers of such Registrable Securities, such prospectus will not contain an untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading and promptly prepare and make available to each such Stockholder and file with the SEC any such supplement or amendment.

(f) (i) Oak Hill shall have the right, in its sole discretion, to select the underwriter or underwriters in connection with any Public Offering resulting from a Demand Registration, which underwriter or underwriters may include any Affiliate of Oak Hill, and (ii) the Company shall select an underwriter or underwriters in connection with any other Public Offering. In connection with any Public Offering, the Company shall enter into customary agreements (including an underwriting agreement in customary form) and take all such other actions as are reasonably required in order to expedite or facilitate the disposition of such Registrable Securities in any such Public Offering, including the engagement of a "qualified independent underwriter" in connection with the qualification of the underwriting arrangements with FINRA.

(g) The Company shall make available for inspection by the Registering Stockholders and any underwriter participating in any disposition pursuant to a registration statement being filed by the Company pursuant to this <u>Section 1.05</u> and any attorney, accountant or other professional retained by any such Stockholder or underwriter (collectively, the "*Inspectors*"), all financial and other records, pertinent corporate documents and properties of the Company (collectively, the "*Records*") as shall be reasonably necessary or desirable to enable them to exercise their due diligence responsibility, and cause the Company's officers, directors and employees to supply all information reasonably requested by any Inspectors in connection with such registration statement. Records that the Company determines, in good faith, to be confidential and that it notifies the Inspectors are confidential shall not be disclosed by the Inspectors unless (i) the disclosure of such Records is necessary to avoid or correct a misstatement or omission in such registration statement or (ii) the release of such Records is ordered pursuant to a subpoena or other order from a court of competent jurisdiction. Each Registering Stockholder agrees that information obtained by it as a result of such inspections shall be deemed confidential and shall not be used by it or its Affiliates as the basis for any market transactions in the Company Securities unless and until such information is made generally available to the public. Each Registering Stockholder further agrees that, upon learning that disclosure of such Records is sought in a court of competent jurisdiction, it shall give notice to the Company and allow the Company, at its expense, to undertake appropriate action to prevent disclosure of the Records deemed confidential.

(h) The Company shall furnish to each Registering Stockholder and to each such underwriter, if any, a signed counterpart, addressed to such Stockholder or underwriter, of (i) an opinion or opinions of counsel to the Company and (ii) a comfort letter or comfort letters from the Company's independent public accountants, each in customary form and covering such matters of the kind customarily covered by opinions or comfort letters, as the case may be, as a majority of such Stockholders or the managing underwriter reasonably requests.

(i) The Company shall otherwise use its best efforts to comply with all applicable rules and regulations of the SEC, and make available to its security holders, as soon as reasonably practicable, an earning statement or such other document that shall satisfy the provisions of Section 11 (a) of the Securities Act and Rule 158 thereunder.

(j) The Company may require each Registering Stockholder promptly to furnish in writing to the Company such information regarding the distribution of the Registrable Securities as the Company may from time to time request and such other information as may be legally required in connection with such registration.

(k) Each Registering Stockholder agrees that, upon receipt of any written notice from the Company of the occurrence of any event requiring the preparation of a supplement or amendment of a prospectus relating to the Registrable Securities covered by a registration statement that is required to be delivered under the Securities Act so that, as thereafter delivered to the purchasers of such Registrable Securities, such prospectus will not contain an untrue statement of a material fact or omit to state any material fact required to be stated therein or to make the statements therein not misleading, such Stockholder shall forthwith discontinue disposition of Registrable Securities pursuant to the registration statement covering such Registrable Securities until such Stockholder's receipt of the copies of a supplemented or amended prospectus, and, if so directed by the Company, such Stockholder shall deliver to the Company all copies, other than any permanent file copies then in such Stockholder's possession, of the most recent prospectus covering such Registrable Securities at the time of receipt of such notice. If the Company shall give such notice, the Company shall extend the period during which such registration statement shall be maintained effective (including the period referred to in <u>Section 1.05(a)</u>) by the number of days during the period from and including the date of the giving of notice pursuant to <u>Section 1.05(e)</u>.

(1) The Company shall use its reasonable efforts to list all Registrable Securities covered by such registration statement on any securities exchange or quotation system on which any of the Registrable Securities are then listed or traded and to maintain such listing so long as any such Registrable Securities remain outstanding.

(m) The Company shall have appropriate officers of the Company (i) prepare and make presentations at any "road shows" and before analysts and rating agencies, as the case may be, (ii) take other actions to obtain ratings for any Registrable Securities and (iii) otherwise use their reasonable efforts to cooperate as requested by the underwriters in the offering, marketing or selling of the Registrable Securities.

Section 1.06. Indemnification by the Company. The Company agrees to indemnify and hold each Registering Stockholder holding Registrable Securities covered by a registration statement, its officers, directors, employees, managers, members, partners and agents, and each Person, if any, who controls any such Persons within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act from and against any and all losses, claims, damages, liabilities and expenses (including reasonable expenses of investigation and reasonable attorneys' fees and expenses) ("Damages") caused by or relating to any untrue statement or alleged untrue

statement of a material fact contained in any registration statement, prospectus or free writing prospectus relating to the Registrable Securities (as amended or supplemented if the Company shall have furnished any amendments or supplements thereto) or any preliminary prospectus, or made during any "road shows", or caused by or relating to any omission or alleged omission to state therein a material fact required to be stated therein or necessary to make the statements therein not misleading, except insofar as such Damages are caused by or related to any such untrue statement or omission or alleged untrue statement or omission so made based upon information furnished in writing to the Company by such Stockholder or on such Stockholder's behalf expressly for use therein. The Company also agrees to indemnify any underwriters of the Registrable Securities, their officers and directors and each Person who controls such underwriters within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act on substantially the same basis as that of the indemnification of the Stockholders provided in this <u>Section 1.06</u>.

Section 1.07. Indemnification by the Participating Stockholders. Each Registering Stockholder holding Registrable Securities included in any registration statement agrees to indemnify and hold harmless the Company, its officers, directors and agents and each Person, if any, who controls the Company within the meaning of either Section 15 of the Securities Act or Section 20 of the Exchange Act to the same extent as the foregoing indemnity from the Company to such Stockholder, but only with respect to information furnished in writing to the Company by such Stockholder or on such Stockholder's behalf expressly for use in any registration statement or prospectus relating to the Registrable Securities, or any amendment or supplement thereto, or any preliminary prospectus. Each such Stockholder also agrees to indemnify and hold harmless underwriters of the Registrable Securities, their officers and directors and each Person who controls such underwriters within the meaning of either Section 15 of the Exchange Act or section 20 of the Exchange Act or substantially the same basis as that of the indemnification of the Company provided in this <u>Section 1.07</u>. As a condition to including Registrable Securities in any registration statement filed in accordance with <u>Article 1</u>, the Company may require that it shall have received an undertaking reasonably satisfactory to it from any underwriter to indemnify and hold it harmless to the extent customarily provided by underwriters with respect to similar securities. No Registering Stockholder shall be liable under this <u>Section 1.07</u> for any Damages in excess of the net proceeds realized by such Stockholder in the sale of Registrable Securities of such Stockholder to which such Damages relate.

Section 1.08. <u>Conduct of Indemnification Proceedings</u>. If any proceeding (including any governmental investigation) shall be instituted involving any Person in respect of which indemnity may be sought pursuant to this <u>Article 1</u>, such Person (an "*Indemnified Party*") shall promptly notify the Person against whom such indemnity may be sought (the "*Indemnifying Party*") in writing and the Indemnifying Party shall assume the defense thereof, including the employment of counsel reasonably satisfactory to such Indemnified Party, and shall assume the payment of all fees and expenses, provided that the failure of any Indemnified Party so to notify the Indemnifying Party shall not relieve the Indemnifying Party of its obligations hereunder except to the extent that the Indemnifying Party is materially prejudiced by such failure to notify. In any such proceeding, any Indemnified Party shall have the right to retain its own counsel, but the fees and expenses of such counsel shall be at the expense of such Indemnified Party unless (i) the Indemnifying Party and the Indemnified Party shall have mutually agreed to the retention of such counsel or (ii) in the reasonable judgment of such Indemnified Party representation of both

parties by the same counsel would be inappropriate due to actual or potential differing interests between them. It is understood that, in connection with any proceeding or related proceedings in the same jurisdiction, the Indemnifying Party shall not be liable for the reasonable fees and expenses of more than one separate firm of attorneys (in addition to any local counsel) at any time for all such Indemnified Parties, and that all such fees and expenses shall be reimbursed as they are incurred. In the case of any such separate firm for the Indemnified Parties, such firm shall be designated in writing by the Indemnified Parties. The Indemnifying Party shall not be liable for any settlement of any proceeding effected without its written consent, but if settled with such consent, or if there be a final judgment for the plaintiff, the Indemnifying Party shall indemnify and hold harmless such Indemnified Parties from and against any Damages (to the extent stated above) by reason of such settlement or judgment. Without the prior written consent of the Indemnified Party shall effect any settlement of any pending in respect of which any Indemnified Party is or could have been a party and indemnified Party, unless such settlement includes an unconditional release of such Indemnified Party from all liability arising out of such proceeding.

Section 1.09. Contribution. If the indemnification provided for in this Article 1 is unavailable to the Indemnified Parties in respect of any Damages, then each such Indemnifying Party, in lieu of indemnifying such Indemnified Party, shall contribute to the amount paid or payable by such Indemnified Party as a result of such Damages (i) as between the Company and the Registering Stockholders holding Registrable Securities covered by a registration statement on the one hand and the underwriters on the other, in such proportion as is appropriate to reflect the relative benefits received by the Company and such Stockholders on the one hand and the underwriters on the other, from the offering of the Registrable Securities, or if such allocation is not permitted by applicable law, in such proportion as is appropriate to reflect not only the relative benefits but also the relative fault of the Company and such Stockholders on the one hand and of such underwriters on the other in connection with the statements or omissions that resulted in such Damages, as well as any other relevant equitable considerations and (ii) as between the Company on the one hand and each such Stockholder on the other, in such proportion as is appropriate to reflect the relative fault of the Company and of each such Stockholder in connection with such statements or omissions, as well as any other relevant equitable considerations. The relative benefits received by the Company and such Stockholders on the one hand and such underwriters on the other shall be deemed to be in the same proportion as the total proceeds from the offering (net of underwriting discounts and commissions but before deducting expenses) received by the Company and such Stockholders bear to the total underwriting discounts and commissions received by such underwriters, in each case as set forth in the table on the cover page of the prospectus. The relative fault of the Company and such Stockholders on the one hand and of such underwriters on the other shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission or alleged omission to state a material fact relates to information supplied by the Company and such Stockholders or by such underwriters. The relative fault of the Company on the one hand and of each such Stockholder on the other shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission or alleged omission to state a material fact relates to information supplied by such party, and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such statement or omission.

The Company and the Registering Stockholders agree that it would not be just and equitable if contribution pursuant to this <u>Section 1.09</u> were determined by pro rata allocation (even if the underwriters were treated as one entity for such purpose) or by any other method of allocation that does not take account of the equitable considerations referred to in the immediately preceding paragraph. The amount paid or payable by an Indemnified Party as a result of the Damages referred to in the immediately preceding paragraph shall be deemed to include, subject to the limitations set forth above, any legal or other expenses reasonably incurred by such Indemnified Party in connection with investigating or defending any such action or claim. Notwithstanding the provisions of this <u>Section 1.09</u>, no underwriter shall be required to contribute any amount in excess of the amount by which the total price at which the Registrable Securities underwritten by it and distributed to the public were offered to the public exceeds the amount of any Damages that such underwriter shall be required to contribute any amount in excess of the amount of any Damages that such underwriter has otherwise been required to pay by reason of such untrue or alleged untrue statement or omission or alleged omission, and no Registering Stockholder shall be required to which such Damages relate exceeds the amount of any Damages that such Stockholder in the sale of Registrable Securities of such Stockholder to which such Damages relate exceeds the amount of any Damages that such Stockholder has otherwise been required to pay by reason or alleged omission. No Person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution from any Person who was not guilty of such fraudulent misrepresentation. Each Registering Stockholder's obligation to contribute pursuant to this <u>Section 1.09</u> is several in the proportion that the proceeds of the offering received by such Stockholder s and no

Section 1.10. <u>Participation in Public Offering</u>. No Stockholder will be permitted to require registration of any Registrable Securities in any Public Offering hereunder unless such Stockholder (a) agrees to sell such Registrable Securities on the basis provided in any underwriting arrangements approved by the Persons entitled hereunder to approve such arrangements and (b) completes and executes all questionnaires, powers of attorney, indemnities, underwriting agreements and other documents reasonably required under the terms of such underwriting arrangements and the provisions of this Agreement in respect of registration rights.

Section 1.11. Other Indemnification. Indemnification similar to that specified herein (with appropriate modifications) shall be given by the Company and each Stockholder participating therein with respect to any required registration or other qualification of securities under any federal or state law or regulation or governmental authority other than the Securities Act.

Section 1.12. <u>Cooperation by the Company</u>. If any Stockholder shall transfer any Registrable Securities pursuant to Rule 144 of the Securities Act, the Company shall cooperate, to the extent commercially reasonable, with such Stockholder and shall provide to such Stockholder such information as such Stockholder shall reasonably request.

Section 1.13. <u>S-8 Registration Following Initial Public Offering</u>. The Company shall file a registration statement on Form S-8 in accordance with applicable securities laws within 180 days after the Initial Public Offering, which registration statement will cover the Common Shares issuable upon exercise of employee options then outstanding.

ARTICLE 2 LEGENDS

Section 2.01. Legends. Upon the written request of Oak Hill or any Other Stockholder in connection with a Transfer of Company Securities pursuant to (i) an effective registration statement under the Securities Act or (ii) another exemption from registration under the Securities Act, the Company shall or shall instruct the Company's transfer agent to replace such Stockholder's certificate with certificates not bearing any legend regarding restrictions on Transfer under the Securities Act (and, with respect to any Other Stockholder, restrictions on Transfer set forth in this Agreement).

ARTICLE 3 DEFINITIONS

The following terms, as used herein, have the following meanings:

Section 3.01. "Affiliate" means, with respect to any Person, any other Person directly or indirectly controlling, controlled by or under common control with such Person; *provided* that no securityholder of the Company shall be deemed an Affiliate of any other securityholder solely by reason of an investment in the Company. For the purpose of this definition, the term "*control*" (including, with correlative meanings, the terms "*controlling*", "*controlled by*" and "*under common control with*"), as used with respect to any Person, shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by contract or otherwise.

Section 3.02. "Business Day" means any day except a Saturday, Sunday or other day on which commercial banks in New York City are authorized by law to close.

Section 3.03. "<u>Company Securities</u>" means (i) the common stock, (ii) any preferred stock, (iii) any other common stock issued by the Company and (iv) any securities convertible into or exchangeable for, or options, warrants or other rights to acquire, common stock or any other stock issued by the Company.

Section 3.04. "Oak Hill" means Oak Hill Capital Partners III, L.P. and Oak Hill Capital Management Partners III, L.P. and their permitted assignees as contemplated by Section 3.03.

Section 3.05. "Exchange Act" means the Securities Exchange Act of 1934, as amended.

Section 3.06. "FINRA" means the Financial Industry Regulatory Authority, Inc.

Section 3.07. "Form S-3" means a Form S-3 registration statement under the Securities Act, and any successor or similar form thereto.

Section 3.08. "Initial Public Offering" means the first Public Offering of common stock of the Company registered on Form S-1.

Section 3.09. "Other Stockholder" means all Stockholders other than Oak Hill.

Section 3.10. "Person" means an individual, corporation, limited liability company, partnership, association, trust or other entity or organization, including a government or political subdivision or an agency or instrumentality thereof.

Section 3.11. "Public Offering" means an underwritten public offering of Company Securities pursuant to an effective registration statement under the Securities Act, other than pursuant to a registration statement on Form S-4 or Form S-8 or any similar or successor form.

Section 3.12. "<u>Registrable Securities</u>" means, at any time, any Company Securities held by any Stockholder until (i) a registration statement covering such Company Securities has been declared effective by the SEC and such Company Securities have been disposed of pursuant to such effective registration statement, (ii) such Company Securities are sold under circumstances in which all of the applicable conditions of Rule 144 (or any similar provisions then in force) under the Securities Act are met or (iii) such Company Securities are otherwise Transferred, the Company has delivered a new certificate or other evidence of ownership for such Company Securities not bearing a legend regarding restrictions on Transfer under the Securities Act (and, with respect to any Other Stockholder, restrictions on Transfer set forth in this Agreement) and such Company Securities may be resold without subsequent registration under the Securities Act.

Section 3.13. "Registration Expenses" means any and all expenses incident to the performance of or compliance with any registration or marketing of securities, including all (i) registration and filing fees, and all other fees and expenses payable in connection with the listing of securities on any securities exchange or automated interdealer quotation system, (ii) fees and expenses of compliance with any securities or "blue sky" laws (including reasonable fees and disbursements of counsel in connection with "blue sky" qualifications of the securities registered), (iii) expenses in connection with the preparation, printing, mailing and delivery of any registration statements, prospectuses and other documents in connection therewith and any amendments or supplements thereto, (iv) security engraving and printing expenses, (v) internal expenses of the Company (including, without limitation, all salaries and expenses of its officers and employees performing legal or accounting duties), (vi) reasonable fees and disbursements of counsel for the Company and customary fees and expenses for independent certified public accountants retained by the Company (including the expenses relating to any comfort letters or costs associated with the delivery by independent certified public accountants of any comfort letters), (vii) reasonable fees and expenses of any special experts retained by the Company in connection with such registration, (viii) reasonable fees and out-of-pocket expenses of counsel to the Stockholders participating in the offering selected (A) by Oak Hill, in the case of any offering in which Oak Hill participates, or (B) in any other case, by the Stockholders holding the majority of the Registrable Securities to be sold for the account of all Stockholders in the offering, (ix) fees and expenses in connection with any review by FINRA of the underwriting arrangements or other terms of the offering, and all fees and expenses of any "qualified independent underwriter," including the fees and expenses of any counsel thereto, (x) fees and disbursements of underwriters customarily paid by issuers or sellers of securities, but excluding any underwriting fees, discounts and commissions attributable to the sale of Registrable Securities, (xi) costs of printing and producing any agreements among underwriters, underwriting agreements, any "blue sky" or legal investment memoranda and any selling agreements and other documents in connection with the offering, sale or delivery of the Registrable Securities, (xii) transfer agents' and registrars' fees and expenses and the fees and

expense of any other agent or trustee appointed in connection with such offering, (xiii) expenses relating to any analyst or investor presentations or any "road shows" undertaken in connection with the registration, marketing or selling of the Registrable Securities and (xiv) fees and expenses payable in connection with any ratings of the Registrable Securities, including expenses relating to any presentations to rating agencies.

Section 3.14. "Required Stockholders" means Stockholders of at least a majority in number of Registrable Securities.

Section 3.15. "SEC" means the Securities and Exchange Commission.

Section 3.16. "Securities Act" means the Securities Act of 1933, as amended.

Section 3.17. "<u>Transfer</u>" means, with respect to any Company Securities, (i) when used as a verb, to sell, assign, dispose of, exchange, pledge, encumber, hypothecate or otherwise transfer such Company Securities or any participation or interest therein, whether directly or indirectly, or agree or commit to do any of the foregoing and (ii) when used as a noun, a direct or indirect sale, assignment, disposition, exchange, pledge, encumbrance, hypothecation, or other transfer of such Company Securities or any participation or interest therein or any agreement or commitment to do any of the foregoing.

Section 3.18. Other Definitional and Interpretive Matters. Unless otherwise expressly provided, for purposes of this Agreement, the following rules of interpretation shall apply:

(a) <u>Gender and Number</u>. Any reference in this Agreement to gender shall include all genders, and words imparting the singular number only shall include the plural and vice versa.

(b) <u>Headings</u>. The provision of a Table of Contents, the division of this Agreement into Articles, Sections and other subdivisions and the insertion of headings are for convenience of reference only and shall not affect or be utilized in construing or interpreting this Agreement. All references in this Agreement to any "Section" are to the corresponding Section of this Agreement unless otherwise specified.

(c) <u>Herein</u>. The words such as "herein," "hereinafter," "hereof," and "hereunder" refer to this Agreement as a whole and not merely to a subdivision in which such words appear unless the context otherwise requires.

(d) <u>Including</u>. Wherever the word "include," "includes," or "including" is used in this Agreement, it shall be deemed to be followed by the words "without limitation".

ARTICLE 4 MISCELLANEOUS

Section 4.01. <u>Termination</u>. All rights and obligations of the Company hereunder shall terminate on the date on which no Registrable Securities are outstanding.

Section 4.02. <u>Termination of Stockholders' Agreement</u>. The parties acknowledge and agree, that upon consummation of the Initial Public Offering, the Stockholders' Agreement of the Company, dated as of June 1, 2010, (including, for the purpose of clarity, Article IV thereof) shall automatically terminate and be of no further force or effect.

Section 4.03. <u>Amendment and Waiver</u>. This Agreement may not be amended except by an instrument in writing signed on behalf of each of the Company, the Required Stockholders and any Stockholder that would be materially and disproportionately affected by such an amendment. Any party hereto may waive any right of such party hereunder by an instrument in writing signed by such party and delivered to the other parties. The failure of any party to enforce any of the provisions of this Agreement shall in no way be construed as a waiver of such provisions and shall not affect the right of such party thereafter to enforce each and every provision of this Agreement in accordance with its terms.

Section 4.04. <u>Successors and Assigns</u>. This Agreement shall not inure to the benefit of, or be binding on, or be assignable or transferable by any Stockholder to, any Person to the extent such Person acquires Company Securities in, or at any time following, the Initial Public Offering; *provided*, *however*, that in connection with any Transfer of Company Securities by Oak Hill (i) to any of its Affiliates or (ii) in connection with any Transfer of Company Securities by Oak Hill (i) to any of its Affiliates or (ii) in connection with any Transfer of Company Securities by Oak Hill or any portion of its rights under this Agreement to any transferee who agrees to be bound by this Agreement.

Section 4.05. <u>Severability</u>. If any term, provision, covenant or restriction of this Agreement is held by a court of competent jurisdiction or other authority to be invalid, void or unenforceable, the remainder of the terms, provisions, covenants and restrictions of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated. Upon such a determination, the parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner so that the transactions contemplated hereby be consummated as originally contemplated to the fullest extent possible.

Section 4.06. Entire Agreement. Except as otherwise expressly set forth herein, this Agreement, together with the several agreements and other documents and instruments referred to herein or therein or annexed hereto or thereto, embody the complete agreement and understanding among the parties hereto with respect to the subject matter hereof and supersede and preempt any prior understandings, agreements or representations by or among the parties, written or oral, that may have related to the subject matter hereof in any way.

Section 4.07. <u>Counterparts</u>; Execution by Facsimile Signature. This Agreement may be executed in any number of counterparts, each of which shall be an original, but all of which together shall constitute one instrument. This Agreement may be executed by facsimile signature(s).

Section 4.08. Notices. All notices required or permitted hereunder shall be in writing and shall be deemed effectively given (i) upon personal delivery to the party to be notified, (ii) when sent by confirmed facsimile if sent during normal business hours of the recipient, if not, then on the next Business Day or (iii) one Business Day after deposit with a nationally recognized overnight courier, specifying next day delivery, with written verification of receipt. All communications shall be sent to the addresses set forth below or such other address or facsimile number as a party may from time to time specify by notice to the other parties hereto:

If to the Company, at:

Dave & Buster's Entertainment, Inc. 2481 Manana Drive Dallas, Texas 75220 Attention: Jay L. Tobin, Esq. Fax: (214) 357-1536

With a copy which shall not constitute notice to:

Oak Hill Capital Management, LLC 65 East 55th Street, 32nd Floor New York, NY 10022 Attention: John R. Monsky, Esq. Fax: (212) 527-8450

and

Weil, Gotshal & Manges LLP 767 Fifth Avenue New York, New York 10153 Attention: Douglas P. Warner, Esq. Fax: (212) 310-8007

If to any Other Stockholder, to such Stockholder's address as set forth in the register of stockholders maintained by the Company.

Section 4.09. <u>Governing Law</u>. This Agreement, and all claims or causes of action (whether at law, in equity, in contract, in tort or otherwise) based upon, arising out of, related to or otherwise in connection with this Agreement or the transactions contemplated hereby, shall be exclusively governed by, and construed in accordance with, the internal laws of the State of Delaware, without giving effect to principles or rules of conflict of laws to the extent such principles or rules would require or permit the application of laws of another jurisdiction (including any claim or cause of action based upon, arising out of or related to any representation or warranty made in or in connection with this Agreement or as an inducement to enter into this Agreement).

Section 4.10. <u>Consent to Jurisdiction</u>. The parties hereby agree that any suit, action or proceeding (whether at law, in equity, in contract, in tort or otherwise) seeking to enforce any provision of, or based on any matter arising out of or in connection with, this Agreement or the transactions contemplated hereby (including any claim or cause of action based upon, arising out of or related to any representation or warranty made in or in connection with this Agreement or as an inducement to enter into this Agreement) shall be exclusively brought in the United States District Court for the Southern District of New York or any New York State court sitting in New York City, so long as one of such courts shall have subject matter jurisdiction over such suit,

action or proceeding, and that any case of action arising out of this Agreement shall be deemed to have arisen from a transaction of business in the State of New York, and each of the parties hereby irrevocably consents to the exclusive jurisdiction of such courts (and of the appropriate appellate courts therefrom) in any such suit, action or proceeding and irrevocably waives, to the fullest extent permitted by law, any objection that it may now or hereafter have to the laying of the venue of any such suit, action or proceeding in any such court or that any such suit, action or proceeding which is brought in any such court has been brought in an inconvenient form. Process in any such suit, action or proceeding may be served on any party anywhere in the world, whether within or without the jurisdiction of any such court. Without limiting the foregoing, each party agrees that service of process on such party as provided in this <u>Section 4.10</u> shall be deemed effective service of process on such party.

Section 4.11. <u>Waiver of Jury Trial</u>. EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING (WHETHER AT LAW, IN EQUITY, IN CONTRACT, IN TORT OR OTHERWISE) ARISING OUT OF OR RELATED TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.

[Remainder of this page is intentionally left blank.]

IN WITNESS WHEREOF, each of the undersigned has caused this Agreement to be duly executed and delivered as of the date first written above.

DAVE & BUSTER'S ENTERTAINMENT, INC.

By:

Name: Stephen M. King Title: Chief Executive Officer

OAK HILL CAPITAL PARTNERS III, L.P.

- By: OHCP GenPar III, L.P., its General Partner
- By: OHCP MGP Partners III, L.P., its General Partner
- By: OHCP MGP III, Ltd., its General Partner

By:

Name: John R. Monsky Title:

OAK HILL CAPITAL MANAGEMENT PARTNERS III, L.P.

- By: OHCP GenPar III, L.P., its General Partner
- By: OHCP MGP Partners III, L.P., its General Partner
- By: OHCP MGP III, Ltd., its General Partner

By:

Name: John R. Monsky Title:

Alan J. Lacy

David A. Jones, individually, and on behalf of, Brenton Alan Kindle, Brooke Nicole Kindle Stephens, Leslie Ann Jones Acosta, Jeffrey David Jones, Dana Michele Jones Smith, David A. Jones 2006 Grandchildren's Trust Dated 12/30/2006 FBO Davis A. Kindle, David A. Jones 2006 Grandchildren's Trust Dated 12/30/2006 FBO Antonio Acosta III, David A. Jones 2006 Grandchildren's Trust Dated 12/30/2006 FBO Dillon A. Jones, David A. Jones 2006 Grandchildren's Trust Dated 12/30/2006 FBO H. Jones Scherer, David A. Jones 2006 Grandchildren's Trust Dated 12/30/2006 FBO Jackson D. Stephens, David A. Jones 2006 Grandchildren's Trust Dated 12/30/2006 FBO Turner Clark Smith, David A. Jones 2006 Grandchildren's Trust Dated 12/30/2006 FBO Tyler J. Kindle, David A. Jones 2006 Grandchildren's Trust Dated 12/30/2006 FBO W. Rhys Smith David A. Jones 2013 Grandchildren's Trust Dated 6/30/2013 FBO Madeline Grace Stephens, and David A. Jones 2013 Grandchildren's Trust Dated 6/30/2013 FBO Madison Alena Jones

Kevin M. Sheehan

Jonathan S. Halkyard

Michael J. Griffith

Stephen M. King, individually, and on behalf of,

Steve and Shauna King Investment Partnership L.P.

[SIGNATURE PAGE TO THE REGISTRATION RIGHTS AGREEMENT]

Dolf Berle

Jay L. Tobin

Brian A. Jenkins

John P. Gleason III

Margo L. Manning

Edward J. Forler

Michael Metzinger

Gregory Clore

Joseph DeProspero

Lisa Warren

[SIGNATURE PAGE TO THE REGISTRATION RIGHTS AGREEMENT]

Dave & Buster's Entertainment, Inc.

2014 Omnibus Incentive Plan

		Page
ARTICLE 1.	ESTABLISHMENT & PURPOSE	1
1.1	Establishment	1
1.2	Purpose of the Plan	1
ARTICLE 2.	DEFINITIONS	1
ARTICLE 3.	ADMINISTRATION	5
3.1	Authority of the Committee	5
3.2	Delegation	6
ARTICLE 4.	ELIGIBILITY AND PARTICIPATION	6
4.1	Eligibility	6
4.2	Type of Awards	6
ARTICLE 5.	SHARES SUBJECT TO THE PLAN AND MAXIMUM AWARDS	7
5.1	General	7
5.2	Annual Award Limits	7
5.3	Additional Shares	7
ARTICLE 6.	STOCK OPTIONS	7
6.1	Grant of Options	7
6.2	Terms of Option Grant	8
6.3	Option Term	8
6.4	Time of Exercise	8
6.5	Method of Exercise	8
6.6	Limitations on Incentive Stock Options	8
6.7	Performance Goals	9
ARTICLE 7.	STOCK APPRECIATION RIGHTS	9
7.1	Grant of Stock Appreciation Rights	9
7.2	Terms of Stock Appreciation Right	9
7.3	Tandem Stock Appreciation Rights and Options	9
ARTICLE 8.	RESTRICTED STOCK	9
8.1	Grant of Restricted Stock	9
8.2	Terms of Restricted Stock Awards	10
8.3	Voting and Dividend Rights	10
8.4	Performance Goals	10
8.5	Section 83(b) Election	10

i

TABLE OF CONTENTS (continued)

_

		Page
ARTICLE 9.	OTHER STOCK-BASED AWARDS; CASH-BASED AWARDS	10
9.1	Other Stock-Based Awards	10
9.2	Cash-Based Awards	11
ARTICLE 10.	PERFORMANCE-BASED COMPENSATION	11
10.1	Grant of Performance-Based Compensation	11
10.2	Performance Measures	11
10.3	Establishment of Performance Goals for Covered Employees	12
10.4	Adjustment of Performance-Based Compensation	12
10.5	Certification of Performance	12
10.6	Interpretation	12
ARTICLE 11.	COMPLIANCE WITH SECTION 409A OF THE CODE AND SECTION 457A OF THE CODE	13
11.1	General	13
11.2	Payments to Specified Employees	13
11.3	Separation from Service	13
11.4	Section 457A	13
ARTICLE 12.	ADJUSTMENTS	14
12.1	Adjustments in Authorized Shares	14
12.2	Change of Control	14
ARTICLE 13.	DURATION, AMENDMENT, MODIFICATION, SUSPENSION AND TERMINATION	15
13.1	Duration of the Plan	15
13.2	Amendment, Modification, Suspension and Termination of Plan	15
ARTICLE 14.	GENERAL PROVISIONS	15
14.1	No Right to Service	15
14.2	Settlement of Awards; Fractional Shares	15
14.3	Tax Withholding	16
14.4	No Guarantees Regarding Tax Treatment	16
14.5	Non-Transferability of Awards	16
14.6	Conditions and Restrictions on Shares	16
14.7	Compliance with Law	16

ii

TABLE OF CONTENTS (continued)

_

		Page
14.8	Awards to Non-U.S. Employees or Directors	17
14.9	Rights as a Shareholder	17
14.10	Severability	17
14.11	Unfunded Plan	18
14.12	No Constraint on Corporate Action	18
14.13	Successors	18
14.14	Governing Law; Jurisdiction; Waiver of Jury Trial	18
14.15	Waiver of Certain Claims	19
14.16	Data Protection	19
14.17	Accounting Restatement	19
14.18	Effective Date	19
14.19	Shareholder Approval	19

iii

Dave & Buster's Entertainment, Inc.

2014 Omnibus Incentive Plan

Article 1. Establishment & Purpose

1.1 Establishment. Dave & Buster's Entertainment, Inc., a Delaware corporation (the "<u>Company</u>") hereby establishes the Dave & Buster's Entertainment, Inc. 2014 Omnibus Incentive Plan (hereinafter referred to as the "<u>Plan</u>") as set forth in this document.

1.2 Purpose of the Plan. The purpose of the Plan is to attract, retain and motivate officers, employees, non-employee directors and consultants providing services to the Company and its Subsidiaries and Affiliates and to promote the success of the Company's business by providing participants with appropriate incentives.

Article 2. Definitions

Whenever capitalized in the Plan, the following terms shall have the meanings set forth below.

2.1 "<u>Affiliate</u>" means any entity that the Company, either directly or indirectly, is in common control with, is controlled by or controls, or any entity in which the Company has a substantial equity interest, direct or indirect; <u>provided</u>, <u>however</u>, to the extent that Awards must cover "service recipient stock" in order to comply with Section 409A of the Code, "Affiliate" shall be limited to those entities which could qualify as an "eligible issuer" under Section 409A of the Code.

2.2 "Annual Award Limit" shall have the meaning set forth in Section 5.2.

2.3 "Award" means any Option, Stock Appreciation Right, Restricted Stock, Other Stock-Based Award or Cash-Based Award that is granted under the Plan.

2.4 "<u>Award Agreement</u>" means either (a) a written agreement entered into by the Company and a Participant setting forth the terms and conditions applicable to an Award, or (b) a written statement issued by the Company to a Participant describing the terms and conditions applicable to an Award.

2.5 "Beneficial Owner" or "Beneficial Ownership" shall have the meaning ascribed to such term in Rule 13d-3 of the General Rules and Regulations under the Exchange Act.

2.6 "Board" means the Board of Directors of the Company.

2.7 "Cash-Based Award" means any right granted under Section 9.2 of the Plan.

2.8 "Change of Control" unless otherwise specified in the Award Agreement, means the occurrence of any of the following events:

(a) Any Person, other than OH, becomes the Beneficial Owner of thirty percent (30%) or more of the combined voting power of the then

outstanding voting securities of the Company entitled to vote generally in the election of the members of the Board (the "<u>Outstanding</u> <u>Company Voting Securities</u>"); provided, however, that for purposes of this Section 2.8, the following acquisitions shall not constitute a Change of Control: (i) any acquisition directly from the Company, including without limitation, a public offering of securities; (ii) any acquisition by the Company or any of its Subsidiaries or Affiliates; (iii) any acquisition by any employee benefit plan or related trust sponsored or maintained by the Company or any of its Subsidiaries or Affiliates; or (iv) any acquisition by any Person pursuant to a transaction which complies with clauses (i), (ii), and (iii) of Section 2.8(c).

- (b) Individuals who constitute the Board as of the Effective Date (the "Incumbent Board") cease for any reason to constitute at least a majority of the Board; provided, that any individual becoming a member of the Board subsequent to the Effective Date whose election to the Board, or nomination for election by one or more of the Company's shareholders, was approved by a vote of at least a majority of the members of the Board then comprising the Incumbent Board, shall be considered as though such individual were a member of the Incumbent Board, but excluding, for this purpose, any such individual whose initial assumption of office is in connection with an actual or threatened election contest relating to the election or removal of any members of the Board or other actual or threatened solicitation of proxies or consents by or on behalf of a Person other than the Board; provided, further, that any individual becoming a member of the Board subsequent to the Effective Date who was designated as a Board member by OH shall be considered as though such individual were a member of the Board subsequent to the Incumbent Board.
- (c) Consummation of a reorganization, merger, amalgamation, statutory share exchange, consolidation or like event to which the Company is a party or a sale or other disposition of all or substantially all of the assets of the Company (a "Business Combination"), unless, following such Business Combination: (i) all or substantially all of the individuals and entities who were the Beneficial Owners of Outstanding Company Voting Securities immediately prior to such Business Combination are the Beneficial Owners, directly or indirectly, of more than fifty percent (50%) of the combined voting power of the outstanding voting securities entitled to vote generally in the election of directors (or election of members of a comparable governing body) of the entity resulting from the Business Combination (including, without limitation, an entity which as a result of such transaction owns all or substantially all of the Company or all or substantially all of the Company's assets either directly or through one or more Subsidiaries) (the "Successor Entity") in substantially the same proportions as their ownership immediately prior to such Business Combination of the Outstanding Company Voting Securities; (ii) no Person (excluding any Successor Entity or any employee benefit plan or

related trust of the Company, such Successor Entity, or any of their Subsidiaries) is the Beneficial Owner, directly or indirectly, of thirty percent (30%) or more of the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors (or comparable governing body) of the Successor Entity, except to the extent that such ownership existed prior to the Business Combination; and (iii) at least a majority of the members of the board of directors (or comparable governing body) of the Successor Entity, except to be members (or comparable governing body) of the Successor Entity, except to be extent that such ownership existed prior to the Business Combination; and (iii) at least a majority of the members of the board of directors (or comparable governing body) of the Successor Entity were members of the Incumbent Board (including persons deemed to be members of the Incumbent Board by reason of the proviso of Section 2.8(b)) at the time of the execution of the initial agreement or of the action of the Board providing for such Business Combination; or

(d) Approval by the shareholders of the Company of a complete liquidation or dissolution of the Company.

To the extent necessary to comply with Section 409A of the Code with respect to the payment of deferred compensation, "Change of Control" shall be limited to a "change in control event" as defined under Section 409A of the Code.

2.9 "Code" means the U.S. Internal Revenue Code of 1986, as amended from time to time.

2.10 "<u>Committee</u>" means the Compensation Committee of the Board, the Plan Subcommittee of the Compensation Committee of the Board or any other committee or subcommittee designated by the Board to administer the Plan. To the extent applicable, the Committee shall have at least two members, each of whom shall be (a) a Non-Employee Director, (b) an Outside Director, and (c) an "independent director" within the meaning of the listing requirements of any exchange on which the Company is listed.

2.11 "Company" shall have the meaning set forth in Section 1.1.

2.12 "<u>Consultant</u>" means any person who provides bona fide services to the Company or any Subsidiary or Affiliate as a consultant or advisor, excluding any Employee or Director.

2.13 "<u>Covered Employee</u>" means for any Plan Year, a Participant designated by the Company as a potential "covered employee" as such term is defined in Section 162(m) of the Code.

2.14 "Director" means a member of the Board who is not an Employee.

2.15 "Effective Date" shall have the meaning set forth in Section 14.17.

2.16 "<u>Employee</u>" means an officer or other employee of the Company, a Subsidiary or Affiliate, including a member of the Board who is an employee of the Company, a Subsidiary or Affiliate and individuals who have accepted a written offer of employment with the Company, a Subsidiary or Affiliate.

2.17 "Exchange Act" means the Securities Exchange Act of 1934, as amended from time to time.

2.18 "Fair Market Value" means, as of any date, the per Share value determined as follows, in accordance with applicable provisions of Section 409A of the Code:

- (a) At the Committee's discretion, any of (i) the average of the high and low trading price, (ii) the average of the high and low trading price for the preceding 30 days, (iii) the closing price, in each case, as reported on NASDAQ or any other recognized national exchange or any established over-the-counter trading system on which dealings take place, or, if no trades were made on any such day, the immediately preceding day on which trades were made or (iv) as otherwise reasonably determined by the Committee in good faith based on actual transactions in Shares; or
- (b) In the absence of an established market for the Shares of the type described in (a) above, the per Share Fair Market Value thereof shall be determined by the Committee in good faith.

2.19 "Incentive Stock Option" means an Option intended to meet the requirements of an incentive stock option as defined in Section 422 of the Code and designated as an Incentive Stock Option.

2.20 "Non-Employee Director" means a person defined in Rule 16b-3(b)(3) promulgated by the Securities and Exchange Commission under the Exchange Act, or any successor definition adopted by the Securities and Exchange Commission.

2.21 "Nonqualified Stock Option" means an Option that is not an Incentive Stock Option.

2.22 "OH" means, collectively, Oak Hill Capital Partners III, L.P. and Oak Hill Capital Management Partners III, L.P. and their respective affiliated funds and investment vehicles.

2.23 "Other Stock-Based Award" means any right granted under Section 9.1 of the Plan.

2.24 "<u>Option</u>" means any stock option granted under Article 6 of the Plan.

2.25 "Option Price" means the purchase price per Share subject to an Option, as determined pursuant to Section 6.2 of the Plan.

2.26 "Outside Director" means a member of the Board who is an "outside director" within the meaning of Section 162(m) of the Code and the regulations promulgated thereunder.

4

2.27 "Participant" means any eligible person as set forth in Section 4.1 to whom an Award is granted.

2.28 "<u>Performance-Based Compensation</u>" means compensation under an Award that is intended to constitute "qualified performance-based compensation" within the meaning of the regulations promulgated under Section 162(m) of Code or any successor provision.

2.29 "Performance Measures" means measures as described in Section 10.2 on which the performance goals are based in order to qualify Awards as Performance-Based Compensation.

2.30 "Performance Period" means the period of time during which the performance goals must be met in order to determine the degree of payout and/or vesting with respect to an Award.

2.31 "Person" shall have the meaning ascribed to such term in Section 3(a)(9) of the Exchange Act and used in Sections 13(d) and 14(d) thereof, including a "group" as defined in Section 13(d) thereof.

2.32 "Plan" shall have the meaning set forth in Section 1.1.

2.33 "Plan Year" means the applicable fiscal year of the Company.

2.34 "<u>Restricted Stock</u>" means any Award granted under Article 8 of the Plan.

2.35 "Restriction Period" means the period during which Restricted Stock awarded under Article 8 of the Plan is subject to forfeiture.

2.36 "Service" means service as an Employee, Director or Consultant.

2.37 "Share" means a share of common stock of the Company, par value \$0.01 per share, or such other class or kind of shares or other securities resulting from the application of Article 12 hereof.

2.38 "Stock Appreciation Right" means any right granted under Article 7 of the Plan.

2.39 "Subsidiary" means any corporation, partnership, limited liability company or other legal entity of which the Company, directly or indirectly, owns stock or other equity interests possessing fifty percent (50%) or more of the total combined voting power of all classes of stock or other equity interests (as determined in a manner consistent with Section 409A of the Code, if applicable).

2.40 "<u>Ten Percent Shareholder</u>" means a person who on any given date owns, either directly or indirectly (taking into account the attribution rules contained in Section 424(d) of the Code), stock possessing more than ten percent (10%) of the total combined voting power of all classes of stock of the Company or a Subsidiary or Affiliate.

Article 3. Administration

3.1 Authority of the Committee. The Plan shall be administered by the Committee, which shall have full power to interpret and administer the Plan and Award Agreements and full

authority to select the Employees, Directors and Consultants to whom Awards will be granted, and to determine the type and amount of Awards to be granted to each such Employee, Director or Consultant, and the terms and conditions of Awards and Award Agreements. Without limiting the generality of the foregoing, the Committee may, in its sole discretion but subject to the limitations in Article 13, clarify, construe or resolve any ambiguity in any provision of the Plan or any Award Agreement, extend the term or period of exercisability of any Awards, or waive any terms or conditions applicable to any Award. Also subject to the limitations in Article 13, Awards may, in the discretion of the Committee, be made under the Plan in assumption of, or in substitution for, outstanding awards previously granted by the Company or any of its Subsidiaries or Affiliates or a company acquired by the Company or with which the Company combines. The Committee shall have full and exclusive discretionary power to adopt rules, forms, instruments, and guidelines for administering the Plan as the Committee deems necessary or proper. Notwithstanding anything in this Section 3.1 to the contrary, the Board, or any other committee or sub-committee established by the Board, is hereby authorized (in addition to any necessary action by the Committee) to grant or approve Awards as necessary to satisfy the requirements of Section 16 of the Exchange Act and the rules and regulations thereunder and to act in lieu of the Committee or by the Board (or any other committee or sub-committee or sub-com

3.2 Delegation. The Committee may delegate to one or more of its members, one or more officers of the Company or any of its Subsidiaries or Affiliates, and one or more agents or advisors such administrative duties or powers as it may deem advisable; <u>provided</u>, that the Committee shall not delegate to officers of the Company or any of its Subsidiaries or Affiliates the power to make grants of Awards to officers of the Company or any of its Subsidiaries or Affiliates; <u>provided</u>, <u>further</u>, that no delegation shall be permitted under the Plan that is prohibited by applicable law.

Article 4. Eligibility and Participation

4.1 Eligibility. Participants will consist of such Employees, Directors and Consultants as the Committee in its sole discretion determines and whom the Committee may designate from time to time to receive Awards. Designation of a Participant in any year shall not require the Committee to designate such person to receive an Award in any other year or, once designated, to receive the same type or amount of Award as granted to the Participant in any other year.

4.2 Type of Awards. Awards under the Plan may be granted in any one or a combination of: (a) Options, (b) Stock Appreciation Rights, (c) Restricted Stock, (d) Other Stock-Based Awards, and (e) Cash-Based Awards. The Plan sets forth the types of performance goals and sets forth procedural requirements to permit the Company to design Awards that qualify as Performance-Based Compensation, as described in Article 10 hereof. Awards granted under the Plan shall be evidenced by Award Agreements (which need not be identical) that provide additional terms and conditions associated with such Awards, as determined by the Committee in its sole discretion; provided, however, that in the event of any conflict between the provisions of the Plan and any such Award Agreement, the provisions of the Plan shall prevail.

Article 5. Shares Subject to the Plan and Maximum Awards

5.1 General. Subject to adjustment as provided in Article 12 hereof, the maximum number of Shares available for issuance to Participants pursuant to Awards under the Plan shall be [] Shares. The number of Shares available for granting Incentive Stock Options under the Plan shall not exceed [] Shares, subject to Article 12 hereof and the provisions of Sections 422 or 424 of the Code and any successor provisions. The Shares available for issuance under the Plan may consist, in whole or in part, of authorized and unissued Shares or treasury Shares. Any Shares delivered to or withheld by the Company as part or full payment for the purchase price of an Award, or to the extent the Committee determines that the availability of Incentive Stock Options will not be compromised, or to satisfy the Company's withholding obligation with respect to an Award, shall again be available for Awards; provided, however, that such Shares shall continue to be counted as outstanding for purposes of determining whether an Annual Award Limit has been attained.

5.2 Annual Award Limits. The maximum number of Shares with respect to Awards denominated in Shares that may be granted to any Participant in any Plan Year shall be [] Shares, subject to adjustments made in accordance with Article 12 hereof, and the maximum value of cash payable with respect to Awards denominated in cash or property that may be granted to any Participant in any Plan Year shall be \$[], subject to adjustments made in accordance with Article 12 hereof (the "<u>Annual Award Limit</u>").

5.3 Additional Shares. In the event that any outstanding Award expires, is forfeited, cancelled or otherwise terminated without the issuance of Shares or is otherwise settled for cash, the Shares subject to such Award, to the extent of any such forfeiture, cancellation, expiration, termination or settlement for cash, shall again be available for Awards. If the Committee authorizes the assumption under the Plan, in connection with any merger, consolidation, acquisition of property or stock, or reorganization, of awards granted under another plan, such assumption shall not (a) reduce the maximum number of Shares available for issuance under the Plan or (b) be subject to or counted against a Participant's Annual Award Limit.

Article 6. Stock Options

6.1 Grant of Options. The Committee is hereby authorized to grant Options to Participants. Each Option shall permit a Participant to purchase from the Company a stated number of Shares at an Option Price established by the Committee, subject to the terms and conditions described in this Article 6 and to such additional terms and conditions, as established by the Committee, in its sole discretion, that are consistent with the provisions of the Plan. Options shall be designated as either Incentive Stock Options or Nonqualified Stock Options; provided, that Options granted to Directors and Consultants shall be Nonqualified Stock Options. None of the Committee, the Company, any of its Subsidiaries or Affiliates, or any of their employees and representatives shall be liable to any Participant or to any other Person if it is determined that an Option intended to be an Incentive Stock Option does not qualify as an Incentive Stock Option. Each Option shall be evidenced by an Award Agreement which shall state the number of Shares covered by such Option. Such Award Agreement shall conform to the requirements of the Plan, and may contain such other provisions, as the Committee shall deem advisable.

6.2 Terms of Option Grant. The Option Price shall be determined by the Committee at the time of grant, but shall not be less than one-hundred percent (100%) of the Fair Market Value of a Share on the date of grant. In the case of any Incentive Stock Option granted to a Ten Percent Shareholder, the Option Price shall not be less than one-hundred-ten percent (110%) of the Fair Market Value of a Share on the date of grant.

6.3 Option Term. The term of each Option shall be determined by the Committee at the time of grant and shall be stated in the Award Agreement, but in no event shall such term be greater than ten (10) years (or, in the case on an Incentive Stock Option granted to a Ten Percent Shareholder, five (5) years).

6.4 Time of Exercise. Options granted under this Article 6 shall be exercisable based on the terms and conditions as the Committee shall in each instance approve, which terms and conditions need not be the same for each grant or for each Participant.

6.5 Method of Exercise. Except as otherwise provided in the Plan or in an Award Agreement, an Option may be exercised for all, or from time to time any part, of the Shares for which it is then exercisable. For purposes of this Article 6, the exercise date of an Option shall be the later of the date a notice of exercise is received by the Company and, if applicable, the date payment is received by the Company pursuant to clauses (a), (b), (c) or (d) of the following sentence and pursuant to Section 14.3 hereof. The aggregate Option Price for the Shares as to which an Option is exercised shall be paid to the Company in full at the time of exercise at the election of the Participant (a) in cash or its equivalent (e.g., by cashier's check), (b) to the extent permitted by the Committee, in Shares (whether or not previously owned by the Participant) having a Fair Market Value equal to the aggregate Option Price for the Shares at such time, subject to such requirements as may be imposed by the Committee, (c) partly in cash and, to the extent permitted by the Committee, shares (as described in (b) above) or (d) if there is a public market for the Shares at such time, subject to such requirements as may be imposed by the Committee, self in the Shares obtained upon the exercise of the Option and to deliver promptly to the Committee, through the delivery of irrevocable instructions to a broker to sell Shares obtained upon the exercise of the Option and to deliver promptly to the Company an amount out of the proceeds of such sale equal to the aggregate Option Price for the Shares being purchased. The Committee may prescribe any other method of payment that it determines to be consistent with applicable law and the purpose of the Plan.

6.6 Limitations on Incentive Stock Options. Incentive Stock Options may be granted only to employees of the Company or of a "parent corporation" or "subsidiary corporation" (as such terms are defined in Section 424 of the Code) at the date of grant. If the aggregate Fair Market Value (generally determined as of the time the Option is granted) of the Shares with respect to which Incentive Stock Options are exercisable for the first time by a Participant during any calendar year under all plans of the Company and of any "parent corporation" or "subsidiary corporation" exceeds one hundred thousand dollars (\$100,000), the portion of such Incentive Stock Options exercisable for such excess value shall be treated as Nonqualified Stock Options. For purposes of the preceding sentence, Incentive Stock Options will be taken into account generally in the order in which they are granted. Each provision of the Plan and each Award Agreement relating to an Incentive Stock Option shall be construed so that each Incentive Stock Option shall be an incentive stock option as defined in Section 422 of the Code, and any provisions of the Award Agreement thereof that cannot be so construed shall be

disregarded; provided, however, to the extent any Option (or portion thereof) granted as an Incentive Stock Option fails to qualify as an Incentive Stock Option, such Option (or portion thereof) shall be treated as a Nonqualified Stock Option.

6.7 Performance Goals. The Committee may condition the grant of Options or the vesting of Options upon the Participant's achievement of one or more performance goal(s) (including the Participant's provision of Services for a designated time period), as specified in the Award Agreement. If the Participant fails to achieve the specified performance goal(s), the Committee shall not grant the Option to such Participant or the Option shall not vest, as applicable.

Article 7. Stock Appreciation Rights

7.1 Grant of Stock Appreciation Rights. The Committee is hereby authorized to grant Stock Appreciation Rights to Participants, including a grant of Stock Appreciation Rights in tandem with any Option at the same time such Option is granted (a "Tandem SAR"). Stock Appreciation Rights shall be evidenced by Award Agreements that shall conform to the requirements of the Plan and may contain such other provisions, as the Committee shall deem advisable. Subject to the terms of the Plan and any applicable Award Agreement, a Stock Appreciation Right granted under the Plan shall confer on the holder thereof a right to receive, upon exercise thereof, the excess of (a) the Fair Market Value of a specified number of Shares on the date of exercise over (b) the grant price of the right as specified by the Committee on the date of the grant. Such payment may be in the form of cash, Shares, other property or any combination thereof, as the Committee shall determine in its sole discretion.

7.2 Terms of Stock Appreciation Right. Subject to the terms of the Plan and any applicable Award Agreement, the grant price (which shall not be less than one hundred percent (100%) of the Fair Market Value of a Share on the date of grant), term, methods of exercise, methods of settlement, and any other terms and conditions of any Stock Appreciation Right shall be as determined by the Committee. The Committee may impose such other conditions or restrictions on the exercise of any Stock Appreciation Right as it may deem appropriate. No Stock Appreciation Right shall have a term of more than ten (10) years from the date of grant.

7.3 Tandem Stock Appreciation Rights and Options. A Tandem SAR shall be exercisable only to the extent that the related Option is exercisable and shall expire no later than the expiration of the related Option. Upon the exercise of all or a portion of a Tandem SAR, a Participant shall be required to forfeit the right to purchase an equivalent portion of the related Option (and, when a Share is purchased under the related Option, the Participant shall be required to forfeit an equivalent portion of the Stock Appreciation Right).

Article 8. Restricted Stock

8.1 Grant of Restricted Stock. An Award of Restricted Stock is a grant by the Committee of a specified number of Shares to the Participant, which Shares are subject to forfeiture upon the occurrence of specified events. Restricted Stock shall be evidenced by an Award Agreement, which shall conform to the requirements of the Plan and may contain such other provisions, as the Committee shall deem advisable.

8.2 Terms of Restricted Stock Awards. Each Award Agreement evidencing a Restricted Stock grant shall specify the period(s) of restriction, the number of Shares of Restricted Stock subject to the Award, the performance, employment or other conditions (including the termination of a Participant's Service whether due to death, disability or other reason) under which the Restricted Stock may be forfeited to the Company and such other provisions as the Committee shall determine. Any Restricted Stock granted under the Plan shall be evidenced in such manner as the Committee may deem appropriate, including book-entry registration or issuance of a stock certificate or certificates (in which case, the certificate(s) representing such Shares shall be legended as to sale, transfer, assignment, pledge or other encumbrances during the Restriction Period and deposited by the Participant, together with a stock power endorsed in blank, with the Company, to be held in escrow during the Restriction Period). At the end of the Restriction Period, assuming satisfaction of the applicable performance, employment or other conditions, the restrictions imposed hereunder and under the Award Agreement shall lapse with respect to the number of Shares of Restricted Stock as determined by the Committee, and the legend shall be removed and such number of Shares delivered to the Participant (or, where appropriate, the Participant's legal representative).

8.3 Voting and Dividend Rights. Unless otherwise provided in an Award Agreement, Participants shall have none of the rights of a stockholder of the Company with respect to Restricted Stock until the end of the Restricted Period; provided, that the Committee shall determine and set forth in a Participant's Award Agreement whether or not a Participant holding Restricted Stock granted hereunder shall have the right to exercise voting rights with respect to the Restricted Stock during the Restriction Period (the Committee may require a Participant to grant an irrevocable proxy and power of substitution); provided, however, that Participants shall have no right to receive dividends on a current basis with respect to the Restricted Stock during the Restriction Period.

8.4 Performance Goals. The Committee may condition the grant of Restricted Stock or the expiration of the Restriction Period upon the Participant's achievement of one or more performance goal(s) (including the Participant's provision of Services for a designated time period), as specified in the Award Agreement. If the Participant fails to achieve the specified performance goal(s), the Committee shall not grant the Restricted Stock to such Participant or the Participant shall forfeit the Award of Restricted Stock to the Company, as applicable.

8.5 Section 83(b) Election. If a Participant makes an election pursuant to Section 83(b) of the Code concerning Restricted Stock, the Participant shall be required to file promptly a copy of such election with the Company.

Article 9. Other Stock-Based Awards; Cash-Based Awards

9.1 Other Stock-Based Awards. The Committee, in its sole discretion, may grant Awards of Shares and Awards that are valued, in whole or in part, by reference to, or are otherwise based on the Fair Market Value of Shares (the "<u>Other Stock-Based Awards</u>"), including without limitation, restricted stock units and other phantom awards. Such Other Stock-Based Awards shall be in such form, and dependent on such conditions, as the Committee shall determine, including, without limitation, the right to receive one or more Shares (or the equivalent cash value of such Shares) upon the completion of a specified period of Service, the

occurrence of an event and/or the attainment of performance objectives. Other Stock-Based Awards may be granted alone or in addition to any other Awards granted under the Plan. Subject to the provisions of the Plan, the Committee shall determine to whom and when Other Stock-Based Awards will be made, the number of Shares to be awarded under (or otherwise related to) such Other Stock-Based Awards, whether such Other Stock-Based Awards shall be settled in cash, Shares or a combination of cash and Shares, and all other terms and conditions of such Awards (including, without limitation, the vesting provisions thereof and provisions ensuring that all Shares so awarded and issued shall be fully paid and non-assessable).

9.2 Cash-Based Awards. The Committee, in its sole discretion, may grant Awards that have a value set by the Committee (the "<u>Cash-Based Awards</u>"). Such Cash-Based Awards shall be in such form, and dependent on such conditions, as the Committee shall determine, including, without limitation, the right to receive cash or one or more Shares upon the completion of a specified period of Service, the occurrence of an event and/or the attainment of performance objectives. Cash-Based Awards may be granted alone or in addition to any other Awards granted under the Plan. Subject to the provisions of the Plan, the Committee shall determine to whom and when Cash-Based Awards will be made, whether such Cash-Based Awards shall be settled in cash, Shares or a combination of cash and Shares, and all other terms and conditions of such Awards (including, without limitation, the vesting provisions thereof and provisions ensuring that all Shares so awarded and issued shall be fully paid and non-assessable).

Article 10. Performance-Based Compensation

10.1 Grant of Performance-Based Compensation. To the extent permitted by Section 162(m) of the Code, the Committee is authorized to design any Award so that the amounts or Shares payable or distributed pursuant to such Award are treated as "qualified performance-based compensation" within the meaning of Section 162(m) of the Code and related regulations.

10.2 Performance Measures. The vesting, crediting and/or payment of Performance-Based Compensation shall be based on the achievement of objective performance goals based on one or more of the following Performance Measures: (a) consolidated earnings before or after taxes (including earnings before interest, taxes, depreciation and amortization ("<u>EBITDA</u>")); (b) net income before or after taxes; (c) operating income; (d) earnings per Share; (e) book value per Share; (f) return on shareholders' equity; (g) expense management; (h) return on investment; (i) improvements in capital structure; (j) profitability of an identifiable business unit or product; (k) maintenance or improvement of profit margins; (l) stock price; (m) market share; (n) revenues or sales; (o) costs; (p) cash flow (including, but not limited to, operating cash flow and free cash flow); (q) working capital; (r) return on assets; (s) attainment of objectives relating to store remodels or repair and maintenance; (t) staff training; (u) corporate social responsibility policy implementation; (v) economic value added; (w) debt reduction; (x) completion of acquisitions or divestitures; (y) operating efficiency; (z) sales per square foot; (a) revenue mix; (bb) capital expenditures versus budgeted expenditures (total, exclusive of IT/Games, or maintenance only); (cc) operating income; (dd) income from franchise units; (ee) unit-level EBITDA less G&A expenses; (ff) manager's operating contribution; (gg) regional operating contribution; (h) profitability of various revenue streams; (ii) cash flow per share (before and after dividends or before and after debt payments); (jj) total shareholder return (relative to industry/peer group

and/or absolute); (kk) lease executions; (ll) franchise unit growth; (mm) employee turnover/retention (for entire population or a subset of employee population); (nn) employee satisfaction; (oo) guest satisfaction (overall and/or specific metrics); (pp) guest traffic; (qq) guest loyalty (including but not limited to participation and satisfaction); (rr) attainment of strategic and operational initiatives (MBOs); (ss) marketing/brand awareness scores; (tt) third-party operational/compliance audits; (uu) balanced scorecard; (vv) culinary product pipeline goals; (ww) guest experience; (xx) inventory turnover; (yy) brand positioning goals; (zz) comparable store sales (aaa) return on invested capital; (bbb) new store openings; (ccc) development pipeline goals; (ddd) attainment of objectives relating to acquisitions or divestitures; (eee) attainment of specified business expansion goals; and (fff) expansion of specified programs or initiatives.

Any Performance Measure may be (i) used to measure the performance of the Company and/or any of its Subsidiaries or Affiliates as a whole, any business unit thereof or any combination thereof against any goal including past performance or (ii) compared to the performance of a group of comparable companies, or a published or special index, in each case that the Committee, in its sole discretion, deems appropriate. Subject to Section 162(m) of the Code, the Committee may adjust the performance goals (including to prorate goals and payments for a partial Plan Year) in the event of the following occurrences: (A) non-recurring events, including divestitures, spin-offs, or changes in accounting standards or policies; (B) mergers and acquisitions; and (C) financing transactions, including selling accounts receivable.

10.3 Establishment of Performance Goals for Covered Employees. No later than ninety (90) days after the commencement of a Performance Period (but in no event after twenty-five percent of such Performance Period has elapsed), the Committee shall establish in writing: (a) the performance goals applicable to the Performance Period; (b) the Performance Measures to be used to measure the performance goals in terms of an objective formula or standard; (c) the formula for computing the amount of compensation payable to the Participant if such performance goals are obtained; and (d) the Participants or class of Participants to which such performance goals apply. The outcome of such performance goals must be substantially uncertain when the Committee establishes the goals.

10.4 Adjustment of Performance-Based Compensation. Awards that are designed to qualify as Performance-Based Compensation may not be adjusted upward. The Committee shall retain the discretion to adjust such Awards downward, either on a formula or discretionary basis or any combination, as the Committee determines.

10.5 Certification of Performance. Except for Awards that pay compensation attributable solely to an increase in the value of Shares, no Award designed to qualify as Performance-Based Compensation shall be vested, credited or paid, as applicable, with respect to any Participant until the Committee certifies in writing that the performance goals and any other material terms applicable to such Performance Period have been satisfied.

10.6 Interpretation. Each provision of the Plan and each Award Agreement relating to Performance-Based Compensation shall be construed so that each such Award shall be "qualified performance-based compensation" within the meaning of Section 162(m) of the Code and related regulations, and any provisions of the Award Agreement thereof that cannot be so construed shall be disregarded.

Article 11. Compliance with Section 409A of the Code and Section 457A of the Code

11.1 General. The Company intends that all Awards be structured in compliance with, or to satisfy an exemption from, Section 409A of the Code and all regulations, guidance, compliance programs and other interpretative authority thereunder ("Section 409A"), such that there are no adverse tax consequences, interest, or penalties as a result of the Awards. Notwithstanding the Company's intention, in the event any Award is subject to Section 409A, the Committee may, in its sole discretion and without a Participant's prior consent, amend the Plan and/or Awards, adopt policies and procedures, or take any other actions (including amendments, policies, procedures and actions with retroactive effect) as are necessary or appropriate to (a) exempt the Plan and/or any Award from the application of Section 409A, (b) preserve the intended tax treatment of any such Award, or (c) comply with the requirements of Section 409A, including without limitation any such regulations guidance, compliance programs and other interpretative authority that may be issued after the date of grant of an Award.

11.2 Payments to Specified Employees. Notwithstanding any contrary provision in the Plan or Award Agreement, any payment(s) of nonqualified deferred compensation (within the meaning of Section 409A) that are otherwise required to be made under the Plan to a "specified employee" (as defined under Section 409A) as a result of his or her separation from service (other than a payment that is not subject to Section 409A) shall be delayed for the first six months following such separation from service (or, if earlier, the date of death of the specified employee) and shall instead be paid (in a manner set forth in the Award Agreement) on the payment date that immediately follows the end of such six-month period or as soon as administratively practicable within thirty (30) days thereafter, but in no event later than the end of the applicable taxable year.

11.3 Separation from Service. A termination of employment shall not be deemed to have occurred for purposes of any provision of the Plan or any Award Agreement providing for the payment of any amounts or benefits that are considered nonqualified deferred compensation under Section 409A upon or following a termination of employment, unless such termination is also a "separation from service" within the meaning of Section 409A and the payment thereof prior to a "separation from service" would violate Section 409A. For purposes of any such provision of the Plan or any Award Agreement relating to any such payments or benefits, references to a "termination," "termination of employment" or like terms shall mean "separation from service."

11.4 Section 457A. The Company intends that all Awards be structured in compliance with, or to satisfy an exemption from, Section 457A of the Code and all regulations, guidance, compliance programs and other interpretative authority thereunder ("Section 457A"), such that there are no adverse tax consequences, interest, or penalties as a result of the Awards. Notwithstanding the Company's intention, in the event any Award is subject to Section 457A, the Committee may, in its sole discretion and without a Participant's prior consent, amend the Plan and/or Awards, adopt policies and procedures, or take any other actions (including amendments, policies, procedures and actions with retroactive effect) as are necessary or appropriate to (a) exempt the Plan and/or any Award from the application of Section 457A, (b) preserve the intended tax treatment of any such Award, or (c) comply with the requirements of Section 457A, including without limitation any such regulations, guidance, compliance programs and other interpretative authority that may be issued after the date of the grant.

Article 12. Adjustments

12.1 Adjustments in Authorized Shares. In the event of any corporate event or transaction involving the Company, a Subsidiary and/or an Affiliate (including, but not limited to, a change in the Shares of the Company or the capitalization of the Company) such as a merger, consolidation, reorganization, recapitalization, separation, extraordinary stock dividend, stock split, reverse stock split, split up, spin-off, combination of Shares, exchange of Shares, dividend in kind, amalgamation, or other like change in capital structure (other than regular cash or stock dividends to shareholders of the Company), or any similar corporate event or transaction, the Committee, to prevent dilution or enlargement of Participants' rights under the Plan, shall substitute or adjust, in its sole discretion, the number and kind of Shares or other property that may be issued under the Plan or under particular forms of Awards, the number and kind of Shares are other property subject to outstanding Awards, the Option Price, grant price or purchase price applicable to outstanding Awards, the Annual Award Limits, and/or other value determinations applicable to the Plan or outstanding Awards.

12.2 Change of Control. Upon the occurrence of a Change of Control after the Effective Date, unless otherwise specifically prohibited under applicable laws or by the rules and regulations of any governing governmental agencies or national securities exchanges, or unless the Committee shall determine otherwise in the Award Agreement, the Committee is authorized (but not obligated) to make adjustments to the terms and conditions of outstanding Awards, including without limitation the following (or any combination thereof): (a) continuation or assumption of such outstanding Awards under the Plan by the Company (if it is the surviving company or corporation) or by the surviving company or corporation or its parent; (b) substitution by the surviving company or corporation or its parent; (b) substitution by the surviving company or corporation or its parent; (b) substitution by the surviving company or corporation or its parent; (b) substitution by the surviving company or corporation or its parent; (b) substitution by the surviving company or corporation or its parent; (b) substitution by the surviving company or corporation or its parent; (b) substitution by the surviving company or corporation or its parent; (b) substitution by the surviving company or corporation or its parent; (b) substitution by the surviving company or corporation or its parent; (b) substitution by the surviving company or corporation or its parent; (b) substitution by the surviving company or corporation or its parent; (b) substitution by the surviving company or corporation or its parent; (b) substitution by the surviving company or corporation or its parent; (b) substitution by the surviving company or corporation or its parent; (b) substitution by the surviving company or corporation or its parent; (b) substitution by the surviving company or corporation or its parent; (b) substitution by the surviving and/or lapse of restrictions under outstanding Awards with substantially the same terms for such outstanding Awards; (c) accelerated exer

Article 13. Duration, Amendment, Modification, Suspension and Termination

13.1 Duration of the Plan. Unless sooner terminated as provided in Section 13.2, the Plan shall terminate on the tenth anniversary of the Effective Date.

13.2 Amendment, Modification, Suspension and Termination of Plan. The Committee may amend, alter, suspend, discontinue, or terminate (for purposes of this Section 13.2, an "Action") the Plan or any portion thereof or any Award (or Award Agreement) thereunder at any time; provided, that no such Action shall be made, other than as permitted under Article 11 or 12, (a) without shareholder approval (i) if such approval is necessary to comply with any tax or regulatory requirement applicable to the Plan, (ii) if such Action increases the number of Shares available under the Plan (other than an increase permitted under Article 5 absent shareholder approval), (iii) if such Action results in a material increase in benefits permitted under the Plan (but excluding increases that are immaterial or that are minor and to benefit the administration of the Plan, to take account of any changes in applicable law, or to obtain or maintain favorable tax, exchange, or regulatory treatment for the Company, a Subsidiary, and/or an Affiliate) or a change in eligibility requirements under the Plan, or (iv) for any Action that results in a reduction of the Option Price or grant price per Share, as applicable, of any outstanding Options or Stock Appreciation Rights or cancellation of any outstanding Options or Stock Appreciation Rights, with an Option Price or grant price per Share, as applicable, that is less than such price of the original Options or Stock Appreciation Rights, and (b) without the written consent of the affected Participant, if such Action would materially diminish the rights of any Participant under any Award thereofore granted to such Participant; provided, further, that the Committee may amend the Plan, any Award or any Award Agreement without such consent of the Participant in such manner as it deems necessary to comply with applicable laws, including without limitation, the Dodd-Frank Wall Street Reform and Consumer Protection Act.

Article 14. General Provisions

14.1 No Right to Service. The granting of an Award under the Plan shall impose no obligation on the Company, any Subsidiary or any Affiliate to continue the Service of a Participant and shall not lessen or affect any right that the Company, any Subsidiary or any Affiliate may have to terminate the Service of such Participant. No Participant or other Person shall have any claim to be granted any Award, and there is no obligation for uniformity of treatment of Participants, or holders or beneficiaries of Awards. The terms and conditions of Awards and the Committee's determinations and interpretations with respect thereto need not be the same with respect to each Participant (whether or not such Participants are similarly situated).

14.2 Settlement of Awards; Fractional Shares. Each Award Agreement shall establish the form in which the Award shall be settled. The Committee shall determine whether cash, Awards, other securities or other property shall be issued or paid in lieu of fractional Shares or whether such fractional Shares or any rights thereto shall be rounded, forfeited or otherwise eliminated.

14.3 Tax Withholding. The Company shall have the power and the right to deduct or withhold automatically from any amount deliverable under the Award or otherwise, or require a Participant to remit to the Company, the minimum statutory amount to satisfy federal, state, and local taxes, domestic or foreign, required by law or regulation to be withheld with respect to any taxable event arising as a result of the Plan. With respect to required withholding, Participants may elect (subject to the Company's automatic withholding right set out above), subject to the approval of the Committee, to satisfy the withholding requirement, in whole or in part, by having the Company withhold Shares having a Fair Market Value on the date the tax is to be determined equal to the minimum statutory total tax that could be imposed on the transaction.

14.4 No Guarantees Regarding Tax Treatment. Participants (or their beneficiaries) shall be responsible for all taxes with respect to any Awards under the Plan. The Committee and the Company make no guarantees to any Person regarding the tax treatment of Awards or payments made under the Plan. Neither the Committee nor the Company has any obligation to take any action to prevent the assessment of any tax on any Person with respect to any Award under Section 409A of the Code or Section 457A of the Code or otherwise and none of the Company, any of its Subsidiaries or Affiliates, or any of their employees or representatives shall have any liability to a Participant with respect thereto.

14.5 Non-Transferability of Awards. Unless otherwise determined by the Committee, an Award shall not be transferable or assignable by the Participant except in the event of his death (subject to the applicable laws of descent and distribution) and any such purported assignment, alienation, pledge, attachment, sale, transfer or encumbrance shall be void and unenforceable against the Company or any Affiliate. No transfer shall be permitted for value or consideration. An award exercisable after the death of a Participant may be exercised by the heirs, legatees, personal representatives or distributees of the Participant. Any permitted transfer of the Awards to heirs, legatees, personal representatives or distributees of the Participant shall not be effective to bind the Company unless the Committee shall have been furnished with written notice thereof and a copy of such evidence as the Committee may deem necessary to establish the validity of the transfer and the acceptance by the transferee or transferees of the terms and conditions hereof.

14.6 Conditions and Restrictions on Shares. The Committee may impose such other conditions or restrictions on any Shares received in connection with an Award as it may deem advisable or desirable. These restrictions may include, but shall not be limited to, a requirement that the Participant hold the Shares received for a specified period of time or a requirement that a Participant represent and warrant in writing that the Participant is acquiring the Shares for investment and without any present intention to sell or distribute such Shares. The certificates for Shares may include any legend which the Committee deems appropriate to reflect any conditions and restrictions applicable to such Shares.

14.7 Compliance with Law. The granting of Awards and the issuance of Shares under the Plan shall be subject to all applicable laws, rules, and regulations, and to such approvals by any governmental agencies, or any stock exchanges on which the Shares are admitted to trading or listed, as may be required. The Company shall have no obligation to issue or deliver evidence of title for Shares issued under the Plan prior to: (a) obtaining any approvals from governmental agencies that the Company determines are necessary or advisable; and (b)

completion of any registration or other qualification of the Shares under any applicable national, state or foreign law or ruling of any governmental body that the Company determines to be necessary or advisable. The restrictions contained in this Section 14.7 shall be in addition to any conditions or restrictions that the Committee may impose pursuant to Section 14.6. The inability of the Company to obtain authority from any regulatory body having jurisdiction, which authority is deemed by the Company's counsel to be necessary to the lawful issuance and sale of any Shares hereunder, shall relieve the Company, its Subsidiaries and Affiliates, and all of their employees and representatives of any liability in respect of the failure to issue or sell such Shares as to which such requisite authority shall not have been obtained.

14.8 Awards to Non-U.S. Employees or Directors. To comply with the laws in countries other than the United States in which the Company or any of its Subsidiaries or Affiliates operates or has Employees, Directors or Consultants, the Committee, in its sole discretion, shall have the power and authority to:

- (a) Determine which Subsidiaries or Affiliates shall be covered by the Plan;
- (b) Determine which Employees, Directors or Consultants outside the United States are eligible to participate in the Plan;
- (c) Modify the terms and conditions of any Award granted to Employees, Directors or Consultants outside the United States to comply with applicable foreign laws;
- (d) Take any action, before or after an Award is made, that it deems advisable to obtain approval or comply with any necessary local government regulatory exemptions or approvals; and
- (e) Establish subplans and modify exercise procedures and other terms and procedures, to the extent such actions may be necessary or advisable. Any subplans and modifications to Plan terms and procedures established under this Section 14.8 by the Committee shall be attached to this Plan document as appendices.

14.9 Rights as a Shareholder. Except as otherwise provided herein or in the applicable Award Agreement, a Participant shall have none of the rights of a shareholder with respect to Shares covered by any Award until the Participant becomes the record holder of such Shares.

14.10 Severability. If any provision of the Plan or any Award is or becomes or is deemed to be invalid, illegal, or unenforceable in any jurisdiction, or as to any Person or Award, or would disqualify the Plan or any Award under any law deemed applicable by the Committee, such provision shall be construed or deemed amended to conform to applicable laws, or if it cannot be so construed or deemed amended without, in the determination of the Committee, materially altering the intent of the Plan or the Award, such provision shall be stricken as to such jurisdiction, Person, or Award, and the remainder of the Plan and any such Award shall remain in full force and effect.

14.11 Unfunded Plan. Participants shall have no right, title, or interest whatsoever in or to any investments that the Company or any of its Subsidiaries or Affiliates may make to aid it in meeting its obligations under the Plan. Nothing contained in the Plan, and no action taken pursuant to its provisions, shall create or be construed to create a trust of any kind, or a fiduciary relationship between the Company and any Participant, beneficiary, legal representative, or any other Person. To the extent that any Person acquires a right to receive payments from the Company under the Plan, such right shall be no greater than the right of an unsecured general creditor of the Company. All payments to be made hereunder shall be paid from the general funds of the Company and no sepregation of assets shall be made to assure payment of such amounts. The Plan is not subject to the U.S. Employee Retirement Income Security Act of 1974, as amended from time to time.

14.12 No Constraint on Corporate Action. Nothing in the Plan shall be construed to (a) limit, impair, or otherwise affect the Company's right or power to make adjustments, reclassifications, reorganizations, or changes of its capital or business structure, or to merge or consolidate, or dissolve, liquidate, sell, or transfer all or any part of its business or assets, or (b) limit the right or power of the Company to take any action which such entity deems to be necessary or appropriate.

14.13 Successors. All obligations of the Company under the Plan with respect to Awards granted hereunder shall be binding on any successor to the Company, whether the existence of such successor is the result of a direct or indirect purchase, merger, consolidation, or otherwise, of all or substantially all of the business or assets of the Company.

14.14 Governing Law; Jurisdiction; Waiver of Jury Trial. The Plan and each Award Agreement and all claims, causes of action or proceedings (whether in contract, in tort, at law or otherwise) that may be based upon, arise out of or relate to the Plan and each Award Agreement shall be governed by the internal laws of the State of Delaware, excluding any conflicts or choice-of-law rule or principle that might otherwise refer construction or interpretation of the Plan to the substantive law of another jurisdiction. Each Participant and each party to an Award Agreement agrees that it shall bring all claims, causes of action and proceedings (whether in contract, in tort, at law or otherwise) that may be based upon, arise out of or be related to the Plan and each Award Agreement exclusively in the Delaware Court of Chancery or, in the event (but only in the event) that such court does not have subject matter jurisdiction over such claim, cause of action or proceeding, exclusively in the United States District Court for the District of Delaware (the "<u>Chosen Court</u>"), and hereby (i) irrevocably submits to the exclusive jurisdiction of the Chosen Court, (ii) waives any objection to laying venue in any such proceeding in the Chosen Court, (ii) waives any objection over any party and (iv) agrees that service of process upon such party in any such claim or cause of action shall be effective if notice is given in accordance with an Award Agreement. EACH PARTICIPANT AND EACH PARTY TO AN AWARD AGREEMENT IRREVOCABLY WAIVES ALL RIGHT TO A TRIAL BY JURY IN ANY CLAIM OR CAUSE OF ACTION (WHETHER IN CONTRACT, IN TORT, AT LAW OR OTHERWISE) INSTITUTED BY OR AGAINST SUCH PARTY IN RESPECT OF ITS, HIS OR HER OBLIGATIONS HEREUNDER OR UNDER AN AWARD AGREEMENT.

14.15 Waiver of Certain Claims. By participating in the Plan, the Participant waives all and any rights to compensation or damages in consequence of the termination of his or her office or Service with the Company, any Subsidiary or Affiliate for any reason whatsoever, whether lawfully or otherwise, insofar as those rights arise or may arise from his or her ceasing to have rights under the Plan as a result of such termination, or from the loss or diminution in value of such rights or entitlements, including by reason of the operation of the terms of the Plan, any determination by the Board or Committee pursuant to a discretion contained in the Plan or any Award Agreement or the provisions of any statute or law relating to taxation.

14.16 Data Protection. By participating in the Plan, the Participant consents to the collection, processing, transmission and storage by the Company in any form whatsoever, of any data of a professional or personal nature which is necessary for the purposes of introducing and administering the Plan. The Company may share such information with any Subsidiary or Affiliate, the trustee of any employee benefit trust, its registrars, trustees, brokers, other third party administrator or any Person who obtains control of the Company or acquires the Company, undertaking or part-undertaking which employs the Participant, wherever situated.

14.17 Accounting Restatement. If a Participant receives compensation pursuant to an Award based on financial statements that are subsequently required to be restated in a way that would decrease the value of such compensation, the Participant will, upon the written request of the Committee, in the Committee's sole discretion, forfeit and repay to the Company the difference between what the Participant received and what the Participant should have received based on the accounting restatement, in accordance with (a) the Company's compensation recovery, "clawback" or similar policy, if any, as may be in effect from time to time and (b) any compensation recovery, "clawback" or similar policy made applicable by law including the provisions of Section 945 of the Dodd-Frank Wall Street Reform and Consumer Protection Act and the rules, regulations and requirements adopted thereunder by the Securities and Exchange Commission and/or any national securities exchange on which the Company's equity securities may be listed.

14.18 Effective Date. The Plan shall be effective as of the later of (i) the date of adoption by the Board, which date is set forth below, and (ii) the effectiveness of the Form 8-A in connection with the Company's initial public offering (the "Effective Date").

14.19 Shareholder Approval. The Plan will be submitted for approval by the shareholders of the Company at an annual meeting or any special meeting of shareholders of the Company within twelve (12) months of the Effective Date. Any Awards granted under the Plan prior to such approval of shareholders shall be effective as of the date of grant, but no such Award may be exercised or settled and no restrictions relating to any Award may lapse prior to such shareholder approval, and if shareholders fail to approve the Plan as specified hereunder, the Plan and any Award shall be terminated and cancelled without consideration.

* * *

This Plan was duly adopted and approved by the Board of Directors of the Company by resolution at a meeting held on the $[\times]$ day of $[\times]$, 2014.

DAVE & BUSTER'S ENTERTAINMENT, INC. 2010 MANAGEMENT INCENTIVE PLAN

AMENDED AND RESTATED NONQUALIFIED STOCK OPTION AGREEMENT

THIS AMENDED AND RESTATED NONQUALIFIED STOCK OPTION AGREEMENT (the "<u>Agreement</u>"), dated as of [\times], 2014 (hereinafter referred to as the "<u>Restatement Date</u>") is made by and between Dave & Buster's Entertainment, Inc. (f/k/a Dave & Buster's Parent, Inc.), a Delaware corporation (the "<u>Company</u>") and the individual set forth on <u>Exhibit A</u> hereto ("<u>Optionee</u>").

WHEREAS, the Company adopted the Dave & Buster's Entertainment, Inc. 2010 Management Incentive Plan (the "Plan") pursuant to which among other things, stock options may be granted to purchase Common Stock of the Company; and

WHEREAS, on the Grant Date (as set forth on Exhibit A hereto), the Company granted the Optionee a Nonqualified Stock Option to purchase a number of shares of Common Stock pursuant to a Nonqualified Stock Option Agreement (the "Pre-IPO Agreement"); and

WHEREAS, in connection with the Company's underwritten initial public offering of Shares (the "<u>Initial Public Offering</u>"), the Company shall issue and sell shares of Common Stock in the Initial Public Offering in an amount and at a price to be determined by the Board, and on such other terms and subject to such conditions as may be determined from time to time by the Board (or a properly designated and duly authorized committee thereof) or any duly authorized officer or officers of the Company; and

WHEREAS, the Company has effected a [X] for one stock split of its Common Stock (the "Stock Split"); and

WHEREAS, in connection with the Initial Public Offering, the Committee and the Optionee desire to amend the vesting terms under the Pre-IPO Agreement as set forth herein; and

WHEREAS, in connection with the Stock Split, to prevent dilution or enlargement of the Optionee's rights under the Plan, the Committee has determined to adjust the number of Option Shares and the Option Price as set forth herein; and

WHEREAS, the Company and the Optionee desire to amend and restate the Pre-IPO Agreement as set forth herein.

NOW, THEREFORE, in consideration of the recitals and the mutual agreements herein contained, the parties hereto agree as follows:

1. Grant of Option.

(a) The Company hereby adjusts the options granted to the Optione (collectively, the "<u>Options</u>" and each an "<u>Option</u>") so that, as of the Restatement Date, the Optione may purchase the number of shares of Common Stock set forth on <u>Exhibit A</u> hereto (such shares, the "<u>Option Shares</u>") on the terms and conditions set forth in this Agreement and as otherwise provided in the Plan. The Options are not intended to be treated as "<u>incentive stock</u> <u>options</u>," as such term is defined in Section 422 of the Internal Revenue Code of 1986, as amended (the "<u>Code</u>"). The IPO Vested Options (as set forth on <u>Exhibit A</u> hereto) shall be vested and exercisable in accordance with <u>Section 2(c)(i)</u> hereof, and the Time Vesting Options (as set forth on <u>Exhibit A</u> hereto) shall vest and become exercisable in accordance with <u>Section 2(c)(ii)</u> hereof.

(b) <u>Incorporation by Reference, Etc.</u> The provisions of the Plan are hereby incorporated herein by reference. Except as otherwise expressly set forth herein, this Agreement shall be construed in accordance with the provisions of the Plan and any capitalized terms not otherwise defined in this Agreement shall have the meaning set forth in the Plan.

2. Terms and Conditions.

(a) <u>Purchase Price</u>. The price at which the Optionee shall be entitled to purchase an Option Share upon the exercise of an Option (the "<u>Option</u> <u>Price</u>") shall be as set forth on <u>Exhibit A</u> hereto, representing the Fair Market Value of a share of Common Stock on the Grant Date and as adjusted pursuant to Section 11 of the Plan.

(b) Expiration Date. The Options shall expire at 11:59 p.m. Eastern Time on the tenth anniversary of the Grant Date (the "Expiration Date").

(c) Exercisability of Option.

(i) IPO Vested Options. The IPO Vested Options shall have become vested and exercisable as of the Restatement Date.

(ii) <u>Time Vesting Options</u>. Except as provided in <u>Section 2(d)</u> and <u>Section 2(f)</u> hereof, subject to the Optionee's continued employment with the Company or one of its Subsidiaries, as applicable, on each applicable vesting date, and the Time Vesting Options shall vest and become exercisable in accordance with the schedule set forth on <u>Exhibit A</u> hereto.

(d) Impact of a Change of Control on Options. If either (i) the Optionee remains employed by the Company or one of its Subsidiaries, as applicable, through the date on which a Change of Control is consummated, or (ii) the Optionee's employment with the Company or one of its Subsidiaries, as applicable, is terminated without Cause or for Good Reason no more than one hundred and eighty (180) days prior to a Change of Control, then, immediately prior to such Change of Control, 100% of the Time Vesting Options shall become vested and exercisable. Any Options that have not vested prior to a Change of Control and that do not vest in connection with a Change of Control will be forfeited by the Optionee upon a Change of Control for no consideration.

(e) <u>Method of Exercise</u>. Vested Options may be exercised only by written notice, in the form set forth on <u>Exhibit B</u>, and delivered by the Optionee in person or sent by mail in accordance with <u>Section 3(a)</u> hereof and, in either case, accompanied by payment therefore (including any applicable withholding amounts). The Option Price shall be payable in cash (by certified check or wire transfer), or in the sole discretion of the Committee, (i) by delivery of shares of Common Stock previously acquired with a fair market value equal to the Option Price, subject to any conditions as the Committee may establish, (ii) if there shall be a public market for the Common Stock, in the discretion of the Committee, by delivering to the Committee a copy of irrevocable instructions to a stockbroker to deliver promptly to the Company an amount of loan proceeds, or proceeds of the sale of the Common Stock subject to the Option, sufficient to pay the Option Price or (iii) by such other method as the Committee may allow. Notwithstanding the foregoing, in no event shall an Optionee be permitted to exercise an Option in the manner described in clause (i) or (ii) of the preceding sentence if the Committee determines that exercising the Option in such manner would violate any applicable law or the applicable rules and regulations of the Securities and Exchange Commission or the applicable rules and regulations of any securities exchange or inter dealer quotation system on which the securities of the Company or any Affiliates are listed or traded.

(f) Exercise Upon Termination of Employment. Subject to Section 2(d) hereof, in the event that the Optionee ceases to be employed by the Company or any of its Subsidiaries (the date of such cessation, the "Termination Date"), the Options held by the Optionee (to the extent then outstanding) shall terminate as follows:

(i) <u>Termination Without Cause or For Good Reason</u>. If (x) the Company or one of its Subsidiaries, as applicable, terminates the Optionee's employment without Cause or (y) if the Optionee is a party to an employment agreement with the Company permitting the Optionee to resign his employment for Good Reason and the Optionee resigns his employment for Good Reason, the vested Options shall remain exercisable until the earlier of twelve (12) months following the Termination Date and the Expiration Date and shall thereafter terminate. Subject to <u>Section 2(d)</u>, the remaining Options, to the extent not vested and exercisable, shall terminate and expire on the Termination Date without further consideration to the Optionee.

(ii) <u>Termination Without Good Reason</u>. If the Optionee resigns his employment other than for Good Reason, the vested Options shall remain exercisable until the earlier of thirty-one (31) days following the Termination Date and the Expiration Date and shall thereafter terminate. The remaining Options, to the extent not vested and exercisable, shall terminate and expire on the Termination Date without further consideration to the Optionee.

(iii) <u>Termination Due to Death or Disability</u>. If the Optionee's employment with the Company or one of its Subsidiaries, as applicable, is terminated by reason of his death or Disability, the vested Options shall remain exercisable until the earlier of twelve (12) months following the Termination Date and the Expiration Date and shall thereafter terminate. The remaining Options, to the extent not vested and exercisable, shall terminate and expire on the Termination Date without further consideration to the Optionee.

(iv) <u>All Other Terminations</u>. If the Optionee's employment is terminated for a reason other than as described in <u>Section 2(f)(i)</u> through <u>Section 2(f)(ii)</u> hereof, then all Options, whether or not vested and exercisable at the time of the Optionee's termination of employment, shall terminate and expire on the Termination Date (or, if earlier, the Expiration Date) without further consideration to the Optionee.

(g) Any Options not exercised within the time periods specified above shall be automatically forfeited for no consideration.

- (h) [Intentionally Omitted]
- (i) [Intentionally Omitted]

(j) <u>Rights as Stockholder</u>. The Optionee shall not be deemed for any purpose to be the owner of any of the Option Shares underlying the Options unless, until and to the extent that (i) an Option shall have been exercised pursuant to its terms and (ii) the Company shall have issued and delivered to the Optionee the Option Share underlying such Option.

(k) <u>Withholding Taxes</u>. Prior to the delivery of a certificate or certificates representing any Option Share, as a condition to the exercise of any Option, the Optionee must pay to the Company in cash any amount that the Company determines it is required to withhold under applicable federal, state or local tax laws in respect of the exercise of such Option and the transfer of such Option Share. Notwithstanding the foregoing, if permitted by the Committee, the Optionee may satisfy such withholding obligation by any other method described in <u>Section 10(c)</u> of the Plan or any combination of methods described in <u>Section 10(c)</u> of the Plan; <u>provided</u>, <u>however</u>, that such other method does not violate the terms of any credit agreement to which the Company, or any of its Affiliates is a party or cause a default thereunder.

3. <u>Miscellaneous</u>.

(a) <u>Notices</u>. All notices, demands or other communications provided for or permitted hereunder shall be made in writing and shall be by registered or certified first class mail, return receipt requested, telecopier, courier service, overnight mail or personal delivery:

(i) if to the Company:

Dave & Buster's Entertainment, Inc. 2481 Mañana Drive Dallas, Texas 75220 Attention: Corporate Secretary Fax: (214) 357-1536

(ii) if to the Participant, at the Participant's last known address on file with the Company.

(b) <u>No Right to Continued Employment</u>. Nothing in the Plan or in this Agreement shall confer upon the Optionee any right to continue in the employ of the Company

or one of its Subsidiaries or shall interfere with or restrict in any way the right of the Company or any of its Subsidiaries, which is hereby expressly reserved, to remove, terminate or discharge the Optionee at any time for any reason whatsoever.

(c) <u>Bound by Plan</u>. By signing this Agreement, the Optionee acknowledges that he has received a copy of the Plan and has had an opportunity to review the Plan and agrees to be bound by all the terms and provisions of the Plan.

(d) <u>Successors</u>. The terms of this Agreement shall be binding upon and inure to the benefit of the Company, its successors and assigns, and of the Optionee and the beneficiaries, executors, administrators, heirs and successors of the Optionee.

(e) <u>Invalid Provision</u>. The invalidity or unenforceability of any particular provision hereof shall not affect the other provisions hereof, and this Agreement shall be construed in all respects as if such invalid or unenforceable provision had been omitted.

(f) <u>Modifications</u>. No change, modification or waiver of any provision of this Agreement shall be valid unless the same be in writing and signed by the parties hereto unless such change, modification or waiver is necessary or appropriate to comply with applicable law, including the provisions of Section 409A of the Code and the regulations thereunder.

(g) Entire Agreement. This Agreement and the Plan contain the entire agreement and understanding of the parties hereto with respect to the subject matter contained herein and therein and supersede all prior communications, representations and negotiations in respect thereto.

(h) <u>Governing Law</u>. This Agreement and the rights of the Optionee hereunder shall be construed and determined in accordance with the laws of the State of Delaware, without regard to conflict of laws principles.

(i) <u>Headings</u>. The headings of the Sections hereof are provided for convenience only and are not to serve as a basis for interpretation or construction, and shall not constitute a part, of this Agreement.

(j) <u>Counterparts</u>. This Agreement may be executed in counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.

[Remainder of page intentionally left blank]

IN WITNESS WHEREOF, this Agreement has been executed and delivered by the parties hereto on the first set forth above.

DAVE & BUSTER'S ENTERTAINMENT, INC.

By:

Name: Title:

[Name of Employee]

Exhibit A

Name of Optionee: [X]

Grant Date: [X]1

Number of Option Shares: [X] Shares²

<u>Option Price</u>: $[\times]$ per Share³

Number of IPO Vested Options: [X] Shares⁴

<u>Number of Time Vesting Options</u>: [X] Shares⁵

Exercisability of Time Vesting Options:

[1/4 of the Time Vesting Options shall vest and become exercisable on each of the second, third, fourth and fifth anniversaries of the Grant Date]6

[1/3 of the Time Vesting Options shall vest and become exercisable on each of the third, fourth and fifth anniversaries of the Grant Date]7

[1/2 of the Time Vesting Options shall vest and become exercisable on each of the fourth and fifth anniversaries of the Grant Date]8

[All of the Time Vesting Options shall vest and become exercisable on the fifth anniversary of the Grant Date]9

¹ Original option grant date.

- ² Number of shares subject to the original grant, reduced by any portion of the original grant that has been exercised or forfeited, with the number adjusted for the IPO (rounded down to the nearest share).
- ³ Exercise price of the original grant, adjusted for the IPO (rounded up to the nearest cent).
- ⁴ Number of Option Shares that are vested as of the Restatement Date.
- 5 Number of Option Shares that are unvested as of the Restatement Date.
- ⁶ Vesting schedule for grants made to officers and key employees on 12/5/12 (assuming the Restatement Date occurs before 12/5/14), 5/3/13, 9/23/13 and 9/27/13.
- 7 Vesting schedule for grants made on 3/8/12 and 4/16/12.
- ⁸ Vesting schedule for grants made on 3/23/11 and 7/13/11.
- 9 Vesting schedule for grants made on 6/1/10. In addition, the Time Vesting Options granted to Lacy and Jones shall vest on 6/1/15. All options granted to Directors on 12/5/12 are fully vested.

Exhibit B

NOTICE OF OPTION EXERCISE

I hereby notify Dave & Buster's Entertainment, Inc. (the "<u>Company</u>") that I elect to purchase shares of the Company's common stock at the option exercise price of \$ per share pursuant to options granted to me under the Dave & Buster's Entertainment, Inc. 2010 Management Incentive Plan. Concurrently with the delivery of this Notice, I shall provide for payment in full as set forth below.

Name of O	ptionee:			
Social Security Number:				
Exact name to appear on stock certificate:				
Address to which stock certificate should be sent:				
Date of exercise:				
Method of payment:				
	Cash, check or wire transfer.			
	Delivery of shares of common stock previously acquired.			
	Other (with the approval of the Committee, as applicable): _			
Tax Withholding. I elect to pay required payroll taxes and income tax withholding in connection with this option exercise by one of the following methods:				

- \Box By certified check, which is enclosed.
- By another method specifically approved by the Committee which administers the Dave & Buster's Entertainment, Inc. Management Incentive Plan.

[signature of person exercising the option]

Date:

[name of person exercising the option]

The following items must be attached to this notice:

- 1. Your Option Agreement.
- 2. Payment in full.
- 3. Payment of any federal or state withholding taxes.
- 4. If you are exercising the option upon the Optionee's death, proof of the Optionee's death and proof of your relationship to the Optionee.
- 5. If you are exercising the option as the Optionee's guardian, proof of your appointment.

Dave & Buster's Entertainment, Inc. 2014 Omnibus Incentive Plan

NONQUALIFIED STOCK OPTION AWARD AGREEMENT

THIS NONQUALIFIED STOCK OPTION AWARD AGREEMENT (this "Award Agreement") is made effective as of [×] (the "Date of Grant"), between Dave & Buster's Entertainment, Inc., a Delaware corporation (the "Company") and [×] (the "Participant").

<u>RECITALS</u>:

WHEREAS, the Company has adopted the Dave & Buster's Entertainment, Inc. 2014 Omnibus Incentive Plan (the "Plan"); and

WHEREAS, the Committee has determined that it would be in the best interests of the Company and its stockholders to grant the option provided for herein to the Participant pursuant to the Plan and the terms set forth herein.

NOW THEREFORE, in consideration of the mutual covenants hereinafter set forth, the parties agree as follows:

1. <u>Grant of the Option</u>. The Company hereby grants to the Participant the right and option (the "**Option**") to purchase, on the terms and conditions hereinafter set forth, all or any part of an aggregate of [\times] Shares as of the Date of Grant. The Option is intended to be a Nonqualified Stock Option.

2. <u>Option Price</u>. The purchase price of the Shares subject to the Option is [x] per Share (the "**Option Price**").

3. Option Term. The term of the Option shall be ten (10) years, commencing on the Date of Grant (the "**Option Term**"). The Option shall automatically terminate upon the expiration of the Option Term, or at such earlier time specified herein or in the Plan.

4. <u>Vesting of the Option</u>. Subject to the Participant's continued Service with the Company through the applicable vesting date, Section 5 of this Award Agreement and the terms of the Plan, the Option shall vest in equal installments on [each of the first three (3) anniversaries of the Date of Grant, such that one-third (1/3) of the Option vests on each such anniversary (each, a "**Vesting Date**")]¹; <u>provided</u>, however, that upon the occurrence of a Change of Control, subject to the Participant's continued Service with the Company through the occurrence of such Change of Control, any then-

¹ Grants at IPO are to vest 50% on the third anniversary of the grant and 50% on the fourth anniversary of the grant.

unvested portion of the Option shall immediately become vested and exercisable. At any time, the portion of the Option which has become vested in accordance with the terms hereof shall be called the "Vested Portion."

5. Termination of Service.

(a) <u>Termination of Service for Cause</u>. Upon a termination of the Participant's Service by the Company for Cause the Option, including the Vested Portion, shall immediately terminate and be forfeited without consideration. For purposes of this Award Agreement, "**Cause**" means (i) "Cause" as defined in any employment agreement between the Participant and the Company or any of its Affiliates, or (ii) if there is no such employment agreement or if it does not define Cause: the willful and continued failure by the Participant to perform the duties assigned by the Company, failure to follow reasonable business-related directions from the Company, gross insubordination, theft from the Company or its Affiliates, habitual absenteeism or tardiness, conviction or plea of guilty or *nolo contendere* to a felony, misdemeanor involving fraud, theft or moral turpitude, or any other reckless or willful misconduct that is contrary to the best interests of the Company or materially and adversely affects the reputation of the Company.

(b) <u>Termination of Service due to death or Disability</u>. Upon a termination of the Participant's Service by reason of death or Disability, any unvested portion of the Option shall immediately become vested and the Vested Portion shall remain exercisable until the earlier of (i) one (1) year following such termination of Service and (ii) the expiration of the Option Term. For purposes of this Award Agreement, "**Disability**" means (i) "Disability" as defined in any employment agreement between the Participant and the Company or any of its Affiliates, or (ii) if there is no such employment agreement or if it does not define Disability: the Participant is disabled to the extent that he or she is unable to engage in any substantial gainful activity by reason of any medically determinable physical or mental impairment which can be expected to result in death or can be expected to last for a continuous period of not less than twelve (12) months, or is receiving income replacement benefits for a period of not less than three (3) months under an accident and health plan covering employees of Dave & Buster's Management Corporation, Inc. The determination of the Participant's disability shall be made in good faith by a physician reasonably acceptable to the Company.

(c) <u>Termination of Service due to Retirement</u>. Upon a termination of the Participant's Service by reason of Retirement, subject to the terms of the Plan, any unvested portion of the Option shall continue to vest on each remaining Vesting Date and the Vested Portion shall remain exercisable until the expiration of the Option Term. For purposes of this Award Agreement, "**Retirement**" means (i) "Retirement" as defined in any employment agreement between the Participant and the Company or any of its Affiliates, or (ii) if there is no such employment agreement or if it does not define Retirement: termination of the Participant's Service, other than for Cause, after attaining (A) age sixty (60) and completing ten (10) years of continued service (i.e., without any termination of Service) with the Company or its Affiliates or (B) age sixty-five (65).

(d) <u>Other Terminations of Service</u>. Upon a termination of the Participant's Service for any reason, other than pursuant to <u>Sections 5(a)</u>, <u>5(b)</u> and <u>5(c)</u> above, any unvested portion of the Option shall immediately terminate and be forfeited without consideration and the Vested Portion shall remain exercisable until the earlier of (i) ninety (90) days following such termination of Service and (ii) the expiration of the Option Term.

6. Exercise Procedures.

(a) <u>Notice of Exercise</u>. To the extent exercisable, the Participant or the Participant's representative may exercise the Vested Portion or any part thereof prior to the expiration of the Option Term by giving written notice to the Company in the form attached hereto as <u>Exhibit A</u> (the "**Notice of Exercise**"). The Notice of Exercise shall be signed by the person exercising such Option. In the event that such Option is being exercised by the Participant's representative, the Notice of Exercise shall be accompanied by proof (satisfactory to the Company) of such representative's right to exercise such Option.

(b) <u>Method of Exercise</u>. The Participant or the Participant's representative shall deliver to the Company, at the time the Notice of Exercise is given, payment (i) in cash or its equivalent (e.g., by cashier's check), (ii) in Shares (whether or not previously owned by the Participant) having a Fair Market Value equal to the aggregate Option Price for the Shares being purchased and satisfying such other requirements as may be imposed by the Committee, (iii) partly in cash and partly in Shares (as described in clause (ii) above), (iv) if there is a public market for the Shares at such time, subject to such administrative requirements as may be imposed by the Committee, through the delivery of irrevocable instructions to a broker to sell Shares obtained upon the exercise of the Option and to deliver promptly to the Company an amount out of the proceeds of such sale equal to the aggregate Option Price for the Shares being purchased or (v) to the extent permitted by the Committee, another form of payment permissible under <u>Section 6.5</u> of the Plan for the full amount of the aggregate Option Price for the exercise Option.

(c) <u>Issuance of Shares</u>. Provided the Company receives a properly completed and executed Notice of Exercise and payment for the full amount of the aggregate Option Price, the Company shall promptly cause the Shares underlying the exercised Option to be issued in the name of the Person exercising the applicable Option.

7. <u>No Right to Continued Service</u>. The granting of the Option evidenced hereby and this Award Agreement shall impose no obligation on the Company or any Affiliate to continue the Service of the Participant and shall not lessen or affect any right that the Company or any Affiliate may have to terminate the service of such Participant.

8. Securities Laws/Legend on Certificates. The issuance and delivery of Shares shall comply with all applicable requirements of law, including (without

limitation) the Securities Act of 1933, as amended, the rules and regulations promulgated thereunder, state securities laws and regulations, and the regulations of any stock exchange or other securities market on which the Company's securities may then be traded. If the Company deems it necessary to ensure that the issuance of securities under the Plan is not required to be registered under any applicable securities laws, the Participant shall deliver to the Company an agreement or certificate containing such representations, warranties and covenants as the Company which satisfies such requirements. The certificates representing the Shares shall be subject to such stop transfer orders and other restrictions as the Committee may deem reasonably advisable, and the Committee may cause a legend or legends to be put on any such certificates to make appropriate reference to such restrictions.

9. <u>Transferability</u>. Unless otherwise provided by the Committee, the Option may not be assigned, alienated, pledged, attached, sold or otherwise transferred or encumbered by the Participant other than by will or by the laws of descent and distribution, and any such purported assignment, alienation, pledge, attachment, sale, transfer or encumbrance shall be void and unenforceable against the Company or any Affiliate; <u>provided</u> that, the designation of a beneficiary shall not constitute an assignment, alienation, pledge, attachment, sale, transfer or encumbrance. No such permitted transfer of the Option to heirs or legatees of the Participant shall be effective to bind the Company unless the Committee shall have been furnished with written notice thereof and a copy of such evidence as the Committee may deem necessary to establish the validity of the transfer and the acceptance by the transferee or transferees of the terms and conditions hereof. During the Participant's lifetime, the Option is exercisable only by the Participant (or, if the Participant is disabled, the Participant's representative).

10. <u>Withholding</u>. The Participant may be required to pay to the Company or any Affiliate and the Company shall have the right and is hereby authorized to withhold any applicable withholding taxes in respect of the Option, its exercise or transfer and to take such other action as may be necessary in the option of the Committee to satisfy all obligations for the payment of such withholding taxes.

11. <u>Notices</u>. Any notification required by the terms of this Award Agreement shall be given in writing and shall be deemed effective upon personal delivery or within three (3) days of deposit with the United States Postal Service, by registered or certified mail, with postage and fees prepaid. A notice shall be addressed to the Company, Attention: General Counsel, at its principal executive office and to the Participant at the address that he or she most recently provided to the Company.

12. <u>Entire Agreement</u>. This Award Agreement and the Plan constitute the entire contract between the parties hereto with regard to the subject matter hereof and supersede any other agreements, representations or understandings (whether oral or written and whether express or implied) which relate to the subject matter hereof.

13. <u>Waiver</u>. No waiver of any breach or condition of this Award Agreement shall be deemed to be a waiver of any other or subsequent breach or condition whether of like or different nature.

14. <u>Successors and Assigns</u>. The provisions of this Award Agreement shall inure to the benefit of, and be binding upon, the Company and its successors and assigns and upon the Participant, the Participant's assigns and the legal representatives, heirs and legatees of the Participant's estate, whether or not any such person shall have become a party to this Award Agreement and have agreed in writing to be joined herein and be bound by the terms hereof.

15. <u>Governing Law</u>; Jurisdiction; Waiver of Jury Trial. This Award Agreement and all claims, causes of action or proceedings (whether in contract, in tort, at law or otherwise) that may be based upon, arise out of or relate to this Award Agreement shall be governed by the internal laws of the State of Delaware, excluding any conflicts or choice-of-law rule or principle that might otherwise refer construction or interpretation of the Award Agreement to the substantive law of another jurisdiction. Each party to this Award Agreement agrees that it shall bring all claims, causes of action and proceedings (whether in contract, in tort, at law or otherwise) that may be based upon, arise out of or be related to the Award Agreement exclusively in the Delaware Court of Chancery or, in the event (but only in the event) that such court does not have subject-matter jurisdiction over such claim, cause of action or proceeding, exclusively in the United States District Court for the District of Delaware (the "**Chosen Court**") and hereby (i) irrevocably submits to the exclusive jurisdiction of the Chosen Court, (ii) waives any objection to laying venue in any such proceeding in the Chosen Court, (iii) waives any objection that the Chosen Court is an inconvenient forum or does not have jurisdiction over any party and (iv) agrees that service of process upon such party in any such claim or cause of action shall be effective if notice is given in accordance with this Award Agreement. EACH OF THE PARTIES HERETO IRREVOCABLY WAIVES ALL RIGHT TO A TRIAL BY JURY IN ANY CLAIM OR CAUSE OF ACTION (WHETHER IN CONTRACT, IN TORT, AT LAW OR OTHERWISE) INSTITUTED BY OR AGAINST SUCH PARTY IN RESPECT OF ITS, HIS OR HER OBLIGATIONS HEREUNDER.

16. Option Subject to Plan. By entering into this Award Agreement the Participant agrees and acknowledges that the Participant has received and read a copy of the Plan. The Option is subject to the Plan. The terms and provisions of the Plan as it may be amended from time to time are hereby incorporated herein by reference. In the event of a conflict between any term or provision contained herein and a term or provision of the Plan, the applicable terms and provisions of the Plan will govern and prevail. Capitalized terms not otherwise defined herein shall have the same meanings as in the Plan.

17. <u>No Guarantees Regarding Tax Treatment</u>. The Participant shall be responsible for all taxes with respect to the Option. The Committee and the Company make no guarantees regarding the tax treatment of the Option.

18. <u>Amendment</u>. The Committee may amend or alter this Award Agreement and the Option granted hereunder at any time, subject to the terms of the Plan.

19. <u>Severability</u>. The provisions of this Award Agreement are severable and if any one or more provisions are determined to be illegal or otherwise unenforceable, in whole or in part, the remaining provisions shall nevertheless be binding and enforceable.

20. <u>Signature in Counterparts</u>. This Award Agreement may be signed in counterparts, each of which shall be an original, with the same effect as if the signatures thereto and hereto were upon the same instrument.

* * *

IN WITNESS WHEREOF, the parties hereto have entered into this Award Agreement.

DAVE & BUSTER'S ENTERTAINMENT, INC.

Name: Title:

Acknowledged as of the date first written above:

PARTICIPANT

Option Award Agreement – [\times]

EXHIBIT A Notice of Exercise

Dave & Buster's Entertainment, Inc. 2481 Mañana Drive Dallas, Texas 75220 Attn: General Counsel

Date of Exercise:

Ladies & Gentlemen:

1. Exercise of Option. This constitutes notice to Dave & Buster's Entertainment, Inc. (the "<u>Company</u>") that pursuant to my Nonqualified Stock Option Award Agreement (the "<u>Award Agreement</u>") under the Company's 2014 Omnibus Incentive Plan (the "<u>Plan</u>") I elect to purchase the number of Shares of Company common stock set forth below and for the price set forth below. By signing and delivering this notice to the Company, I hereby acknowledge that I am the holder of the stock option (the "<u>Option</u>") exercised by this notice and have full power and authority to exercise the same.

Date of Grant:	
Number of Shares as to which the Option is exercised (" <u>Optioned Shares</u> "):	
Shares to be issued in name of:	
Total exercise price:	\$
Cash payment or other method of payment permitted under Section 6(b) of the Award Agreement delivered herewith:	\$
Method:	

2. Form of Payment. The Option may be exercised (a) by delivery of cash or its equivalent (e.g., cashier's check), (b) by delivery of Shares (whether or not such shares are Optioned Shares) having a Fair Market Value equal to the aggregate exercise price for the Optioned Shares being purchased and satisfying such other requirements as may be imposed by the Committee (as defined in the Plan), (c) partly in cash and partly in Shares (as described in clause (b) above) or (d) subject to such administrative

requirements as may be imposed by the Committee, through the delivery of irrevocable instructions to a broker to sell Optioned Shares obtained upon the exercise of the Option and to deliver promptly to the Company an amount out of the proceeds of such sale equal to the aggregate exercise price for the Optioned Shares being purchased. Forms of payment other than as described in the immediately preceding sentence are limited by the Plan and are permissible only to the extent approved by the Committee, in its sole discretion.

3. Delivery of Payment. With this notice, I hereby deliver to the Company the full exercise price of the Optioned Shares and any and all withholding taxes due in connection with the exercise of my Option or have otherwise satisfied such requirements.

4. *Rights as Stockholder*. While the Company will endeavor to process this notice in a timely manner, I acknowledge that until the issuance of the shares underlying the Optioned Shares (as evidenced by the appropriate entry on the books of the Company or of a duly authorized transfer agent of the Company), no right to vote or receive dividends or any other rights as a stockholder shall exist with respect to such shares, notwithstanding the exercise of my option(s). No adjustment shall be made for a dividend or other right for which the record date is prior to the date of issuance of the optioned stock.

5. Interpretation. Any dispute regarding the interpretation of this notice shall be submitted promptly by me or by the Company to the Committee. The resolution of such a dispute by the Committee shall be final and binding on all parties.

6. Governing Law; Severability. This notice is governed by the internal substantive laws but not the choice of law rules, of Delaware. In the event that any provision hereof becomes or is declared by a court of competent jurisdiction to be illegal, unenforceable or void, this notice will continue in full force and effect without said provision.

7. Entire Agreement. The Plan and the Award Agreement under which the Optioned Shares were granted are incorporated herein by reference, and together with this notice constitute the entire agreement of the parties with respect to the subject matter hereof.

Very truly yours,

(social security number)

FORM OF AMENDED AND RESTATED

INDEMNIFICATION AGREEMENT

THIS AMENDED AND RESTATED INDEMNIFICATION AGREEMENT (the "<u>Agreement</u>") is made and entered into as of [\times], 2014 between Dave & Buster's Entertainment, Inc., a Delaware corporation (the "<u>Company</u>"), and [\times] ("<u>Indemnitee</u>"). Capitalized terms not defined elsewhere in this Agreement are used as defined in Section 13.

WHEREAS, the Company and Indemnitee made and entered into that certain Indemnification Agreement on [December 6, 2012] (the "Original Agreement"), which the Company and Indemnitee wish to amend and restate in its entirety.

WHEREAS, highly competent persons have become more reluctant to serve corporations as directors or officers or in other capacities unless they are provided with adequate protection through insurance or adequate indemnification against inordinate risks of claims and actions against them arising out of their service to and activities on behalf of the corporation;

WHEREAS, the Board of Directors of the Company (the "<u>Board</u>") has determined that, in order to attract and retain qualified individuals, the Company will attempt to maintain on an ongoing basis, at its sole expense, liability insurance to protect persons serving the Company and its subsidiaries from certain liabilities. Although the furnishing of such insurance has been a customary and widespread practice among United States-based corporations and other business enterprises, the Company believes that, given current market conditions and trends, such insurance may be available to it in the future only at higher premiums and with more exclusions. At the same time, directors, officers, and other persons in service to corporations or business enterprises are being increasingly subjected to expensive and time-consuming litigation relating to, among other things, matters that traditionally would have been brought only against the Company or business enterprise itself. The certificate of incorporation of the Company (a samended, the "<u>Charter</u>") requires indemnification of the officers and directors of the Company. Indemnite may also be entitled to indemnification provisions set forth therein are not exclusive, and thereby contemplate that contracts may be entered into between the Company and members of the board of directors, officers and other persons with respect to indemnification;

WHEREAS, the uncertainties relating to such insurance and to indemnification have increased the difficulty of attracting and retaining such persons;

WHEREAS, the Board has determined that the increased difficulty in attracting and retaining such persons is detrimental to the best interests of the Company's stockholders and that the Company should act to assure such persons that there will be increased certainty of such protection in the future;

WHEREAS, it is reasonable, prudent and necessary for the Company contractually to obligate itself to indemnify, and to advance expenses on behalf of, such persons to the fullest extent permitted by applicable law so that they will serve or continue to serve the Company free from undue concern that they will not be so indemnified;

WHEREAS, the Board has determined that it is in the best interest of the Company to indemnify, and to advance expenses on behalf of, directors and officers of the Company and its subsidiaries, including, without limitation, any officer of any subsidiary of the Company who may be considered a "named executive officer" for purposes of any filings made with the United States Securities and Exchange Commission;

WHEREAS, this Agreement is a supplement to and in furtherance of the Charter of the Company and any resolutions adopted pursuant thereto, and shall not be deemed a substitute therefor, nor to diminish or abrogate any rights of Indemnitee thereunder;

WHEREAS, Indemnitee does not regard the protection available under the Company's Charter and insurance as adequate in the present circumstances, and may not be willing to serve as a director or officer without adequate protection, and the Company desires Indemnitee to serve in such capacity. Indemnitee is willing to serve, continue to serve and to take on additional service for or on behalf of the Company on the condition that he be so indemnified;

WHEREAS, Indemnitee may have certain rights to indemnification and/or insurance provided by a certain source (an "<u>Indemnification/Insurance</u> <u>Source</u>") which Indemnitee and such Indemnification/Insurance Source intends to be secondary to the primary obligation of the Company to indemnify Indemnitee as provided herein, with the Company's acknowledgement and agreement to the foregoing being a material condition to Indemnitee's willingness to serve on the Board; and

NOW, THEREFORE, in consideration of Indemnitee's agreement to serve as a director or officer from and after the date hereof, the parties hereto agree as follows:

1. <u>Indemnity of Indemnitee</u>. The Company hereby agrees to hold harmless and indemnify Indemnitee to the fullest extent permitted by law, as such may be amended from time to time. In furtherance of the foregoing indemnification, and without limiting the generality thereof:

(a) <u>Proceedings Other Than Proceedings by or in the Right of the Company</u>. Indemnitee shall be entitled to the rights of indemnification provided in this <u>Section 1(a)</u> if, by reason of his Corporate Status, Indemnitee is, or is threatened to be made, a party to or participant in any Proceeding other than a Proceeding by or in the right of the Company. Pursuant to this <u>Section 1(a)</u>, Indemnitee shall be indemnified against all Expenses, judgments, penalties, fines and amounts paid in settlement actually and reasonably incurred by him, or on his behalf, in connection with such Proceeding or any claim, issue or matter therein, if Indemnitee acted in good faith and in a manner Indemnitee reasonably believed to be in or not opposed to the best interests of the Company, and with respect to any criminal Proceeding, had no reasonable cause to believe Indemnitee's conduct was unlawful.

(b) <u>Proceedings by or in the Right of the Company</u>. Indemnitee shall be entitled to the rights of indemnification provided in this <u>Section 1(b)</u> if, by reason of his Corporate Status, Indemnitee is, or is threatened to be made, a party to or participant in any

Proceeding brought by or in the right of the Company. Pursuant to this <u>Section 1(b)</u>, Indemnitee shall be indemnified against all Expenses actually and reasonably incurred by Indemnitee, or on Indemnitee's behalf, in connection with such Proceeding if Indemnitee acted in good faith and in a manner Indemnitee reasonably believed to be in or not opposed to the best interests of the Company; provided, however, if applicable law so provides, no indemnification against such Expenses shall be made in respect of any claim, issue or matter in such Proceeding as to which Indemnitee shall have been adjudged to be liable to the Company unless and to the extent that the Court of Chancery of the State of Delaware shall determine that such indemnification may be made.

(c) Indemnification for Expenses of a Party Who is Wholly or Partly Successful. Notwithstanding any other provision of this Agreement, to the extent that Indemnitee is, by reason of his Corporate Status, a party to and is successful, on the merits or otherwise, in any Proceeding, he shall be indemnified to the maximum extent permitted by law, as such may be amended from time to time, against all Expenses actually and reasonably incurred by him or on his behalf in connection therewith. If Indemnitee is not wholly successful in such Proceeding but is successful, on the merits or otherwise, as to one or more but less than all claims, issues or matters in such Proceeding, the Company shall indemnify Indemnitee against all Expenses actually and reasonably incurred by him or on his behalf in connection with each successfully resolved claim, issue or matter. For purposes of this Section 1(c) and without limitation, the termination of any claim, issue or matter in such a Proceeding by dismissal, with or without prejudice, shall be deemed to be a successful result as to such claim, issue or matter.

2. Additional Indemnity. In addition to, and without regard to any limitations on, the indemnification provided for in Section 1 of this Agreement, the Company shall and hereby does indemnify and hold harmless Indemnitee against all Expenses, judgments, penalties, fines and amounts paid in settlement actually and reasonably incurred by him or on his behalf if, by reason of his Corporate Status, he is, or is threatened to be made, a party to or participant in any Proceeding (including a Proceeding by or in the right of the Company), including, without limitation, all liability arising out of the negligence or active or passive wrongdoing of Indemnitee. The only limitation that shall exist upon the Company's obligations pursuant to this Agreement shall be that the Company shall not be obligated to make any payment to Indemnitee that is finally determined (under the procedures, and subject to the presumptions, set forth in Sections 6 and 7 hereof) to be unlawful.

3. Contribution.

(a) Whether or not the indemnification provided in <u>Sections 1</u> and <u>2</u> hereof is available, in respect of any threatened, pending or completed action, suit or proceeding in which the Company is jointly liable with Indemnitee (or would be if joined in such action, suit or proceeding), the Company shall pay, in the first instance, the entire amount of any judgment or settlement of such action, suit or proceeding without requiring Indemnitee to contribute to such payment and the Company hereby waives and relinquishes any right of contribution it may have against Indemnitee. The Company shall not, without the Indemnitee's prior written consent, enter into any such settlement of any action, suit or proceeding (in whole or in part) unless such settlement (i) provides for a full and final release of all claims asserted against Indemnitee and (ii) does not impose any Expense, judgment, fine, penalty or limitation on Indemnitee.

(b) Without diminishing or impairing the obligations of the Company set forth in the preceding subparagraph, if, for any reason, Indemnitee shall elect or be required to pay all or any portion of any judgment or settlement in any threatened, pending or completed action, suit or proceeding in which the Company is jointly liable with Indemnitee (or would be if joined in such action, suit or proceeding), the Company shall contribute to the amount of Expenses, judgments, fines and amounts paid in settlement actually and reasonably incurred and paid or payable by Indemnitee in proportion to the relative benefits received by the Company and all officers, directors or employees of the Company, other than Indemnitee, who are jointly liable with Indemnitee (or would be if joined in such action, suit or proceeding), on the one hand, and Indemnitee, on the other hand, from the transaction from which such action, suit or proceeding arose; provided, however, that the proportion determined on the basis of relative benefit may, to the extent necessary to conform to law, be further adjusted by reference to the relative fault of the Company and all officers, directors or employees of the Company other than Indemnitee who are jointly liable with Indemnitee (or would be if joined in such action, suit or proceeding), on the one hand, and Indemnitee, on the other hand, in connection with the events that resulted in such expenses, judgments, fines or settlement amounts, as well as any other equitable considerations which the Law may require to be considered. The relative fault of the Company and all officers, directors or employees of the Company, other than Indemnitee, who are jointly liable with Indemnitee (or would be if joined in such action, suit or proceeding), on the one hand, and Indemnitee, on the other hand, shall be determined by reference to, among other things, the degree to which their actions were motivated by intent to gain personal profit or advantage, the degree to which their liability is primary or secondary and

(c) The Company hereby agrees to fully indemnify and hold Indemnitee harmless from any claims of contribution which may be brought by officers, directors or employees of the Company, other than Indemnitee, who may be jointly liable with Indemnitee.

(d) To the fullest extent permissible under applicable law, if the indemnification provided for in this Agreement is unavailable to Indemnitee for any reason whatsoever, the Company, in lieu of indemnifying Indemnitee, shall contribute to the amount incurred by Indemnitee, whether for judgments, fines, penalties, excise taxes, amounts paid or to be paid in settlement and/or for Expenses, in connection with any claim relating to an indemnifiable event under this Agreement, in such proportion as is deemed fair and reasonable in light of all of the circumstances of such Proceeding in order to reflect (i) the relative benefits received by the Company and Indemnitee as a result of the event(s) and/or transaction(s) giving cause to such Proceeding; and/or (ii) the relative fault of the Company (and its directors, officers, employees and agents) and Indemnitee in connection with such event(s) and/or transaction(s).

4. <u>Indemnification for Expenses of a Witness</u>. Notwithstanding any other provision of this Agreement, to the extent that Indemnitee is, by reason of his Corporate Status, a witness, or is made (or asked to) respond to discovery requests, in any Proceeding to which Indemnitee is not a party, he shall be indemnified against all Expenses actually and reasonably incurred by him or on his behalf in connection therewith.

5. Advancement of Expenses. Notwithstanding any other provision of this Agreement, the Company shall advance all Expenses incurred by or on behalf of Indemnitee in connection with any Proceeding by reason of Indemnitee's Corporate Status within thirty (30) days after the receipt by the Company of a statement or statements from Indemnitee requesting such advance or advances from time to time, whether prior to or after final disposition of such Proceeding. Such statement or statements shall reasonably evidence the Expenses incurred by Indemnitee and shall include or be preceded or accompanied by a written undertaking by or on behalf of Indemnitee to repay any Expenses advanced if it shall ultimately be determined that Indemnitee is not entitled to be indemnified against such Expenses. Any advances and undertakings to repay pursuant to this Section 5 shall be unsecured and interest free.

6. <u>Procedures and Presumptions for Determination of Entitlement to Indemnification</u>. It is the intent of this Agreement to secure for Indemnitee rights of indemnity that are as favorable as may be permitted under the DGCL and public policy of the State of Delaware. Accordingly, the parties agree that the following procedures and presumptions shall apply in the event of any question as to whether Indemnitee is entitled to indemnification under this Agreement:

(a) To obtain indemnification under this Agreement, Indemnitee shall submit to the Company a written request, including therein or therewith such documentation and information as is reasonably available to Indemnitee and is reasonably necessary to determine whether and to what extent Indemnitee is entitled to indemnification. The Secretary of the Company shall, promptly upon receipt of such a request for indemnification, advise the Board of Directors in writing that Indemnitee has requested indemnification. Notwithstanding the foregoing, any failure of Indemnitee to provide such a request to the Company, or to provide such a request in a timely fashion, shall not relieve the Company of any liability that it may have to Indemnitee unless, and to the extent that, such failure actually and materially prejudices the interests of the Company.

(b) Upon written request by Indemnitee for indemnification pursuant to the first sentence of <u>Section 6(a)</u> hereof, a determination with respect to Indemnitee's entitlement thereto shall be made in the specific case by one of the following four methods, which shall be at the election of the board: (1) by a majority vote of the disinterested directors, even though less than a quorum, (2) by a committee of disinterested directors designated by a majority vote of the disinterested directors, even though less than a quorum, (3) if there are no disinterested directors, or if the disinterested directors so direct, by Independent Counsel in a written opinion to the Board of Directors, a copy of which shall be delivered to Indemnitee, or (4) if so directed by the Board of Directors, by the stockholders of the Company; provided, however, that if a Change in Control has occurred, the determination with respect to Indemnitee's entitlement to indemnification shall be made by Independent Counsel. For purposes hereof, disinterested directors are those members of the board of directors of the Company who are not parties to the action, suit or proceeding in respect of which indemnification is sought by Indemnitee.

(c) In the event the determination of entitlement to indemnification is to be made by Independent Counsel, the Independent Counsel shall be selected as provided in this Section 6(c). If a Change in Control has not occurred, the Independent Counsel shall be selected by the Board of Directors (including a vote of a majority of the Disinterested Directors if obtainable), and the Company shall give written notice to the Indemnitee advising him of the identity of the Independent Counsel so selected. Indemnitee may, within 10 days after such written notice of selection shall have been given, deliver to the Company a written objection to such selection; provided, however, that such objection may be asserted only on the ground that the Independent Counsel so selected does not meet the requirements of "Independent Counsel" as defined in Section 13 of this Agreement, and the objection shall set forth with particularity the factual basis of such assertion. Absent a proper and timely objection, the person so selected shall act as Independent Counsel. If a written objection is made and substantiated, the Independent Counsel selected may not serve as Independent Counsel unless and until such objection is withdrawn or a court has determined that such objection is without merit. If a Change in Control has occurred, the Independent Counsel shall be selected by the Indemnitee (unless the Indemnitee shall request that such selection be made by the Board of Directors, in which event the preceding sentence shall apply), and approved by the Board of Directors (which approval shall not be unreasonably withheld). If (i) an Independent Counsel is to make the determination of entitlement pursuant to this Section 6, and (ii) within 20 days after submission by Indemnitee of a written request for indemnification pursuant to Section 6(a) hereof, no Independent Counsel shall have been selected and not objected to, either the Company or Indemnitee may petition the Court of Chancery of the State of Delaware or other court of competent jurisdiction for resolution of any objection which shall have been made by Indemnitee to the Company's selection of Independent Counsel and/or for the appointment as Independent Counsel of a person selected by the court or by such other person as the court shall designate, and the person with respect to whom all objections are so resolved or the person so appointed shall act as Independent Counsel under Section 6(b) hereof. The Company shall pay any and all reasonable fees and expenses of Independent Counsel incurred by such Independent Counsel in connection with acting pursuant to Section 6(b) hereof, and the Company shall pay all reasonable fees and expenses incident to the procedures of this Section 6(c), regardless of the manner in which such Independent Counsel was selected or appointed.

(d) In making a determination with respect to entitlement to indemnification hereunder, the person or persons or entity making such determination shall presume that Indemnitee is entitled to indemnification under this Agreement. Anyone seeking to overcome this presumption shall have the burden of proof and the burden of persuasion by clear and convincing evidence. Neither the failure of the Company (including by its directors or independent legal counsel) to have made a determination prior to the commencement of any action pursuant to this Agreement that indemnification is proper in the circumstances because Indemnitee has met the applicable standard of conduct, nor an actual determination by the Company (including by its directors or independent legal counsel) that Indemnitee has not met such applicable standard of conduct, shall be a defense to the action or create a presumption that Indemnitee has not met the applicable standard of conduct.

(e) Indemnitee shall be deemed to have acted in good faith if Indemnitee's action is based on the records or books of account of the Enterprise, including financial statements, or on information supplied to Indemnitee by the officers of the Enterprise in

the course of their duties, or on the advice of legal counsel for the Enterprise or on information or records given or reports made to the Enterprise by an independent certified public accountant or by an appraiser or other expert selected with reasonable care by the Enterprise. In addition, the knowledge and/or actions, or failure to act, of any director, officer, agent or employee of the Enterprise shall not be imputed to Indemnitee for purposes of determining the right to indemnification under this Agreement. Whether or not the foregoing provisions of this <u>Section 6(e)</u> are satisfied, it shall in any event be presumed that Indemnitee has at all times acted in good faith and in a manner he reasonably believed to be in or not opposed to the best interests of the Company. Anyone seeking to overcome this presumption shall have the burden of proof and the burden of persuasion by clear and convincing evidence.

(f) If the person, persons or entity empowered or selected under this <u>Section 6</u> to determine whether Indemnitee is entitled to indemnification shall not have made a determination within sixty (60) days after receipt by the Company of the request therefor, the requisite determination of entitlement to indemnification shall be deemed to have been made and Indemnitee shall be entitled to such indemnification absent (i) a misstatement by Indemnitee of a material fact, or an omission of a material fact necessary to make Indemnitee's statement not materially misleading, in connection with the request for indemnification, or (ii) a prohibition of such indemnification under applicable law; provided, however, that such 60-day period may be extended for a reasonable time, not to exceed an additional thirty (30) days, if the person, persons or entity making such determination with respect to entitlement to indemnification in good faith requires such additional time to obtain or evaluate documentation and/or information relating thereto; and provided, further, that the foregoing provisions of this <u>Section 6(f)</u> shall not apply if the determination of entitlement to indemnification is to be made by the stockholders pursuant to <u>Section 6(b)</u> of this Agreement and if (A) within fifteen (15) days after receipt by the Company of the request for such determination, the Board of Directors or the Disinterested Directors, if appropriate, resolve to submit such determination to the stockholders for their consideration at annual meeting thereof to be held within seventy-five (75) days after such receipt and such determination, such meeting is held for such purpose within sixty (60) days after hereat.

(g) Indemnitee shall cooperate with the person, persons or entity making such determination with respect to Indemnitee's entitlement to indemnification, including providing to such person, persons or entity upon reasonable advance request any documentation or information which is not privileged or otherwise protected from disclosure and which is reasonably available to Indemnitee and reasonably necessary to such determination. Any Independent Counsel, member of the Board of Directors or stockholder of the Company shall act reasonably and in good faith in making a determination regarding Indemnitee's entitlement to indemnification under this Agreement. Any costs or expenses (including attorneys' fees and disbursements) incurred by Indemnitee in so cooperating with the person, persons or entity making such determination shall be borne by the Company (irrespective of the determination) and the Company hereby indemnifies and agrees to hold Indemnitee harmless therefrom.

(h) The Company acknowledges that a settlement or other disposition short of final judgment may be successful if it permits a party to avoid expense, delay, distraction, disruption and uncertainty. In the event that any action, claim or proceeding to which Indemnitee is a party is resolved in any manner other than by adverse judgment against Indemnitee (including, without limitation, settlement of such action, claim or proceeding with or without payment of money or other consideration) it shall be presumed that Indemnitee has been successful on the merits or otherwise in such action, suit or proceeding. Anyone seeking to overcome this presumption shall have the burden of proof and the burden of persuasion by clear and convincing evidence.

(i) The termination of any Proceeding or of any claim, issue or matter therein, by judgment, order, settlement or conviction, or upon a plea of nolo contendere or its equivalent, shall not (except as otherwise expressly provided in this Agreement) of itself adversely affect the right of Indemnitee to indemnification or create a presumption that Indemnitee did not act in good faith and in a manner which he reasonably believed to be in or not opposed to the best interests of the Company or, with respect to any criminal Proceeding, that Indemnitee had reasonable cause to believe that his conduct was unlawful.

7. Remedies of Indemnitee.

(a) In the event that (i) a determination is made pursuant to Section 6 of this Agreement that Indemnitee is not entitled to indemnification under this Agreement, (ii) advancement of Expenses is not timely made pursuant to Section 5 of this Agreement, (iii) no determination of entitlement to indemnification is made pursuant to Section 6(b) of this Agreement within 90 days after receipt by the Company of the request for indemnification, (iv) payment of indemnification is not made pursuant to this Agreement within ten (10) days after receipt by the Company of a written request therefor or (v) payment of indemnification is not made within ten (10) days after a determination has been made that Indemnitee is entitled to indemnification or such determination is deemed to have been made pursuant to Section 6 of this Agreement, Indemnitee shall be entitled to an adjudication in an appropriate court of the State of Delaware, or in any other court of competent jurisdiction, of Indemnitee's entitlement to such indemnification, contribution or advancement of Expenses. Alternatively, Indemnitee, at his option, may seek an award in arbitration to be conducted by a single arbitrator pursuant to the Commercial Arbitration Rules of the American Arbitration Association. Except as set forth herein, the provisions of Delaware law (without regard to its conflict of law rules) shall apply to any such arbitration. The Company shall not oppose Indemnitee's right to seek any such adjudication or award in arbitration.

(b) In the event that a determination shall have been made pursuant to <u>Section 6(b)</u> of this Agreement that Indemnitee is not entitled to indemnification, any judicial proceeding commenced pursuant to this <u>Section 7</u> shall be conducted in all respects as a de novo trial, or arbitration, on the merits, and Indemnitee shall not be prejudiced by reason of the adverse determination under <u>Section 6(b)</u>. In any judicial proceeding or arbitration commenced pursuant to this <u>Section 7</u>, Indemnitee shall be presumed to be entitled to indemnification under this Agreement and the Company shall have the burden of proving Indemnitee is not entitled to indemnification or advancement of Expenses, as the case may be, and the Company may not refer to or introduce into evidence any determination pursuant to <u>Section 6(b)</u> of this Agreement

adverse to Indemnitee for any purpose. If Indemnitee commences a judicial proceeding or arbitration pursuant to this <u>Section 7</u>, Indemnitee shall not be required to reimburse the Company for any advances pursuant to <u>Section 5</u> until a final determination is made with respect to Indemnitee's entitlement to indemnification (as to which all rights of appeal have been exhausted or lapsed).

(c) If a determination shall have been made pursuant to <u>Section 6(b)</u> of this Agreement that Indemnitee is entitled to indemnification, the Company shall be bound by such determination in any judicial proceeding commenced pursuant to this <u>Section 7</u>, absent (i) a misstatement by Indemnitee of a material fact, or an omission of a material fact necessary to make Indemnitee's misstatement not materially misleading in connection with the application for indemnification, or (ii) a prohibition of such indemnification under applicable law.

(d) In the event that Indemnitee, pursuant to this <u>Section 7</u>, seeks a judicial adjudication of his rights under, or to recover damages for breach of, this Agreement, or to recover under any directors' and officers' liability insurance policies maintained by the Company, the Company shall pay on his behalf, in advance, any and all expenses (of the types described in the definition of Expenses in <u>Section 13</u> of this Agreement) actually and reasonably incurred by him in such judicial adjudication, regardless of whether Indemnitee ultimately is determined to be entitled to such indemnification, advancement of expenses or insurance recovery.

(e) The Company shall be precluded from asserting in any judicial proceeding commenced pursuant to this <u>Section 7</u> that the procedures and presumptions of this Agreement are not valid, binding and enforceable and shall stipulate in any such court that the Company is bound by all the provisions of this Agreement. The Company shall indemnify Indemnitee against any and all Expenses and, if requested by Indemnitee, shall (within ten (10) days after receipt by the Company of a written request therefore) advance, to the extent not prohibited by law, such expenses to Indemnitee, which are incurred by Indemnitee in connection with any action brought by Indemnitee for indemnification or advance of Expenses from the Company under this Agreement or under any directors' and officers' liability insurance policies maintained by the Company, regardless of whether Indemnitee ultimately is determined to be entitled to such indemnification, advancement of Expenses or insurance recovery, as the case may be.

(f) Notwithstanding anything in this Agreement to the contrary, no determination as to entitlement to indemnification under this Agreement shall be required to be made prior to the final disposition of the Proceeding.

8. Non-Exclusivity; Survival of Rights; Insurance; Primacy of Indemnification; Subrogation.

(a) The rights of indemnification and to receive advancement of expenses as provided by this Agreement shall not be deemed exclusive of any other rights to which Indemnitee may at any time be entitled under applicable law, the Certificate of Incorporation, the By-laws, any agreement, a vote of stockholders, a resolution of directors or otherwise, of the Company. No amendment, alteration or repeal of this Agreement or of any

provision hereof shall limit or restrict any right of Indemnitee under this Agreement in respect of any action taken or omitted by such Indemnitee in his Corporate Status prior to such amendment, alteration or repeal. To the extent that a change in the DGCL, whether by statute or judicial decision, permits greater indemnification than would be afforded currently under the Certificate of Incorporation, By-laws and this Agreement, it is the intent of the parties hereto that Indemnitee shall enjoy by this Agreement the greater benefits so afforded by such change. No right or remedy herein conferred is intended to be exclusive of any other right or remedy, and every other right and remedy shall be cumulative and in addition to every other right and remedy given hereunder or now or hereafter existing at law or in equity or otherwise. The assertion or employment of any right or remedy hereunder, or otherwise, shall not prevent the concurrent assertion or employment of any other right or remedy.

(b) The Company shall obtain and maintain in effect during the entire period for which the Company is obligated to indemnify Indemnitee under this Agreement, one or more policies of insurance with reputable insurance companies to provide the directors and certain officers of the Company with coverage for losses from wrongful acts and omissions and to ensure the Company's performance of its indemnification obligations under this Agreement. Indemnitee shall be covered by such policy or policies in accordance with its or their terms to the maximum extent of the coverage available for any such officer or director under such policy or policies. In all such insurance policies, Indemnitee shall be named as an insured in such a manner as to provide Indemnitee with the same rights and benefits as are accorded to the most favorably insured of the Company's directors and officers. At the time of the receipt of a notice of a claim pursuant to the terms hereof, the Company shall give prompt notice of the commencement of such proceeding to the insurers in accordance with the procedures set forth in the respective policies. The Company shall thereafter take all necessary or desirable action to cause such insurers to pay, on behalf of Indemnitee, all amounts payable as a result of such proceeding in accordance with the terms of such policies.

(c) The Company hereby acknowledges that Indemnitee may have certain rights to indemnification, advancement of expenses and/or insurance provided by an Indemnification/Insurance Source. The Company hereby agrees, if applicable (i) that it is the indemnitor of first resort (i.e., its obligations to Indemnite are primary and any obligation of such Indemnification/Insurance Source to advance expenses or to provide indemnification or to provide insurance for the same expenses or liabilities incurred by Indemnitee are secondary), (ii) that it shall be required to advance the full amount of expenses incurred by Indemnitee and shall be liable for the full amount of all Expenses, judgments, penalties, fines and amounts paid in settlement to the extent legally permitted and as required by the terms of this Agreement and the Certificate of Incorporation or Bylaws of the Company (or any other agreement between the Company and Indemnitee), without regard to any rights Indemnitee may have against such Indemnification/Insurance Source, and, (iii) that it irrevocably waives, relinquishes and releases such Indemnification/Insurance Source from any and all rights and claims against any Indemnification/Insurance Source for contribution, subrogation or any other recovery of any kind in respect thereof. Any insurance coverage provided, obtained or paid for by the Company, on the one hand, and an Indemnification/Insurance Source, on the other hand, shall be subject to the same primary and secondary liability hierarchy set forth in this <u>Section 8(c)</u>. The Company shall use its reasonable best efforts to cause any insurance coverage policy contemplated by this <u>Section 8(c)</u> and obtained by the Company to contain the same primary and secondary liability

hierarchy as set forth in this <u>Section 8(c)</u>. The Company further agrees that no advancement or payment by an Indemnification/Insurance Source on behalf of Indemnitee with respect to any claim for which Indemnitee has sought indemnification from the Company shall affect the foregoing and such Indemnification/Insurance Source shall have a right of contribution and/or be subrogated to the extent of such advancement or payment to all of the rights of recovery of Indemnitee against the Company. The Company and Indemnitee agree that any Indemnification/Insurance Source, if applicable, is an express third party beneficiary of the terms of this <u>Section 8(c)</u>.

(d) Except as provided in <u>Section 8(c)</u> above, in the event of any payment under this Agreement, the Company shall be subrogated to the extent of such payment to all of the rights of recovery of Indemnitee (other than against an Indemnification/Insurance Source, if applicable), who shall execute all papers required and take all action necessary to secure such rights, including execution of such documents as are necessary to enable the Company to bring suit to enforce such rights.

(e) Except as provided in <u>Section 8(c)</u> above, the Company's obligation to indemnify or advance Expenses hereunder to Indemnitee who is or was serving at the request of the Company as a director, officer, employee or agent of any other corporation, partnership, joint venture, trust, employee benefit plan or other Enterprise shall be reduced by any amount Indemnitee has actually received as indemnification or advancement of expenses from such other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise.

9. Exception to Right of Indemnification. Notwithstanding any provision in this Agreement, the Company shall not be obligated under this Agreement to make any indemnity in connection with any claim made against Indemnitee:

(a) except as provided in Section 8(c) above, the Company shall not be liable under this Agreement to make any payment of amounts otherwise indemnifiable hereunder if and to the extent that Indemnitee has otherwise actually received such payment under any insurance policy, contract, agreement or otherwise, except with respect to any excess beyond the amount paid under any insurance policy or other indemnity provision; provided, that the foregoing shall not affect the rights of any Indemnitee or any Indemnification/Insurance Source, if applicable;

(b) for an accounting of profits made from the purchase and sale (or sale and purchase) by Indemnitee of securities of the Company within the meaning of Section 16(b) of the Exchange Act or similar provisions of state statutory law or common law;

(c) for reimbursement to the Company of any bonus or other incentive-based or equity-based compensation or of any profits realized by Indemnitee from the sale of securities of the Company in each case as required under the Exchange Act; or

(d) in connection with any Proceeding (or any part of any Proceeding) initiated by Indemnitee, including any Proceeding (or any part of any Proceeding) initiated by Indemnitee against the Company or its directors, officers, employees or other indemnitees,

unless (i) the Company has joined in or the Board of Directors of the Company authorized the Proceeding (or any part of any Proceeding) prior to its initiation, (ii) the Company provides the indemnification, in its sole discretion, pursuant to the powers vested in the Company under applicable law, or (iii) the Proceeding is one to enforce Indemnitee's rights under this Agreement.

10. <u>Non–Disclosure of Payments</u>. Except as expressly required by the securities laws of the United States of America, neither party shall disclose any payments under this Agreement unless prior approval of the other party is obtained. If any payment information must be disclosed, the Company shall afford the Indemnitee an opportunity to review all such disclosures and, if requested, to explain in such statement any mitigating circumstances regarding the events to be reported.

11. <u>Duration of Agreement</u>. All agreements and obligations of the Company contained herein shall continue upon the later of (a) ten (10) years after the date that Indemnitee shall have ceased to serve as a director or officer of the Company or a director, officer, trustee, partner, managing member, fiduciary, employee or agent of any other corporation, partnership, joint venture, trust, employee benefit plan or other Enterprise which Indemnitee served at the request of the Company; or (b) one (1) year after the final termination of any Proceeding (including any rights of appeal thereto) in respect of which Indemnitee is granted rights of indemnification or advancement of Expenses hereunder and of any Proceeding. This Agreement shall be binding upon and inure to the benefit of and be enforceable by the parties hereto and their respective successors (including any direct or indirect successor by purchase, merger, consolidation or otherwise to all or substantially all of the business or assets of the Company), assigns, spouses, heirs, executors and personal and legal representatives.

12. Security. To the extent requested by Indemnitee and approved by the Board of Directors of the Company, the Company may at any time and from time to time provide security to Indemnitee for the Company's obligations hereunder through an irrevocable bank line of credit, funded trust or other collateral. Any such security, once provided to Indemnitee, may not be revoked or released without the prior written consent of Indemnitee.

13. Definitions. For purposes of this Agreement:

(a) "<u>Change in Control</u>" means any transaction whereby (i) any Person or Persons or "group" (other than the Oak Hill Funds) becomes the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under the Exchange Act), directly or indirectly, in one or a series of transactions, of more than 50% of the total voting power or economic interest of the Company (or any entity which controls the Company or which is a successor to all or substantially all of the assets of the Company), including by way of issuance, merger, consolidation, tender or exchange offer or otherwise, or (ii) all or substantially all of the consolidated assets of the Company and its subsidiaries are sold to any Person or Persons or "group" (other than the Oak Hill Funds). For the avoidance of doubt, neither the initial public offering of Common Stock of the Company nor any secondary public offering shall be deemed a Change in Control.

(b) "<u>Beneficial Owner</u>" shall have the meaning given to such term in Rule 13d-3 under the Exchange Act; provided, however, that Beneficial Owner shall exclude any Person otherwise becoming a Beneficial Owner by reason of the stockholders of the Company approving a merger of the Company with another entity.

(c) "<u>Common Stock</u>" means the Company's common stock, par value \$0.01 per share, and any stock into which such Common Stock may thereafter be converted, changed, reclassified or exchanged.

(d) "<u>Corporate Status</u>" describes the status of a person who is or was a director, officer, employee, agent or fiduciary of the Company, any direct or indirect subsidiary of the Company, or of any other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise that such person is or was serving at the request of the Company.

(e) "Disinterested Director" means a director of the Company who is not and was not a party to the Proceeding in respect of which indemnification is sought by Indemnitee.

(f) "<u>Enterprise</u>" shall mean the Company and any other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise that Indemnitee is or was serving at the request of the Company as a director, officer, employee, agent or fiduciary.

(g) "Exchange Act" shall mean the Securities Exchange Act of 1934, as amended.

(h) "Expenses" shall include all reasonable attorneys' fees, retainers, court costs, transcript costs, fees of experts, witness fees, travel expenses, duplicating costs, printing and binding costs, telephone charges, postage, delivery service fees and all other disbursements or expenses of the types customarily incurred in connection with prosecuting, defending, preparing to prosecute or defend, investigating, participating, or being or preparing to be a witness in a Proceeding, or responding to, or objecting to, a request to provide discovery in any Proceeding. Expenses also shall include Expenses incurred in connection with any appeal resulting from any Proceeding and any federal, state, local or foreign taxes imposed on Indemnitee as a result of the actual or deemed receipt of any payments under this Agreement, including without limitation the premium, security for, and other costs relating to any cost bond, supersede as bond, or other appeal bond or its equivalent. Expenses, however, shall not include amounts paid in settlement by Indemnitee or the amount of judgments or fines against Indemnitee.

(i) "Independent Counsel" means a law firm, or a member of a law firm, that is experienced in matters of corporation law and neither presently is, nor in the past five years has been, retained to represent: (i) the Company or Indemnitee in any matter material to either such party (other than with respect to matters concerning Indemnitee under this Agreement, or of other indemnitees under similar indemnification agreements), or (ii) any other party to the Proceeding giving rise to a claim for indemnification hereunder. Notwithstanding the foregoing, the term "Independent Counsel" shall not include any person who, under the applicable standards of professional conduct then prevailing, would have a conflict of interest in

representing either the Company or Indemnitee in an action to determine Indemnitee's rights under this Agreement. The Company agrees to pay the reasonable fees of the Independent Counsel referred to above and to fully indemnify such counsel against any and all Expenses, claims, liabilities and damages arising out of or relating to this Agreement or its engagement pursuant hereto.

(j) "<u>Person</u>" shall have the meaning as set forth in Sections 13(d) and 14(d) of the Exchange Act; provided, however, that Person shall exclude (i) the Company, (ii) any trustee or other fiduciary holding securities under an employee benefit plan of the Company, and (iii) any corporation owned, directly or indirectly, by the stockholders of the Company in substantially the same proportions as their ownership of stock of the Company.

(k) "<u>Proceeding</u>" includes any threatened, pending or completed action, suit, arbitration, alternate dispute resolution mechanism, investigation, inquiry, administrative hearing or any other actual, threatened or completed proceeding, whether brought by or in the right of the Company or otherwise and whether civil, criminal, administrative or investigative, in which Indemnitee was, is or will be involved as a party or otherwise, by reason of the fact that Indemnitee is or was a director and/or officer of the Company, by reason of any action taken by him or of any inaction on his part while acting as a director and/or officer of the Company, by reason of any action taken by him or of any inaction on his part while acting as a director and/or officer of the Company, or by reason of the fact that he is or was serving at the request of the Company as a director, officer, employee, agent or fiduciary of another corporation, partnership, joint venture, trust or other Enterprise; in each case whether or not he is acting or serving in any such capacity at the time any liability or expense is incurred for which indemnification can be provided under this Agreement; including one pending on or before the date of this Agreement, but excluding one initiated by an Indemnitee pursuant to Section 7 of this Agreement to enforce his rights under this Agreement.

(1) "<u>Oak Hill Funds</u>" means, collectively, Oak Hill Capital Partners III, L.P., Oak Hill Capital Management Partners III, L.P., funds managed or advised by Oak Hill Capital Management, LLC and each of their respective affiliates (within the meaning of Section 12b-2 under the Exchange Act).

14. <u>Severability</u>. If any provision or provisions of this Agreement shall be held to be invalid, illegal or unenforceable for any reason whatsoever: (a) the validity, legality, and enforceability of the remaining provisions of this Agreement (including, without limitation, each portion of any Section, paragraph or sentence of this Agreement containing any such provision held to be invalid, illegal or unenforceable, that is not itself invalid, illegal or unenforceable) shall not in any way be affected or impaired thereby and shall remain enforceable to the fullest extent permitted by law; (b) such provision or provisions shall be deemed reformed to the fullest extent necessary to conform to applicable law and to give the maximum effect to the intent of the parties hereto; and (c) to the fullest extent possible, the provision of this Agreement (including, without limitation, each portion of any Section, paragraph or sentence of this Agreement containing any such provision held to be invalid, illegal or unenforceable, that is not itself invalid, illegal or unenforceable) shall be construed so as to give effect to the intent manifested thereby. Without limiting the generality of the foregoing, this Agreement is intended to confer upon Indemnitee indemnification rights to the fullest extent permitted by applicable laws.

15. Enforcement and Binding Effect.

(a) The Company expressly confirms and agrees that it has entered into this Agreement and assumed the obligations imposed on it hereby in order to induce Indemnitee to serve as a director and/or officer of the Company, and the Company acknowledges that Indemnitee is relying upon this Agreement in serving as a director and/or officer of the Company.

(b) This Agreement hereby amends and restates the Original Agreement in all respects. Without limiting any of the rights of Indemnitee under the Charter or Bylaws of the Company as they may be amended from time to time, this Agreement constitutes the entire agreement between the parties hereto with respect to the subject matter hereof and supersedes all prior agreements and understandings, oral, written and implied, between the parties hereto with respect to the subject matter hereof, including, without limitation, the Original Agreement.

(c) The indemnification and advancement of expenses provided by, or granted pursuant to this Agreement shall be binding upon and be enforceable by the parties hereto and their respective successors and assigns (including any direct or indirect successor by purchase, merger, consolidation or otherwise to all or substantially all of the business or assets of the Company), shall continue as to an Indemnitee who has ceased to be a director, officer, employee or agent of the Company or of any other Enterprise at the Company's request, and shall inure to the benefit of Indemnitee and his or her spouse, assigns, heirs, devisees, executors and administrators and other legal representatives.

(d) The Company shall require and cause any successor (whether direct or indirect by purchase, merger, consolidation or otherwise) to all, substantially all or a substantial part, of the business and/or assets of the Company to expressly to assume and agree to perform this Agreement in the same manner and to the same extent that the Company would be required to perform if no such succession had taken place.

(e) The Company and Indemnitee agree herein that a monetary remedy for breach of this Agreement, at some later date, may be inadequate, impracticable and difficult of proof, and further agree that such breach may cause Indemnitee irreparable harm. Accordingly, the parties hereto agree that Indemnitee may enforce this Agreement by seeking injunctive relief and/or specific performance hereof, without any necessity of showing actual damage or irreparable harm and that by seeking injunctive relief and/or specific performance, Indemnitee shall not he precluded from seeking or obtaining any other relief to which he may be entitled. The Company and Indemnitee further agree that Indemnitee shall be entitled to such specific performance and injunctive relief, including temporary restraining orders, preliminary injunctions and permanent injunctions, without the necessity of posting bonds or other undertaking in connection therewith. The Company acknowledges that in the absence of a waiver, a bond or undertaking may be required of Indemnitee by the Court, and the Company hereby waives any such requirement of such a bond or undertaking.

16. <u>Modification and Waiver</u>. No supplement, modification, termination or amendment of this Agreement shall be binding unless executed in writing by both of the parties

hereto. No waiver of any of the provisions of this Agreement shall be deemed or shall constitute a waiver of any other provisions hereof (whether or not similar) nor shall such waiver constitute a continuing waiver.

17. <u>Notice By Indemnitee</u>. Indemnitee agrees promptly to notify the Company in writing upon being served with or otherwise receiving any summons, citation, subpoena, complaint, indictment, information or other document relating to any Proceeding or matter which may be subject to indemnification covered hereunder. The failure to so notify the Company shall not relieve the Company of any obligation which it may have to Indemnitee under this Agreement or otherwise.

18. <u>Notices</u>. All notices and other communications given or made pursuant to this Agreement shall be in writing and shall be deemed effectively given: (a) upon personal delivery to the party to be notified, (b) when sent by confirmed electronic mail or facsimile if sent during normal business hours of the recipient, and if not so confirmed, then on the next business day, (c) five (5) days after having been sent by registered or certified mail, return receipt requested, postage prepaid, or (d) one (1) day after deposit with a nationally recognized overnight courier, specifying next day delivery, with written verification of receipt. All communications shall be sent:

(a) To Indemnitee at the address set forth below Indemnitee signature hereto.

(b) To the Company at:

Dave & Buster's Entertainment, Inc. 2481 Manana Drive Dallas, Texas 75220 Attention: Jay L. Tobin, Esq. Fax: (214) 357-1536

With a copy to:

Oak Hill Capital Management, LLC 65 East 55th Street, 32nd Floor New York, NY 10022 Attention: John R. Monsky, Esq. Fax: (212) 527-8450

and

Weil, Gotshal & Manges LLP 767 Fifth Avenue New York, New York 10153 Attention: Douglas P. Warner, Esq. Fax: (212) 310-8007

or to such other address as may have been furnished to Indemnitee by the Company or to the Company by Indemnitee, as the case may be.

19. <u>Counterparts</u>. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same Agreement. This Agreement may also be executed and delivered by facsimile signature and in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.

20. <u>Headings</u>. The headings of the paragraphs of this Agreement are inserted for convenience only and shall not be deemed to constitute part of this Agreement or to affect the construction thereof.

21. Usage of Pronouns. Use of the masculine pronoun shall be deemed to include usage of the feminine pronoun where appropriate.

22. <u>Governing Law and Consent to Jurisdiction</u>. This Agreement and the legal relations among the parties shall be governed by, and construed and enforced in accordance with, the laws of the State of Delaware, without regard to its conflict of laws rules. The Company and Indemnitee hereby irrevocably and unconditionally (i) agree that any action or proceeding arising out of or in connection with this Agreement shall be brought only in the Chancery Court of the State of Delaware (the "Delaware Court"), and not in any other state or federal court in the United States of America or any court in any other country, (ii) generally and unconditionally consent to submit to the exclusive jurisdiction of the Delaware Court for purposes of any action or proceeding arising out of or in connection with this Agreement, (iii) waive any objection to the laying of venue of any such action or proceeding in the Delaware Court, and (iv) waive, and agree not to plead or to make, any claim that any such action or proceeding brought in the Delaware Court has been brought in an improper or inconvenient forum. The foregoing consent to jurisdiction shall not constitute general consent to service of process in the state for any purpose except as provided above, and shall not be deemed to confer rights on any person other than the parties to this Agreement.

[SIGNATURE PAGE TO FOLLOW]

¹⁷

IN WITNESS WHEREOF, the parties hereto have executed this Agreement on and as of the day and year first above written.

DAVE & BUSTER'S ENTERTAINMENT, INC.

[**x**]

INDEMNITEE

[**X**]

Address: